UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

Form 10-K

(Mark One) ☑

ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the fiscal year ended December 31, 2014

OR

□ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

Commission file number: 333-191106

HINES GLOBAL REIT II, INC.

(Exact Name of Registrant as Specified in its Charter)

Maryland

(State or Other Jurisdiction of Incorporation or Organization)

80-0947092

(I.R.S. Employer Identification No.)

77056-6118 (Zip code)

Houston, Texas (Address of principal executive offices)

> Registrant's telephone number, including area code: (888) 220-6121 Securities registered pursuant to Section 12(b) of the Act: None. Securities registered pursuant to Section 12(g) of the Act: None.

Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes 🗆 No 🗹

Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Exchange Act. Yes 🗆 No 🗹

Indicate by check mark whether the registrant: (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \square No \square

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (\$232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). Yes \square No \square

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K (229.405 of this chapter) is not contained herein, and will not be contained, to the best of the registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K.

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer \Box	Accelerated filer \Box	Non-accelerated filer \Box	Smaller reporting company 🗹	
(Do not check if a smaller reporting company)				

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes 🗆 No 🗹

Aggregate market value of the common stock held by non-affiliates of the registrant: No established market exists for the registrant's common stock.

The registrant had 1.9 million shares of common stock outstanding as of March 18, 2015.

2800 Post Oak Boulevard Suite 5000

TABLE OF CONTENTS PART I

Item 1.	Business	<u>1</u>		
Item 1A.	Risk Factors	<u>1</u> <u>5</u>		
Item 1B.	Unresolved Staff Comments	<u>36</u>		
Item 2.	Properties	<u>37</u>		
Item 3.	Legal Proceedings	<u>38</u>		
Item 4.	Mine Safety Disclosures	<u>38</u>		
PART II				
Item 5.	Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities	<u>39</u>		
Item 6.	Selected Financial Data	<u>43</u>		
Item 7.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>44</u>		
Item 7A.	Quantitative and Qualitative Disclosures About Market Risk	<u>53</u>		
Item 8.	Financial Statements and Supplementary Data	<u>54</u>		
Item 9.	Changes in and Disagreements With Accountants on Accounting and Financial Disclosure	<u>71</u>		
Item 9A.	Controls and Procedures	<u>71</u>		
Item 9B.	Other Information	<u>71</u>		
	PART III			
Item 10.	Directors, Executive Officers and Corporate Governance	<u>72</u>		
Item 11.	Executive Compensation	<u>75</u>		
Item 12.	Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters	<u>76</u>		
Item 13.	Certain Relationships and Related Transactions, and Director Independence	<u>77</u>		
Item 14.	Principal Accounting Fees and Services	<u>81</u>		
	PART IV			
Item 15.	Exhibits, Financial Statement Schedules	<u>83</u>		
SIGNATURE	2 <u>S</u>			
EX-10.1	Amended and Restated Agreement of Limited Partnership			
EX- 21.1	List of Subsidiaries			
EX- 31.1	Certification			
EX- 31.2	Certification			
EX- 32.1	Certification of CEO & CFO pursuant to Section 906			
EX- 101	Instance Document			
EX-101	Schema Document			
EX-101	Calculation Linkbase Document			
EX-101	Labels Linkbase Document			
EX-101	Presentation Linkbase Document			
EX-101	Definition Linkbase Document			

PART I

Special Note Regarding Forward-Looking Statements

This Annual Report on Form 10-K includes certain statements that may be deemed forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended (the "Securities Act"), and Section 21E of the Securities Exchange Act of 1934, as amended (the "Exchange Act"). Such forward-looking statements relate to, without limitation, potential future acquisitions, the completion of projects in development, economic conditions that may impact our operations, our future leverage and financial position, our future capital expenditures, future distributions, other developments and trends in the commercial real estate industry and our business strategy. Forward-looking statements are generally identifiable by the use of the words "may," "will," "should," "expect," "could," "intend," "plan," "anticipate," "estimate," "believe," "continue," "predict," "potential" or the negative of these words or other comparable terminology. These statements are not guarantees of future performance, and involve certain risks, uncertainties and assumptions that are difficult to predict.

The forward-looking statements in this Annual Report on Form 10-K are based on our current expectations, plans, estimates, assumptions and beliefs that involve numerous risks and uncertainties. Assumptions relating to the foregoing involve judgments with respect to, among other things, future economic, competitive and market conditions and future business decisions, all of which are difficult or impossible to predict accurately and many of which are beyond our control. Any of the assumptions underlying forward-looking statements could prove to be inaccurate. To the extent that our assumptions differ from actual results, our ability to meet such forward-looking statements, including our ability to generate positive cash flow from operations, provide distributions to our stockholders and maintain the value of the real estate properties in which we hold an interest, may be significantly hindered.

Our stockholders are cautioned not to place undue reliance on any forward-looking statement in this Annual Report on Form 10-K. All forward-looking statements are made as of the date of this Annual Report on Form 10-K, and the risk that actual results will differ materially from the expectations expressed in this Annual Report on Form 10-K may increase with the passage of time. In light of the significant uncertainties inherent in the forward-looking statements in this Annual Report on Form 10-K, the inclusion of such forward-looking statements should not be regarded as a representation by us or any other person that the objectives and plans set forth in this Annual Report on Form 10-K will be achieved. Please see "Item 1A. Risk Factors" for a discussion of some of the risks and uncertainties that could cause actual results to differ materially from those presented in certain forward-looking statements.

Item 1. Business

General Description of Business and Operations

Hines Global REIT II, Inc. ("Hines Global II") was formed as a Maryland corporation on July 31, 2013, primarily for the purpose of investing in a diversified portfolio of quality commercial real estate properties and other real estate investments located throughout the United States and internationally. On August 20, 2014, Hines Global II commenced an offering of up to \$2.5 billion in shares of Class A Common Stock ("common stock") for sale to the public (the "Offering"). On September 26, 2014, Hines Global II received an investment of \$2.0 million from Hines Global REIT II Investor Limited Partnership and achieved the minimum offering requirements pursuant to the terms of the Offering in all states except for Pennsylvania and Washington, which have higher minimum offering requirements. As of March 18, 2015, Hines Global II had received gross offering proceeds of \$18.3 million from the sale of 1.9 million common shares.

Hines Global II expects to conduct most of its activities through, and most of its real estate investments are held directly or indirectly by, Hines Global REIT II Properties, LP (the "Operating Partnership"), which was formed on July 31, 2013. Hines Global II contributes the proceeds it receives from the issuance of common shares to the Operating Partnership and the Operating Partnership in turn issues general partner interests to Hines Global II. The general partner interests entitle Hines Global II to receive its share of the Operating Partnership's earnings or losses and distributions of cash flow. Hines Global II is constructed in a manner that would allow the Operating Partnership to issue limited partner units from time to time in exchange for real estate properties. By structuring acquisitions in this manner, the sellers of the real estate will generally be able to defer the taxation of gains until they exchange their limited partner units from common shares of Hines Global II or sell or redeem their units.

We refer to Hines Global II, the Operating Partnership and its wholly-owned subsidiaries as the "Company," and the use of "we," "our," "us" or similar pronouns in this annual report refers to Hines Global II or the Company as required by the context in which such pronoun is used.

We made our initial real estate investment in December 2014 and owned one property, 2819 Loker Avenue East, as of December 31, 2014. The Class A industrial building, located in Carlsbad, California, consists of 161,310 square feet of leasable space. Additionally, in March 2015, we acquired Bishop's Square, a Class A office building located in Dublin, Ireland, which consists of approximately 153,529 square feet of leasable space. We may purchase properties or make other real estate investments that relate to varying property types including office, retail, industrial, multi-family residential and hospitality or leisure. We may invest in operating properties, properties under development, and undeveloped properties such as land. Other real estate investments may include equity or debt interests, including securities, in other real estate entities and debt related to properties such as mortgages, mezzanine loans, B-notes, bridge loans, construction loans and securitized debt.

We have no employees. Our business is managed by Hines Global REIT II Advisors LP (the "Advisor"), an affiliate of Hines Interests Limited Partnership ("Hines"), under the terms and conditions of an advisory agreement between us and the Advisor (the "Advisory Agreement"). As compensation for these services, we will pay the Advisor asset management, acquisition and disposition fees and we reimburse certain of the Advisor's expenses incurred on our behalf in accordance with the Advisory Agreement. We expect Hines or affiliates of Hines to manage the leasing and operations of most of the properties in which we invest and, accordingly, we will pay Hines property management and leasing fees in connection with these services. Hines is owned and controlled by, or for the benefit of, Gerald D. Hines and his son Jeffrey C. Hines, the Chairman of our board of directors. Hines and its 3,450 employees have over 50 years of experience in the areas of investment selection, underwriting, due diligence, portfolio management, asset management, property management, leasing, disposition, finance, accounting and investor relations.

Our office is located at 2800 Post Oak Boulevard, Suite 5000, Houston, Texas 77056-6118. Our telephone number is 1-888-220-6121. Our web site is www.HinesSecurities.com. The information on our website is not incorporated by reference into this report.

Primary Investment Objectives

Our primary investment objectives are to:

- preserve invested capital;
- invest in a diversified portfolio of quality commercial real estate properties and other real estate investments;
- provide income in the form of regular, stable cash distributions;
- provide modest growth in the value of invested capital;
- achieve attractive total returns upon the ultimate sale of our investments or occurrence of another liquidity event; and
- qualify as a real estate investment trust ("REIT") for federal income tax purposes, beginning with the year ended December 31, 2015.

Acquisition and Investment Policies

We intend to invest primarily in a diversified portfolio of quality commercial real estate properties and other real estate investments throughout the United States and internationally. We may purchase properties or make other real estate investments that relate to varying property types including office, retail, industrial, multi-family residential and hospitality or leisure. We may invest in operating properties, properties under development, and undeveloped properties such as land. Other real estate investments may include equity or debt interests including securities in other real estate entities and debt related to properties such as mortgages, mezzanine loans, B-notes, bridge loans, construction loans and securitized debt. We believe that there is an opportunity to create attractive total returns by employing a strategy of investing in a diversified portfolio of such investments which are well-selected, well-managed and disposed of at an optimal time. Our principal targeted assets are investments in properties, and other real estate investments are, or relate to, properties generally located in central business districts or suburban markets of major metropolitan cities worldwide. We intend to invest in a geographically diverse portfolio in order to reduce the risk of reliance on a particular market, a particular property and/or a particular tenant. We anticipate that international real estate investments may comprise a substantial portion of our portfolio.

We may invest in real estate properties and other real estate investments directly by owning 100% of such investments or indirectly by owning less than 100% of such investments through co-ownership or joint-venture arrangements with third parties or with other Hines-affiliated entities. We intend to fund our future acquisitions and investments primarily with proceeds raised in the Offering and potential follow-on offerings as well as with proceeds from debt financings.

We are not limited as to the asset types or geographic areas in which we may invest and conduct our operations. We are not specifically limited in the number or size of investments we may make, or on the percentage of net proceeds of the Offering that we may invest in a single property, real estate investment or loan. Although the actual percentages may vary from those presently anticipated, after the proceeds of the Offering and any subsequent offerings have been fully invested, we anticipate that international real estate investments will comprise between 40% and 60% of our portfolio and real estate investments other than the acquisition of commercial real estate properties will comprise less than 30% of our portfolio. The number, size and mix of investments we make will depend upon real estate and market conditions and other circumstances existing at the time we are evaluating investment opportunities and the amount of proceeds we raise in the Offering and any subsequent offerings.

Financing Strategy and Policies

We expect that once we have fully invested the proceeds of the Offering and other potential subsequent offerings, our debt financing, including our pro rata share of the debt financing of entities in which we invest, will be in the range of approximately 40%—60% of the aggregate value of our real estate investments and other assets. Financing for acquisitions and investments may be obtained at the time an asset is acquired or an investment is made or at such later time as we determine to be appropriate. In addition, debt financing may be used from time to time for property improvements, lease inducements, tenant improvements and other working capital needs, including the payment of distributions. Additionally, the amount of debt placed on an individual property or related to a particular investment, including our pro rata share of the amount of debt incurred by an individual entity in which we invest, may be less than 40% or more than 60% of the value of such property/ investment or the value of the assets owned by such entity, depending on market conditions and other factors. Our aggregate borrowings, secured and unsecured, must be reasonable in relation to our net assets (which approximates 75% of the cost of our assets) unless any excess borrowing is approved by a majority of our independent directors and is disclosed to our stockholders in our next quarterly report along with justification for the excess. As described below, our independent directors have approved borrowings in excess of these limitations in connection with our first two investments, as we are in the early stages of raising capital through the Offering.

In November 2014, our board of directors, including all of our independent directors, approved a \$75.0 million unsecured credit facility (the "Hines Credit Facility") between us and Hines to be used to fund acquisitions and other working capital needs. Additionally, in December 2014, our board of directors, including all of our independent directors, approved a \$24.2 million borrowing under this credit facility to fund our acquisition of 2819 Loker Avenue East. This borrowing represents approximately 95% of the cost of 2819 Loker Avenue East. Further, subsequent to December 31, 2014, our board of directors, including all of our independent directors, approved s 45.2 million secured credit facility (\$62.1 million using \$1.12 per Euro as of the transaction date) for the purchase of Bishop's Square in March 2015. In total, these borrowings represent approximately 104% of the cost of Bishop's Square based on the contract purchase price. In March 2015, we repaid \$17.0 million in borrowings under the Hines Credit Facility using proceeds from the Offering.

Our existing indebtedness and any additional indebtedness we do incur will likely be subject to continuing covenants, and we will likely be required to make continuing representations and warranties about the Company in connection with such debt. Moreover, some or all of our debt may be secured by some or all of our assets. If we default on the payment of interest or principal on any such debt, breach any representation or warranty in connection with any borrowing or violate any covenant in any loan document, our lender may accelerate the maturity of such debt, requiring us to immediately repay all outstanding principal.

Distribution Objectives

In order to qualify as a REIT for federal income tax purposes, we must distribute at least 90% of our taxable income (excluding capital gains) to our stockholders. We intend, although we are not legally obligated, to continue to make regular monthly distributions to holders of our common shares in excess of the level required to maintain our REIT status unless our results of operations, our general financial condition, general economic conditions or other factors inhibit us from doing so. Distributions are authorized at the discretion of our board of directors, which is directed, in substantial part, by its obligation to cause us to comply with the REIT requirements of the Internal Revenue Code of 1986, as amended (the "Code").

We declare distributions to our stockholders as of daily record dates and aggregate and pay such distributions monthly. With the authorization of our board of directors, we declared distributions to our stockholders for the period from October 1, 2014 through March 31, 2015. These distributions were calculated based on stockholders of record for each day in an amount equal to \$0.001575342 per share, per day. All distributions were or will be paid in cash or reinvested in shares of our common stock for those participating in our distribution reinvestment plan and will be paid or issued, respectively, on the first business day following the completion of the month to which they relate.

In addition to the distributions described above, our board of directors authorized special stock dividends as of daily record dates for the period from October 1, 2014 through March 31, 2015. Stock dividends for our common shares were calculated based on stockholders of record each day in an amount equal to 0.0000273973 of a share of common stock per share, per day. Shares issued related to our stock dividends were or will be issued on the first business day of the month following the quarter to which they relate.

In our initial quarters of operations, and from time to time thereafter, our cash flows from operations have been and may continue to be insufficient to fund distributions to stockholders. We funded 100% of total distributions for the year ended December 31, 2014 with cash flows from financing activities, which includes offering proceeds. See "Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations — Financial Condition, Liquidity and Capital Resources" for additional information regarding our distributions.

In September 2014, our Advisor agreed to waive the asset management fee otherwise payable to it for the quarter ended December 31, 2014, to the extent that our modified funds from operations ("MFFO") for the quarter ended December 31, 2014, as disclosed in this Annual Report on Form 10-K for such year, amounts to less than 100% of the aggregate distributions declared to our stockholders for such quarter. As a result of this waiver, our Advisor has waived asset management fees of \$16,258 otherwise payable to it for the year ended December 31, 2014, in order to more closely align the amount of distributions paid with our operations. Our advisor has also agreed to waive the asset management fees otherwise payable to it for each of the quarters ended March 31, 2015 and June 30, 2015, to the extent that our MFFO for such quarters, as disclosed in our Quarterly Report on Form 10-Q, amounts to less than 100% of the aggregate distributions declared to our stockholders for and june 30, 2015, to the extent that our MFFO for such quarters, as disclosed in our Quarterly Report on Form 10-Q, amounts to less than 100% of the aggregate distributions declared to our stockholders for each respective quarter. These fee waivers are not deferrals and accordingly, these fees will not be paid to the Advisor in cash at any time in the future.

Tax Status

We intend to make an election to be taxed as a REIT, under Sections 856 through 860 of the Internal Revenue Code of 1986, as amended ("the Code"), effective for our taxable year ended December 31, 2015. In order to qualify as a REIT, an entity must meet certain organizational and operational requirements, including a requirement to distribute at least 90% of its annual ordinary taxable income to stockholders. REITs are generally not subject to federal income tax on taxable income that they distribute to their stockholders. If we fail to qualify as a REIT in any taxable year, we will then be subject to federal income tax on taxable income at regular corporate rates and will not be permitted to qualify for treatment as a REIT for federal income tax purposes for four years following the year during which qualification is lost unless the Internal Revenue Service granted us relief under certain statutory provisions. Such an event could materially adversely affect our net income and net cash available for distribution to stockholders. However, we believe that we have been organized and operated in such a manner as to qualify for treatment as a REIT beginning with our taxable year ended December 31, 2015 and intend to operate in the foreseeable future in such a manner so that we will remain qualified as a REIT for federal income tax purposes. We have not elected to be taxed as a REIT for the taxable year ended December 31, 2014. This did not have an impact on our tax liability of our stockholders that invested in the Company's public offering during 2014 since we had no taxable income for the year ended December 31, 2014.

Competition

Numerous real estate companies, real estate investment trusts and U.S. institutional and foreign investors, including Hines Global REIT, Inc. ("Hines Global I") and Hines Real Estate Investment Trust, Inc. ("Hines REIT") compete with us in acquiring properties or making other real estate investments and obtaining creditworthy tenants to occupy such properties. Many of these entities have significant financial and other resources, allowing them to compete effectively with us. Principal factors of competition in our primary business of acquiring properties or making other real estate investments (including rent and other charges and allowances for inducements and tenant improvements), the quality and breadth of tenant services provided, and reputation as an owner and operator of commercial real estate investments in the relevant market. Additionally, our ability to compete depends upon, among other factors, trends of the global, national and local economies, investment alternatives, financial condition and operating results of current and prospective tenants, availability and cost of capital, taxes, governmental regulations, legislation and demographic trends.

We believe Hines' extensive real estate experience and depth and breadth of its organization of approximately 3,450 employees located in over 100 cities across the United States and 18 foreign countries allows it to better identify investment opportunities for us. However, competition may increase our cost of acquisitions.

Customers

We are dependent upon the ability of current tenants to pay their contractual rent amounts as the rents become due. As described previously, the Company owned one real estate investment as of December 31, 2014, which is a single-tenant industrial property that is 100% leased to Acushnet, through a lease that expires in 2019. As a result, all of the Company's total rental revenue for the year ended December 31, 2014, was earned from this tenant.

Available Information

Stockholders may obtain copies of our filings with the Securities and Exchange Commission ("SEC"), free of charge from the website maintained by the SEC at www.sec.gov or from our website at www.HinesSecurities.com. Further, a copy of this Annual Report on Form 10-K is located at the SEC's Public Reference Room at 100 F Street NE, Washington, D.C. 20549. Information on the operation of the Public Reference Room can be obtained by calling the SEC at 1-800-SEC-0330. Our filings will be available on our website as soon as reasonably practicable after we electronically file such materials with the SEC. However, the information from our website is not incorporated by reference into this report.

Item 1A. Risk Factors

You should carefully read and consider the risks described below, together with all other information in this report. If certain of the following risks actually occur, it could have a material adverse effect on our business, financial condition, and results of operations and our ability to pay distributions would likely suffer materially or could be eliminated entirely. As a result, the value of our common shares may decline, and our stockholders could lose all or part of the money they paid to buy our common shares.

Investment Risks

We have no prior operating history, and the prior performance of other Hines affiliated entities may not be a good measure of our future results; therefore there is a higher risk that we will not be able to achieve our investment objectives compared to a real estate investment trust with an operating history.

We have no prior operating history. As a result, an investment in our shares of common stock may entail more risk than the shares of common stock of a real estate investment trust with an operating history and we may not be able to achieve our investment objectives. In addition, our stockholders should not rely on the past performance of investments by other investment vehicles sponsored by Hines to predict our future results. Our investment strategy and key employees may differ from the investment strategies and key employees of our affiliates in the past, present and future.

We offer a share redemption program for shareholders seeking liquidity of their shares. However, there is no public market for our common shares; therefore, it will be difficult for our stockholders to sell their shares and, if they are able to sell their shares, they will likely sell them at a substantial discount.

There is no public market for our common shares, and we do not expect one to develop. We have a share redemption program, but it is limited in terms of the amount of shares which may be redeemed. It will therefore be difficult for our stockholders to sell their shares of common stock promptly or at all. Additionally, our charter contains restrictions on the ownership and transfer of our shares, and these restrictions may limit the ability of our stockholders to sell their shares. If they are able to sell their shares, they may only be able to sell them at a substantial discount from the price they paid. This may be the result, in part, of the fact that the amount of funds available for investment are reduced by funds used to pay certain up-front fees and expenses, including organization and offering costs, such as issuer costs, selling commissions, and the dealer manager fee and acquisition fees and expenses in connection with our public offerings. Unless our aggregate investments increase in value to compensate for these up-front fees and expenses, which may not occur, it is unlikely that our stockholders will be able to sell their shares, without incurring a substantial loss. Stockholders may also experience substantial losses if we dispose of our assets or in connection with a liquidation event. We cannot assure stockholders that their shares will ever appreciate in value to equal the price they paid for their shares. Thus, prospective stockholders should consider our common shares as illiquid and a long-term investment, and they must be prepared to hold their shares for an indefinite length of time.

This is a fixed price offering and the offering price of our common shares was not established on an independent basis; therefore, as it was arbitrarily determined, the fixed offering price will not accurately represent the current value of our assets at any particular time and may be higher than the value of our assets per share of our common stock at the time of the purchase.

This is a fixed price offering, which means that the prices for our common shares in the offering was fixed and does not vary based on the underlying value of our assets at any time. Our board of directors arbitrarily determined the offering price in its sole discretion. We do not intend to adjust the offering price after we acquire assets and, therefore, the fixed offering price established for our common shares will not accurately represent the value of our assets and the actual value of our stockholders' investment may be substantially less than what they pay. Our offering price may not be indicative of either the price our stockholders would receive if they sold their shares, the price at which shares of our common stock would trade if they were listed on a national securities exchange or if we were liquidated or dissolved. Similarly, the amount our stockholders may receive upon redemption of their shares, if they determine to participate in our share redemption program, may be less than the amount they paid for such shares, regardless of any increase in the underlying value of any assets we own.

Because we are conducting an ongoing offering, we are providing information about our net tangible book value per share. As of December 31, 2014, our net tangible book value per share was \$(2.71), which is less than the offering price for shares of our common stock. Net tangible book value is a rough approximation of value calculated simply as total book value of assets minus total liabilities (all of which are adjusted for noncontrolling interests). It assumes that the value of real estate assets diminishes predictably over time as shown through the depreciation and amortization of real estate investments. Real estate values have historically risen or fallen with market conditions. Net tangible book value is used generally as a conservative measure of net worth that we do not believe reflects our estimated value per share. It is not intended to reflect the value of our assets upon an orderly liquidation of the Company in accordance with our investment objectives. However, net tangible book value does reflect certain dilution in value of our common stock from the issue price as a result of (i) accumulated depreciation and amortization of real estate investments, (ii) the funding of distributions from sources other than our cash flow from operations, (iii) the substantial fees paid in connection with our public offerings, such as selling commissions and marketing fees, all or a portion of which have been re-allowed by our dealer manager to participating broker dealers and (iv) the fees and expenses paid to our advisor and its affiliates in connection with the selection, acquisition, management and sale of our investments.

The Offering is a "blind pool" offering and our stockholders do not have the opportunity to evaluate our future investments prior to purchasing shares of our common stock.

Our stockholders will not be able to evaluate the economic merits, transaction terms or other financial or operational data concerning our future investments prior to purchasing shares of our common stock. In addition, our investment policies and strategies are very broad and permit us to invest in all types of properties and other real estate investments. Our stockholders must rely on our Advisor and our board of directors to implement our investment policies, to evaluate our investment opportunities and to structure the terms of our investments. Because our stockholders cannot evaluate our future investments in advance of purchasing shares of our common stock, a "blind pool" offering may entail more risk than other types of offerings. This additional risk may hinder our stockholders' ability to achieve their personal investment objectives related to portfolio diversification, risk-adjusted investment returns and other objectives.

The Offering is a "best efforts" offering and if we are unable to raise substantial funds, we will be limited in the number and type of investments we may make which could negatively impact an investment in shares of our common stock.

The Offering is being made on a "best efforts" basis, whereby the broker dealers participating in the Offering are only required to use their best efforts to sell shares of our common stock and have no firm commitment or obligation to purchase any of the shares of our common stock. As a result, the amount of proceeds we raise in the Offering may be substantially less than the amount we would need to achieve a diversified industrial portfolio. Our inability to raise substantial funds would increase our fixed operating expenses as a percentage of gross income, and our financial condition and ability to make distributions could be adversely affected. As of the date of this Annual Report on Form 10-K, we have raised approximately \$18.3 million from the sale of shares in the Offering. If we are unable to raise substantially more in the Offering, we will make fewer additional investments in properties, and will more likely focus on making investments in loans and real estate related entities, resulting in less diversification in terms of the number of investments owned, the geographic regions in which our property investments are located and the types of investments that we make. As a result, the likelihood increases that any single investment's poor performance would materially affect our overall investment performance.

The availability and timing of distributions to our stockholders is uncertain and cannot be assured.

There is no assurance that distributions will continue to be authorized and paid. We cannot assure our stockholders that we will have sufficient cash to pay distributions to them or that the amount of any such distributions will increase over time. Should we fail for any reason to distribute at least 90% of our REIT taxable income, we would not qualify for the favorable tax treatment accorded to REITs.

We have and may continue to pay distributions from sources other than our cash flow from operations, including advances, deferrals or waivers of fees from our Advisor or affiliates, borrowings and/or proceeds of the Offering. We have not placed a cap on the amount of our distributions that may be paid from any of these sources. The use of sources other than our cash flow from operations to fund distributions could adversely impact our ability to pay distributions in future periods, decrease the amount of cash we have available for operations and new investments and/or potentially impact the value or result in dilution of our stockholders' investment.

In our initial quarters of operations, and from time to time thereafter, our cash flow from operations may be insufficient to fund distributions to stockholders. Our organizational documents permit us to make distributions from any source and we may choose to pay distributions when we do not have sufficient cash flow from operations to fund such distributions. We may choose to use advances, deferrals or waivers of fees, if available, from our Advisor or affiliates, borrowings and/or proceeds of the Offering or other sources to fund distributions to our stockholders. The Advisor has agreed to waive the asset management fee otherwise payable to it pursuant to our Advisory Agreement for the fourth quarter of 2014 and each of the quarters ended March 31, 2015 and June 30, 2015, to the extent that our MFFO for each respective quarter, as disclosed in our Annual Report on Form 10-Q, as applicable, amounts to less than 100% of the aggregate distributions declared for such quarter.

For example, we funded 100% of total distributions for 2014, with cash flows from financing activities which includes offering proceeds. In addition, the Advisor waived \$16,258 in asset management fees otherwise payable to it for the fourth quarter of 2014. We have not placed a cap on the amount of our distributions that may be paid from sources other than cash flows from operations, including proceeds from our debt financings, proceeds from our public offerings, cash advances by our Advisor and cash resulting from a waiver or deferral of fees. However, the Advisor and affiliates are under no obligation to advance funds to us or to defer or, subsequent to the second quarter of 2015, to continue to waive fees in order to support our distributions. When we pay distributions in excess of earnings and we use cash flows from financing activities, including offering proceeds and borrowings, to fund distributions, then we will have less funds available for operations and for acquiring properties and other investments, which could adversely impact our ability to pay distributions in future periods, may reduce our stockholders' overall return and may result in the dilution of our stockholders' investment. In addition, our Advisor or its affiliates could choose to receive shares of our common stock or interests in the Operating Partnership in lieu of cash or deferred fees or the repayment of advances to which they are entitled, and the issuance of such securities may dilute our stockholders' interest in us. Furthermore, to the extent distributions exceed cash flow from operations, a stockholder's basis in our stockholder is basis, the stockholder may recognize capital gain.

Payments to the holders of the Special OP Units or any other OP Units will reduce cash available for distribution to our stockholders.

An affiliate of Hines has received OP Units in return for its \$190,000 contribution to the Operation Partnership. Our Advisor or its affiliates may also choose to receive OP Units in lieu of certain fees. The holders of all OP Units will be entitled to receive cash from operations pro rata with the distributions being paid to us and such distributions to the holder of the OP Units will reduce the cash available for distribution to our stockholders. In addition, Hines Global REIT II Associates Limited Partnership, the holder of the Special OP Units, will be entitled to cash distributions, under certain circumstances, including from sales of our real estate investments, refinancings and other sources, which may reduce cash available for distribution to our stockholders and may negatively affect the value of our shares of common stock. Furthermore, under certain circumstances the Special OP Units and any other OP Units held by Hines or its affiliates are required to be repurchased, in cash at the holder's election and there may not be sufficient cash to make such a repurchase payment; therefore, we may need to use cash from operations, borrowings, or other sources to make the payment, which will reduce cash available for distribution to our stockholders.

Our stockholders' ability to have their shares redeemed is limited under our share redemption program, and if they are able to have their shares redeemed, it may be at a price that is less than the price they paid for the shares and the thencurrent market value of the shares.

Our share redemption program contains significant restrictions and limitations. For example, only stockholders who purchase their shares directly from us or who received their shares through a non-cash transaction, not in the secondary market, are eligible to participate, and stockholders must generally hold their shares for a minimum of one year before they can participate in our share redemption program. In addition, our share redemption program generally provides that only funds received from the prior month's distribution reinvestment plan may be used in the subsequent month to redeem shares. Our board of directors may terminate, suspend or amend the share redemption program upon 30 days' written notice without stockholder approval. As a result of these limitations, the redemption price our stockholders may receive upon any such redemption may not be indicative of the price our stockholders would receive if our shares were actively traded or if we were liquidated, and our stockholders should not assume that they will be able to sell all or any portion of their shares back to us pursuant to our share redemption program or to third parties at a price that reflects the then current market value of the shares or at all.

The actual value of shares that we redeem under our share redemption program may be substantially less than what we pay.

Under our share redemption program, shares currently may be repurchased at varying prices depending on the number of years the shares have been held and whether the redemptions are sought upon a stockholder's death or disability. Although the offering price for the shares represents the most recent price at which most investors are willing to purchase such shares, it will not accurately represent the current value of our assets per share of our common stock at any particular time and may be higher or lower than the actual value of our assets per share at such time. Accordingly, when we repurchase shares of our common stock at the offering price or at a percentage of the offering price, the actual value of the shares that we repurchase may be less, and, if so, then the repurchase will be dilutive to our remaining stockholders.

Our stockholders will not have the benefit of an independent due diligence review in connection with the Offering and, if a conflict of interest arises between us and Hines, we may incur additional fees and expenses.

Because our Advisor and our Dealer Manager are affiliates of Hines, our stockholders will not have the benefit of an independent due diligence review and investigation of the type normally performed by an unaffiliated, independent underwriter in connection with a securities offering. In addition, Greenberg Traurig, LLP has acted as counsel to us, our Advisor and our Dealer Manager in connection with the Offering and, therefore, investors will not have the benefit of a due diligence review and investigation that might otherwise be performed by independent counsel which increases the risk of their investment. If any situation arises in which our interests are in conflict with those of our Dealer Manager or its affiliates, and we are required to retain additional counsel, we will incur additional fees and expenses.

The fees we pay in connection with the Offering and the agreements entered into with Hines and its affiliates were not determined on an arm's-length basis and therefore may not be on the same terms we could achieve from a third party.

The compensation paid to our Advisor, Dealer Manager, Hines and other affiliates for services they provide us was not determined on an arm's-length basis. All service agreements, contracts or arrangements between or among Hines and its affiliates, including our Advisor and us, were not negotiated at arm's-length. Such agreements include our Advisory Agreement, our Dealer Manager Agreement, and any property management and leasing agreements. A third party unaffiliated with Hines may be willing and able to provide certain services to us at a lower price.

We will pay substantial compensation to Hines, our Advisor and their affiliates, which may be increased during the Offering or future offerings by our independent directors.

Subject to limitations in our charter, the fees, compensation, income, expense reimbursements, interests and other payments payable to Hines, our Advisor and their affiliates may increase during the Offering or in the future if such increase is approved by a majority of our independent directors.

We do not, and do not expect to, have research analysts reviewing our performance.

We do not, and do not expect to, have research analysts reviewing our performance or our securities on an ongoing basis. Therefore, our stockholders will not have an independent review of our performance and the value of our common stock relative to publicly traded companies.

Investors who invest in us at the beginning of our offering may realize a lower rate of return than later investors.

We expect to have little, if any, cash flow from operations available for distribution until we make a substantial amount of investments. In addition, to the extent our investments are in development or redevelopment projects or in properties that have significant capital requirements, our ability to pay distributions may be negatively impacted, especially during our early periods of operation. Until such time as we have sufficient cash flow from operations, we may not be able to make, or may be limited in the amount that we can pay towards, distributions. As a result, investors who invest in us before we make a substantial amount of real estate investments or generate significant cash flow may realize a lower rate of return than later investors.

Our stockholders may experience dilution.

Our stockholders do not have preemptive rights. If we engage in a subsequent offering of common shares or securities convertible into common shares, issue shares pursuant to our distribution reinvestment plan or otherwise issue additional shares, investors who purchase shares in the Offering who do not participate in those other stock issuances will experience dilution in their percentage ownership of our outstanding shares. Furthermore, stockholders may experience a dilution in the value of their shares depending on the terms and pricing of any share issuances (including the shares being sold in the Offering) and the value of our assets at the time of issuance.

The price of our common shares may be adjusted to a price less than the price our shareholders paid for their shares.

The price of our common shares may be adjusted periodically in the discretion of our board of directors and therefore any future adjustments may result in an offering price lower than the price our shareholders paid for their shares.

We will be required to disclose an estimated net asset value per share of our common stock prior to or following the conclusion of the Offering and the purchase price our shareholders pay for shares of our common stock in the Offering may be higher than such estimated net asset value per share. The estimated net asset value per share may not be an accurate reflection of the fair market value of our assets and liabilities and likely will not represent the amount of net proceeds that would result if we were liquidated or dissolved.

We will be required, due to rule amendments that have been adopted by the Financial Industry Regulatory Authority, or FINRA, and take effect on April 11, 2016 and due to contractual obligations in the selling agreements between our participating broker dealers and the Dealer Manager, to disclose an estimated net asset value per share of our shares. We presently expect to provide an estimated net asset value per share no later than November 15, 2016. If we have not provided an estimated net asset value per share prior to April 11, 2016, our stockholders' customer account statements will include a value per share that is less than the Offering price for such shares in the Offering, because the amended FINRA rules will require the "value" on the customer account statement to be equal to the Offering price per share less up-front underwriting compensation and organization and offering expenses. Once the amended rules take effect, we expect to use this "value" from customer account statements as the deemed estimated per share value for purposes of reports to fiduciaries of retirement plans. Further, if we provide an estimated net asset value per share prior to the conclusion of the Offering, our board of directors may determine to modify the offering price, including the prices at which the shares are offered pursuant to our distribution reinvestment plan, but the offering price may differ from the estimated net asset value per share. Any estimated net asset value per share that we disclose in the future may not be an accurate reflection of the fair value of our assets and liabilities in accordance with GAAP, may not reflect the price at which we would be able to sell all or substantially all of our assets or the outstanding shares of our common stock in an arm's-length transaction, may not represent the value that stockholders could realize upon a sale of the company or upon the liquidation of our assets and settlement of our liabilities, and may not be indicative of the price at which shares of our common stock would trade if they were listed on a national securities exchange. In addition, any estimated net asset value per share that we disclose in the future may not be the equivalent of the disclosure of a market price by an openended real estate fund.

The methodology used to determine the estimated net asset value per share of our common stock may be based upon assumptions, estimates and judgments that may not be accurate or complete, such that, if different property-specific and general real estate and capital market assumptions, estimates and judgments were used, it could result in an estimated net asset value per share that is significantly different.

Risks Related to Our Business in General

Delays in purchasing properties or making other real estate investments with the proceeds received from the Offering may result in a lower rate of return to investors.

Our ability to locate and commit to purchase specific properties, or make investments, will be partially dependent on our ability to raise sufficient funds for such acquisitions and investments. We may be substantially delayed in making investments due to delays in:

- the sale of our common shares,
- obtaining debt financing,
- negotiating or obtaining the necessary purchase documentation,
- locating suitable investments, or
- other factors.

We expect to invest proceeds we receive from the Offering in short-term, highly-liquid investments until we use such funds in our operations. We expect that the income we earn on these temporary investments will not be substantial. Further, we may use the principal amount of these investments, and any returns generated on these investments, to pay for fees and expenses in connection with the Offering and distributions. Therefore, delays in investing proceeds we raise from the Offering could impact our ability to generate cash flow for distributions.

A prolonged national or world-wide economic downturn or volatile capital market conditions could adversely affect our results of operations and our ability to pay distributions to our stockholders.

If disruptions in the capital and credit markets occur again, as have been experienced during recent years, they could adversely affect our ability to obtain loans, credit facilities, debt financing and other financing, or, when available, to obtain such financing on reasonable terms, which could negatively impact our ability to implement our investment strategy.

If these disruptions in the capital and credit markets should occur again as a result of, among other factors, uncertainty, changing or increased regulation, reduced alternatives or additional failures of significant financial institutions, our access to liquidity could be significantly impacted. Prolonged disruptions could result in us taking measures to conserve cash until the markets stabilize or until alternative credit arrangements or other funding for our business needs could be arranged. Such measures could include deferring investments, reducing or eliminating the number of shares redeemed under our share redemption program and reducing or eliminating distributions we make to our stockholders.

We believe the risks associated with our business are more severe during periods of economic downturn if these periods are accompanied by declining values in real estate. For example, a prolonged economic downturn could negatively impact our property investments as a result of increased customer delinquencies and/or defaults under our leases, generally lower demand for rentable space, potential oversupply of rentable space leading to increased concessions, and/or customer improvement expenditures, or reduced rental rates to maintain occupancies.

Our operations could be negatively affected to a greater extent if an economic downturn occurs again, is prolonged or becomes more severe, which could significantly harm our revenues, results of operations, financial condition, liquidity, business prospects and our ability to make distributions to our stockholders and may result in a decrease in the value of our stockholders' investment.

Yields on and safety of deposits may be lower due to the extensive decline in the financial markets.

Until we invest the proceeds of the Offering in real properties and other real estate investments, we may hold those funds in investments, including money market funds, bank money market accounts and CDs or other accounts at third-party depository institutions. Continued or unusual declines in the financial markets could result in a loss of some or all of these funds. In particular, money market funds have recently experienced intense redemption pressure and have had difficulty satisfying redemption requests. As a result, we may not be able to access the cash in our money market investments. In addition, current yields from these investments are minimal.

The failure of any bank in which we deposit our funds could reduce the amount of cash we have available to pay distributions and make additional investments.

The Federal Deposit Insurance Corporation only insures amounts up to \$250,000 per depositor. It is likely that we will have cash and cash equivalents and restricted cash deposited in certain financial institutions in excess of federally insured levels. If any of the banking institutions in which we deposit funds ultimately fails, we may lose any amounts of our deposits over federally insured levels. The loss of our deposits could reduce the amount of cash we have available to distribute or invest and could result in a decline in the value of our stockholders' investment.

Because of our inability to retain earnings, we will rely on debt and equity financings for acquisitions, and if we do not have sufficient capital resources from such financings, our growth may be limited.

If we obtain qualification as a REIT, we are required to distribute to our stockholders at least 90% of our annual ordinary taxable income to maintain such qualification. This requirement limits our ability to retain income or cash flow from operations to finance the acquisition of new investments. We will explore acquisition opportunities from time to time with the intention of expanding our operations and increasing our profitability. We anticipate that we will use debt and equity financing for such acquisitions because of our inability to retain significant earnings. Consequently, if we cannot obtain debt or equity financing on acceptable terms, our ability to acquire new investments and expand our operations will be adversely affected.

We may need to incur borrowings that would otherwise not be incurred to meet REIT minimum distribution requirements.

In order to qualify as a REIT, we are required to distribute to our stockholders at least 90% of our annual ordinary taxable income. In addition, we will be subject to a 4% nondeductible excise tax on the amount, if any, by which certain distributions paid (or deemed paid) by us with respect to any calendar year are less than the sum of (i) 85% of our ordinary income for that year, (ii) 95% of our capital gain net income for that year and (iii) 100% of our undistributed taxable income from prior years.

We expect our income, if any, to consist almost solely of our share of the Operating Partnership's income, and the cash available for the payment of distributions by us to our stockholders will consist of our share of cash distributions made by the Operating Partnership. As the general partner of the Operating Partnership, we will determine the amount of any distributions made by the Operating Partnership. However, we must consider a number of factors in making such distributions, including:

- the amount of the cash available for distribution;
- the impact of such distribution on other partners of the Operating Partnership;
- the Operating Partnership's financial condition;
- the Operating Partnership's capital expenditure requirements and reserves therefore; and
- the annual distribution requirements contained in the Code necessary to qualify and maintain our qualification as a REIT.

Differences in timing between the actual receipt of income and actual payment of deductible expenses and the inclusion of such income and deduction of such expenses when determining our taxable income, as well as the effect of nondeductible capital expenditures, the creation of reserves, the use of cash to purchase shares under our share redemption program or required debt amortization payments, could result in our having taxable income that exceeds cash available for distribution.

In view of the foregoing, we may be unable to meet the REIT minimum distribution requirements and/or avoid the 4% excise tax described above. In certain cases, we may decide to borrow funds in order to meet the REIT minimum distribution and/or avoid the 4% excise tax even if our management believes that the then prevailing market conditions generally are not favorable for such borrowings or that such borrowings would not be advisable in the absence of such tax considerations.

Lenders may require us to enter into restrictive covenants that relate to or otherwise limit our operations, which could limit our ability to make distributions to our stockholders, to replace the Advisor or to otherwise achieve our investment objectives.

When providing financing, a lender may impose restrictions on us that affect our distribution and operating policies and our ability to incur additional debt. Loan documents we enter into may contain covenants that limit our ability to further mortgage property, discontinue insurance coverage, or make distributions under certain circumstances. In addition, provisions of our loan documents may deter us from replacing the Advisor because of the consequences under such agreements and may limit our ability to replace the property manager or terminate certain operating or lease agreements related to the property. These or other limitations may adversely affect our flexibility and our ability to achieve our investment objectives.

Actions of our joint venture partners, including other Hines investment vehicles and third parties, could negatively impact our performance.

We may purchase or develop properties or other real estate investments or make investments in joint ventures or partnerships, co-tenancies or other co-ownership arrangements with Hines affiliates, the sellers of the properties, developers or similar persons. Joint ownership of properties or other investments, under certain circumstances, may involve risks not otherwise present with other methods of owning real estate or other real estate investments. Examples of these risks include:

- the possibility that our partners or co-investors might become insolvent or bankrupt;
- that such partners or co-investors might have economic or other business interests or goals that are inconsistent with
 our business interests or goals, including inconsistent goals relating to the sale of properties or other investments held
 in the joint venture or the timing of the termination and liquidation of the venture;
- the possibility that we may incur liabilities as the result of actions taken by our partners or co-investors; or
- that such partners or co-investors may be in controlling positions and/or be in a position to take actions contrary to our instructions or requests or contrary to our policies or objectives, including our policy with respect to qualifying and maintaining our qualification as a REIT.

Actions by a co-venturer, co-tenant or partner may result in subjecting the assets of the joint venture to unexpected liabilities. Under joint venture arrangements, neither co-venturer may have the power to control the venture, and under certain circumstances, an impasse could result and this impasse could have an adverse impact on the operations and profitability of the joint venture.

If we have a right of first refusal or buy/sell right to buy out a co-venturer or partner, we may be unable to finance such a buy-out if it becomes exercisable or we are required to purchase such interest at a time when it would not otherwise be in our best interest to do so. If our interest is subject to a buy/sell right, we may not have sufficient cash, available borrowing capacity or other capital resources to allow us to elect to purchase an interest of a co-venturer subject to the buy/sell right, in which case we may be forced to sell our interest as the result of the exercise of such right when we would otherwise prefer to keep our interest. Finally, we may not be able to sell our interest in a joint venture if we desire to exit the venture for any reason or if our interest is likewise subject to a right of first refusal of our co-venturer or partner, our ability to sell such interest may be adversely impacted by such right. Joint ownership arrangements with Hines affiliates may also entail conflicts of interest.

If we invest in a limited partnership as a general partner, we could be responsible for all liabilities of such partnership.

In some joint ventures or other investments we may make, if the entity in which we invest is a limited partnership, we may acquire all or a portion of our interest in such partnership as a general partner. As a general partner, we could be liable for all the liabilities of such partnership. Additionally, we may acquire a general partner interest in the form of a non-managing general partner interest. As a non-managing general partner, we are potentially liable for all liabilities of the partnership without having the same rights of management or control over the operation of the partnership as the managing general partner. Therefore, we may be held responsible for all of the liabilities of an entity in which we do not have full management rights or control, and our liability may far exceed the amount or value of investment we initially made or then had in the partnership.

We may acquire various financial instruments for purposes of "hedging" or reducing our risks, which may be costly and ineffective and may reduce our cash available for distribution to our stockholders.

We may enter into currency rate swaps and caps, or similar hedging or derivative transactions or arrangements, in order to manage or mitigate our risk of exposure to the effects of currency changes as a result of our international investments. Similarly, we may enter into interest rate swaps and caps, or similar hedging or derivative transactions or arrangements, in order to manage or mitigate our risk of exposure to the effects of interest rate changes due to variable interest rate debt that we may have.

We are different in some respects from other investment vehicles sponsored by Hines, and therefore the past performance of such investments may not be indicative of our future results and Hines has limited experience in acquiring and operating certain types of real estate investments that we may acquire.

We are Hines' third publicly-offered real estate investment vehicle. We collectively refer to real estate joint ventures, funds and programs as real estate investment vehicles. All but two of the previous real estate investment vehicles of Hines and its affiliates were conducted through privately-held entities not subject to either the up-front commissions, fees and expenses

associated with the Offering or all of the laws and regulations that govern us, including reporting requirements under the federal securities laws and tax and other regulations applicable to REITs.

The past performance of other investment vehicles sponsored by Hines or its affiliates may not be indicative of our future results, and we may not be able to successfully operate our business and implement our investment strategy, which may be different in a number of respects from the operations previously conducted by Hines. In addition, Hines has limited experience in acquiring and operating certain types of real estate investments that we may acquire, as a significant amount of real estate investments that have been made by Hines' other investment vehicles have consisted of acquisitions and development of office or industrial properties or land. We may therefore need to use third parties to source or manage investments in which Hines has limited experience. In addition, a significant portion of Hines' other programs and investments involve development projects. Although we are able to invest in development projects. As a result of all of these factors, our stockholders should not rely on the past performance of other investment vehicles sponsored by Hines and its affiliates to predict, or as an indication of, our future performance.

Our success will be dependent on the performance of Hines as well as key employees of Hines. Certain other investment vehicles sponsored by Hines have experienced adverse developments in recent years and there is a risk that we may experience similar adverse developments.

Our ability to achieve our investment objectives and to pay distributions is dependent upon the performance of Hines and its affiliates as well as key employees of Hines in the identification and acquisition of investments, the selection of tenants, the determination of any financing arrangements, the management of our assets and operation of our day-to-day activities. Our board of directors and our Advisor have broad discretion when identifying, evaluating, making and managing our investments with the proceeds of the Offering. Our stockholders will have no opportunity to evaluate the terms of transactions or other economic or financial data concerning our investments. We will rely on the management ability of Hines and the oversight of our board of directors as well as the management of any entities or ventures in which we invest.

We may not be able to retain our key employees. To the extent we are unable to retain and/or find qualified successors for key employees that depart from the company, our results of operations may be adversely impacted. Our officers and the management of our Advisor also serve in similar capacities for numerous other entities. If Hines (or any of its key employees) is distracted by these other activities or suffers from adverse financial or operational problems in connection with its operations unrelated to us, the ability of Hines and its affiliates to allocate time and/or resources to our operations may be adversely affected. If Hines is unable to allocate sufficient resources to oversee and perform our operations for any reason, our results of operations would be adversely impacted. We will not provide key-man life insurance policies for any of Hines' key employees.

Certain other investment vehicles sponsored by Hines have experienced adverse developments in recent years. Although it was re-opened with respect to ordinary redemption requests in April 2013, Hines REIT suspended its share redemption program, except with respect to redemptions in connection with the death or disability of a stockholder in December 2009. As of December 31, 2014, shares redeemed pursuant to the share redemption program were redeemed at \$5.45 per share, with respect to ordinary redemption requests, and \$6.40 per share with respect to requests in connection with the death or disability of a stockholder. In May 2011, November 2012, April 2013, November 2013 and December 2014, Hines REIT's board of directors determined an estimated per share net asset value, or NAV, of \$7.78, \$7.61, \$6.75, \$6.40 and \$6.50, respectively, each of which was lower than the most recent primary offering price of \$10.08. The reduction in the estimated NAV between November 2012 and April 2013 was due to Hines REIT's payment to its stockholders of special distributions in excess of \$0.80 per share (all of which represented a return of capital). In addition, Hines REIT decreased its distribution rate in July 2010 and further decreased the rate in April 2013.

In addition to Hines REIT, Hines Global REIT, Inc., and HMS Income Fund, Inc. ("HMS"), Hines has sponsored more than 20 privately-offered programs in the past ten years. Several of Hines' privately-offered programs have experienced adverse economic developments in recent years due to the global financial crisis and deteriorating economic conditions in several European and South American countries, Mexico and several U.S. markets. The adverse market conditions may cause these programs to alter their investment strategy, generate returns lower than originally expected, or ultimately incur losses. There is a risk that we may experience similar adverse developments, as an investment vehicle sponsored by Hines.

Terrorist attacks and other acts of violence, civilian unrest or war may affect the markets in which we operate, our business and our profitability.

Terrorist attacks and other acts of violence, civilian unrest or war may negatively affect our operations and our stockholders' investment in our shares. We may acquire real estate investments located in or that relate to real estate located in

areas that are susceptible to attack. In addition, any kind of terrorist activity or violent criminal acts, including terrorist acts against public institutions or buildings or modes of public transportation (including airlines, trains or buses) could have a negative effect on our business. These events may directly impact the value of our assets through damage, destruction, loss or increased security costs. We may not be able to obtain insurance against the risk of terrorism because it may not be available or may not be available on terms that are economically feasible. Further, even if we do obtain terrorism insurance, we may not be able to obtain sufficient coverage to fund any losses we may incur. Risks associated with potential acts of terrorism in the areas in which we acquire properties or other real estate investments could sharply increase the premiums we pay for coverage against property and casualty claims. Additionally, mortgage lenders in some cases have begun to insist that specific coverage against terrorism be purchased by commercial owners as a condition for providing loans.

The consequences of any armed conflict are unpredictable, and we may not be able to foresee events that could have an adverse effect on our business or our stockholders' investment. More generally, any terrorist attack, other act of violence or war, including armed conflicts, could result in increased volatility in or damage to, the United States and worldwide financial markets and economy. They also could result in a continuation of the current economic uncertainty in the United States or abroad. Our revenues will be dependent upon the payment of rent and the return of our other investments which may be particularly vulnerable to uncertainty in the local economy. Increased economic volatility could adversely affect our tenants' ability to pay rent or the return on our other investments or our ability to borrow money or issue capital stock at acceptable prices and have a material adverse effect on our business, results of operations, cash flows and financial condition and our ability to make distributions to our stockholders and the value of their investment.

Risks Related to Investments in Real Estate

Geographic concentration of our portfolio may make us particularly susceptible to adverse economic developments in the real estate markets of those areas.

In the event that we have a concentration of properties in, or real estate investments that invest in properties located in, a particular geographic area, our operating results and ability to make distributions are likely to be impacted by economic changes affecting the real estate markets in that area. An investment in our common stock will therefore be subject to greater risk to the extent that we lack a geographically diversified portfolio. As of March 25, 2015, based on the estimated aggregate value of our real estate investments, approximately 20% of our portfolio consists of a property located in Carlsbad, California and approximately 80% of our portfolio consists of a property located in Dublin, Ireland.

Industry concentration of our tenants may make us particularly susceptible to adverse economic developments in these industries.

In the event we have a concentration of tenants in a particular industry, our operating results and ability to make distributions may be adversely affected by adverse developments in those industries and we will be subject to a greater risk to the extent that our tenants are not diversified by industry. For example, based on leased square footage, as of March 25, 2015, approximately 51% of our portfolio is leased to tenants in the manufacturing industry, approximately 33% is leased to tenants in the government sector, and approximately 10% is leased to tenants in the finance and insurance industry.

We will depend on tenants for our revenue, and therefore our revenue will be dependent on the success and economic viability of our tenants. Our reliance on single or significant tenants in certain buildings may decrease our ability to lease vacated space.

We expect that rental income from real property will, directly or indirectly, constitute a significant portion of our income. Delays in collecting accounts receivable from tenants could adversely affect our cash flows and financial condition. In addition, the inability of a single major tenant or a number of smaller tenants to meet their rental obligations would adversely affect our income. Therefore, our financial success will be indirectly dependent on the success of the businesses operated by the tenants in our properties or in the properties securing loans we may own. The weakening of the financial condition or the bankruptcy or insolvency of a significant tenant or a number of smaller tenants and vacancies caused by defaults of tenants or the expiration of leases, may adversely affect our operations and our ability to pay distributions.

Generally, under U.S. bankruptcy law, a debtor tenant has 120 days to exercise the option of assuming or rejecting the obligations under any unexpired lease for nonresidential real property, which period may be extended once by the bankruptcy court. If the tenant assumes its lease, the tenant must cure all defaults under the lease and may be required to provide adequate assurance of its future performance under the lease. If the tenant rejects the lease, we will have a claim against the tenant's bankruptcy estate. Although rent owing for the period between filing for bankruptcy and rejection of the lease may be afforded administrative expense priority and paid in full, pre-bankruptcy arrears and amounts owing under the remaining term of the

lease will be afforded general unsecured claim status (absent collateral securing the claim). Moreover, amounts owing under the remaining term of the lease will be capped. Other than equity and subordinated claims, general unsecured claims are the last claims paid in a bankruptcy and therefore funds may not be available to pay such claims in full. In addition, while the specifics of the bankruptcy laws of international jurisdictions may differ from the U.S. bankruptcy laws described herein, the bankruptcy or insolvency of a significant tenant or a number of smaller tenants at any of the international properties we may acquire, may similarly adversely impact our operations and our ability to pay distributions.

Some of our properties may be leased to a single or significant tenant and, accordingly, may be suited to the particular or unique needs of such tenant. We may have difficulty replacing such a tenant if the floor plan of the vacant space limits the types of businesses that can use the space without major renovation. In addition, the resale value of the property could be diminished because the market value of a particular property will depend principally upon the value of the leases of such property.

Due to the risks involved in the ownership of real estate investments and real estate acquisitions, a return on an investment in us is not guaranteed, and our stockholders may lose some or all of their investment.

By owning our shares, stockholders will be subjected to significant risks associated with owning and operating real estate investments. The performance of their investment in us will be subject to such risks, including:

- changes in the general economic climate;
- changes in local conditions such as an oversupply of space or reduction in demand for real estate;
- changes in interest rates and the availability of financing;
- changes in property level operating expenses due to inflation or otherwise;
- changes in laws and governmental regulations, including those governing real estate usage, zoning and taxes; and
- changes due to factors that are generally outside of our control, such as terrorist attacks and international instability, natural disasters and acts of God, over-building, adverse national, state or local changes in applicable tax, environmental or zoning laws and a taking of any of the properties which we own or in which we otherwise have interests by eminent domain.

In addition, we expect to acquire properties in the future, which may subject us to additional risks associated with real estate property acquisitions, including the risks that:

- the investments will fail to perform in accordance with our expectations because of conditions or liabilities we did not know about at the time of acquisition; and
- our projections or estimates with respect to the performance of the investments, the costs of operating or improving the properties or the effect of the economy or capital markets on the investments will prove inaccurate.

Any of these factors could have a material adverse effect on our business, results of operations, cash flows and financial condition and our ability to make distributions to our stockholders and the value of their investment.

An economic slowdown or rise in interest rates or other unfavorable changes in economic conditions in the markets in which we operate could adversely impact our business, results of operations, cash flows and financial condition and our ability to make distributions to our stockholders and the value of their investment.

The development of negative economic conditions in the markets in which we operate may significantly affect occupancy, rental rates and our ability to collect rent from our tenants, as well as our property values, which could have a material adverse impact on our cash flows, operating results and carrying value of investment property. For example, an economic recession or rise in interest rates could make it more difficult for us to lease real properties, may require us to lease the real properties we acquire at lower rental rates and may lead to an increase in tenant defaults. In addition, these conditions may also lead to a decline in the value of our properties and make it more difficult for us to dispose of these properties at an attractive price. Other risks that may affect conditions in the markets in which we operate include:

- Local conditions, such as an oversupply of the types of properties we invest in or a reduction in demand for such properties in the area; and
- Increased operating costs, if these costs cannot be passed through to tenants.

International, national, regional and local economic climates have been adversely affected by the slow job growth of recent years. To the extent any of the adverse conditions described above occurs in the specific markets in which we operate, market rents, occupancy rates and our ability to collect rents from our tenants will likely be affected and the value of our

properties may decline. We could also face challenges related to adequately managing and maintaining our properties, should we experience increased operating cost and as a result, we may experience a loss of rental revenues. Any of these factors may adversely affect our business, results of operations, cash flows and financial condition, our ability to make distributions to our stockholders and the value of their investment.

Our use of borrowings to partially fund acquisitions and improvements on properties could result in foreclosures and unexpected debt service expenses upon refinancing, both of which could have an adverse impact on our operations and cash flow.

We are relying and intend to continue to rely in part on borrowings under credit facilities and other external sources of financing to fund the costs of new investments, capital expenditures and other items. Accordingly, we are subject to the risks that our cash flow will not be sufficient to cover required debt service payments and that we will be unable to meet other covenants or requirements in the credit agreements.

If we cannot meet our required debt obligations, the property or properties securing such indebtedness could be foreclosed upon by, or otherwise transferred to, our lender, with a consequent loss of income and asset value to us. For tax purposes, a foreclosure of any of our properties would be treated as a sale of the property for a purchase price equal to the outstanding balance of the debt secured by the mortgage. If the outstanding balance of the debt secured by the mortgage exceeds our tax basis in the property, we would recognize taxable income on foreclosure, but we may not receive any cash proceeds. Additionally, we may be required to refinance our debt subject to "lump sum" or "balloon" payment maturities on terms less favorable than the original loan or at a time we would otherwise prefer to not refinance such debt. A refinancing on such terms or at such times could increase our debt service payments, which would decrease the amount of cash we would have available for operations, new investments and distribution payments and may cause us to determine to sell one or more properties at a time when we would not otherwise do so.

Uninsured losses relating to real property may adversely impact the value of our portfolio.

We will attempt to ensure that all of our properties are adequately insured to cover casualty losses. However, there are types of losses, generally catastrophic in nature, which are uninsurable, are not economically insurable or are only insurable subject to limitations. Examples of such catastrophic events include acts of war or terrorism, earthquakes, floods, hurricanes and pollution or environmental matters. We may not have adequate coverage in the event we or our buildings suffer casualty losses. If we do not have adequate insurance coverage, the value of our assets will be reduced as the result of, and to the extent of, any such uninsured losses. Additionally, we may not have access to capital resources to repair or reconstruct any uninsured damage to a property.

We may be unable to obtain desirable types of insurance coverage at a reasonable cost, if at all, and we may be unable to comply with insurance requirements contained in mortgage or other agreements due to high insurance costs.

We may not be able either to obtain certain desirable types of insurance coverage, such as terrorism, earthquake, flood, hurricane and pollution or environmental matter insurance, or to obtain such coverage at a reasonable cost in the future, and this risk may limit our ability to finance or refinance debt secured by our properties. Additionally, we could default under debt or other agreements if the cost and/or availability of certain types of insurance make it impractical or impossible to comply with covenants relating to the insurance we are required to maintain under such agreements. In such instances, we may be required to self-insure against certain losses or seek other forms of financial assurance.

The real estate industry is subject to extensive regulation, which may result in higher expenses or other negative consequences that could adversely affect us.

Our activities are subject to federal, state and municipal laws, and to regulations, authorizations and license requirements with respect to, among other things, zoning, environmental protection and historical heritage, all of which may affect our business. We may be required to obtain licenses and permits with different governmental authorities in order to acquire and manage our assets.

In addition, the Dodd-Frank Wall Street Reform and Consumer Protection Act (the "Dodd-Frank Act"), which generally took effect in 2011, contains a sweeping overhaul of the regulation of U.S. financial institutions and financial markets. Key provisions of the Dodd-Frank Act require extensive rulemaking by the SEC and the U.S. Commodity Futures Trading Commission, some of which remains ongoing. Thus, the full impact of the Dodd-Frank Act on our business cannot be fully assessed until all final implementing rules and regulations are promulgated.

Various rules currently in effect under the Dodd-Frank Act may have a significant impact on our business, including, without limitation, provisions of the legislation that increase regulation of and disclosure requirements related to investment advisors, swap transactions and hedging policies, corporate governance and executive compensation, investor protection and enforcement provisions, and asset-backed securities.

For example, but not by way of limitation, the Dodd-Frank Act and the rules and regulations promulgated thereunder provides for significantly increased regulation of the derivatives markets and transactions that affect our interest rate hedging activities, including: (i) regulatory reporting, (ii) subject to limited exemptions, mandated clearing through central counterparties and execution on regulated exchanges or execution facilities, and (iii) margin and collateral requirements. While the full impact of the Dodd-Frank Act on our interest rate hedging activities cannot be fully assessed until all final implementing rules and regulations, may increase our costs in entering into such transactions, and/or may result in us entering into such transactions on less favorable terms than prior to the Dodd-Frank Act. For example, subject to an exception for "end-users" of swaps upon which we may seek to rely, we may be required to clear certain interest rate hedging transactions by submitting them to a derivatives clearing organization. To the extent we are required to clear any such transactions, we will be required to, among other things, post margin in connection with such transactions. The occurrence of any of the foregoing events may have an adverse effect on our business and our stockholders' return.

In addition, public authorities may enact new and more stringent standards, or interpret existing laws and regulations in a more restrictive manner, which may force companies in the real estate industry, including us, to spend funds to comply with these new rules. Any such action on the part of public authorities may adversely affect our results from operations.

In the event of noncompliance with such laws, regulations, licenses and authorizations, we may face the payment of fines, project shutdowns, cancellation of licenses, and revocation of authorizations, in addition to other civil and criminal penalties.

We operate in a competitive business, and many of our competitors have significant resources and operating flexibility, allowing them to compete effectively with us.

Numerous real estate companies that operate in the markets in which we may operate will compete with us in acquiring real estate investments and obtaining creditworthy tenants to occupy such properties or the properties owned by such investments. Such competition could adversely affect our business. There are numerous real estate companies, real estate investment trusts and U.S. institutional and foreign investors that will compete with us in seeking investments and tenants for properties, including Hines REIT and Hines Global I. Many of these entities have significant financial and other resources, including operating experience, allowing them to compete effectively with us. In addition, our ability to charge premium rental rates to tenants may be negatively impacted. This increased competition may increase our costs of acquisitions or investments or lower our occupancy rates and the rent we may charge tenants. In addition, the arrival of new competitors in the immediate areas where we have assets could require unplanned investments in our assets, which may adversely affect us. We may also have difficulty in renewing leases or in leasing to new tenants, which may lead to a reduction in our cash flow and operating income, since the proximity of new competitors could divert existing or new tenants to such competitors, resulting in vacancies.

We may have difficulty selling real estate investments, and our ability to distribute all or a portion of the net proceeds from such sales to our stockholders may be limited.

Real estate investments are relatively illiquid. We will have a limited ability to vary our portfolio in response to changes in economic or other conditions. We will also have a limited ability to sell assets in order to fund working capital and similar capital needs such as share redemptions. We expect to generally hold a real estate investment for the long term. When we sell any of our real estate investments, we may not realize a gain on such sale or the amount of our taxable gain could exceed the cash proceeds we receive from such sale. We may not distribute any proceeds from the sale of real estate investments to our stockholders. Rather, we may use such proceeds to:

- purchase additional real estate investments;
- repay debt;
- buy out interests of any co-venturers or other partners in any joint venture in which we are a party;
- purchase shares under our share redemption program;
- create working capital reserves; or
- make repairs, maintenance, tenant improvements or other capital improvements or expenditures to our other properties.

Our ability to sell our properties may also be limited by our need to avoid a 100% penalty tax that is imposed on gain recognized by a REIT from the sale of property characterized as dealer property. In order to avoid such characterization and to

take advantage of certain safe harbors under the Code, we may determine to hold our properties for a minimum period of time, generally two years.

Potential liability as the result of, and the cost of compliance with, environmental matters could adversely affect our operations.

Under various federal, state and local environmental laws, ordinances and regulations, a current or previous owner or operator of real property may be liable for the cost of removal or remediation of hazardous or toxic substances on such property. Such laws often impose liability whether or not the owner or operator knew of, or was responsible for, the presence of such hazardous or toxic substances.

We expect to invest in, or make investments in real estate investments that have interests in, properties historically used for industrial, manufacturing and commercial purposes. These properties are more likely to contain, or may have contained, underground storage tanks for the storage of petroleum products and other hazardous or toxic substances. All of these operations create a potential for the release of petroleum products or other hazardous or toxic substances. Leasing properties to tenants that engage in industrial, manufacturing, and commercial activities will cause us to be subject to increased risk of liabilities under environmental laws and regulations. The presence of hazardous or toxic substances, or the failure to properly remediate these substances, may adversely affect our ability to sell, rent or pledge such property as collateral for future borrowings.

Environmental laws also may impose restrictions on the manner in which properties may be used or businesses may be operated, and these restrictions may require expenditures. Such laws may be amended so as to require compliance with stringent standards which could require us to make unexpected, substantial expenditures. Environmental laws provide for sanctions in the event of noncompliance and may be enforced by governmental agencies or, in certain circumstances, by private parties. We may be potentially liable for such costs in connection with the acquisition and ownership of our properties in the United States. In addition, we may invest in properties located in countries that have adopted laws or observe environmental management standards that are less stringent than those generally followed in the United States, which may pose a greater risk that releases of hazardous or toxic substances have occurred to the environment. The cost of defending against claims of liability, compliance with environmental regulatory requirements or remediating any contaminated property could be substantial and require a material portion of our cash flow.

We face possible risks associated with the physical effects of climate change.

We cannot predict with certainty whether climate change is occurring and, if so, at what rate. However, the physical effects of climate change could have a material adverse effect on our properties, operations and business. To the extent climate change causes changes in weather patterns, our markets could experience increases in storm intensity, such as those experienced in Super Storm Sandy in October 2012, and rising sea-levels. Over time, these conditions could result in declining demand for office space in our buildings or the inability of us to operate the buildings at all. Climate change may also have indirect effects on our business by increasing the cost of (or making unavailable) property insurance on terms we find acceptable, increasing the cost of energy and increasing the cost of snow removal at our properties. There can be no assurance that climate change will not have a material adverse effect on our properties, operations or business.

The properties we acquire will be subject to property taxes that may increase in the future, which could adversely affect our cash flow.

Any properties we acquire will be subject to real and personal property taxes that may increase as property tax rates change and as the properties are assessed or reassessed by taxing authorities. We anticipate that most of our leases will generally provide that the property taxes, or increases therein, are charged to the lessees as an expense related to the properties that they occupy. As the owner of the properties, however, we are ultimately responsible for payment of the taxes to the government. If property taxes increase, our tenants may be unable to make the required tax payments, ultimately requiring us to pay the taxes. In addition, we will generally be responsible for property taxes related to any vacant space. If we purchase residential properties, the leases for such properties typically will not allow us to pass through real estate taxes and other taxes to residents of such properties. Consequently, any tax increases may adversely affect our results of operations at such properties.

Our properties may contain or develop harmful mold, which could lead to liability for adverse health effects and costs of remediating the problem.

If any of our properties has or develops mold we may be required to undertake a costly program to remediate, contain or remove the mold. Mold growth may occur when moisture accumulates in buildings or on building materials. Some molds may produce airborne toxins or irritants. Concern about indoor exposure to mold has been increasing because exposure to mold may cause a variety of adverse health effects and symptoms, including allergic or other reactions. We may become liable to our tenants, their employees and others if property damage or health concerns arise, all of which could have a material adverse effect on our business, results of operations, cash flows and financial condition and our ability to make distributions to our stockholders and the value of their investment.

If we set aside insufficient working capital reserves, we may be required to defer necessary or desirable property improvements.

If we do not establish sufficient reserves for working capital to supply necessary funds for capital improvements or similar expenses, we may be required to defer necessary or desirable improvements to our properties. If we defer such improvements, the applicable properties may decline in value, it may be more difficult for us to attract or retain tenants to such properties or the amount of rent we can charge at such properties may decrease.

Risks related to the development of real properties may have an adverse effect on our results of operations and returns to our stockholders.

We may invest in properties on which developments or improvements are to be constructed or completed. As such, we may be subject to the risks associated with development and construction activities including the following:

- long periods of time may elapse between the commencement and the completion of our projects;
- our original estimates may not be accurate and our actual construction and development costs may exceed those estimates;
- the level of interest of potential tenants for a recently launched development may be low;
- construction materials and equipment may be unavailable or cost more than expected due to changes in supply and demand;
- construction and sales may not be completed on time, resulting in a cost increase;
- we may not be able to acquire or we may pay too much for the land we acquire for new developments or properties;
- labor may be in limited availability; and
- changes in tax, real estate and zoning laws may be unfavorable to us.

In addition, our reputation and the construction quality of our real estate developments, whether operated individually or through partnerships, may be determining factors for our ability to lease space and grow. The timely delivery of real estate projects and the quality of our developments, however, depend on certain factors beyond our full control, including the quality and timeliness of construction materials delivered to us and the technical capabilities of our contractor. If one or more problems affect our real estate developments, our reputation and future performance may be negatively affected and we may be exposed to civil liability.

We depend on a variety of factors outside of our control to build, develop and operate real estate projects. These factors include, among others, the availability of market resources for financing, land acquisition and project development. Any scarcity of market resources, including human capital, may decrease our development capacity due to either difficulty in obtaining credit for land acquisition or construction financing or a need to reduce the pace of our growth. The combination of these risks may adversely affect our business, results of operations, cash flows and financial condition and our ability to make distributions to our stockholders and the value of their investment.

Delays in the development and construction of real properties may have adverse effects on portfolio diversification, results of operations and returns to our stockholders.

If we experience delays in the development of our real properties, it could adversely affect returns to our stockholders. When properties are acquired prior to the start of construction or during the early stages of construction, it will typically take several months or longer to complete construction, to rent available space, and for rent payments to commence. Therefore, we may not receive any income from these properties and our ability to pay distributions to our stockholders could suffer. If we are delayed in the completion of any such construction project, our tenants may have the right to terminate preconstruction leases for space at such newly developed project. We may incur additional risks when we make periodic progress payments or other

advances to builders prior to completion of construction. Each of these factors could result in increased costs of a project or loss of our investment. In addition, we will be subject to normal lease-up risks relating to newly-constructed projects. Furthermore, the price we agree to pay for a real property will be based on our projections of rental income and expenses and estimates of the fair market value of the real property upon completion of construction. If our projections are inaccurate, we may pay too much for a property.

Changes in supply of or demand for similar properties in a particular area may increase the price of real estate assets we may seek to purchase or adversely affect the value of the properties we own.

The real estate industry is subject to market forces and we are unable to predict certain market changes including changes in supply of or demand for similar properties in a particular area. For example, if demand for the types of real estate assets in which we seek to invest were to sharply increase or supply of those assets were to sharply decrease, the prices of those assets could rise significantly. Any potential purchase of an overpriced asset could decrease our rate of return on these investments and result in lower operating results and overall returns to you. Likewise, a sharp increase in supply could adversely affect lease rates and occupancy, which could result in lower operating results and overall returns to you.

Retail properties depend on anchor tenants to attract shoppers and could be adversely affected by the loss of a key anchor tenant.

We may acquire retail properties in the future. Retail properties, like other properties, are subject to the risk that tenants may be unable to make their lease payments or may decline to extend a lease upon its expiration. A lease termination by a tenant that occupies a large area of a retail center (commonly referred to as an anchor tenant) could impact leases of other tenants. Other tenants may be entitled to modify the terms of their existing leases in the event of a lease termination by an anchor tenant, or the closure of the business of an anchor tenant that leaves its space vacant even if the anchor tenant continues to pay rent. Any such modifications or conditions could be unfavorable to us as the property owner and could decrease rents or expense recoveries. Additionally, major tenant closures may result in decreased customer traffic, which could lead to decreased sales at other stores. In the event of default by a tenant or anchor store, we may experience delays and costs in enforcing our rights as landlord to recover amounts due to us under the terms of our agreements with those parties.

Leases with retail properties' tenants may restrict us from re-leasing space.

Most leases with retail tenants contain provisions giving the particular tenant the exclusive right to sell particular types of merchandise or provide specific types of services within the particular retail center. These provisions may limit the number and types of prospective tenants interested in leasing space in a particular retail property.

Recent disruptions in the financial markets could adversely affect the multifamily property sector's ability to obtain financing and credit enhancement from Fannie Mae and Freddie Mac, which could adversely impact us if we attempt to secure financing through Fannie Mae and Freddie Mac for investments in multifamily residential properties.

We may invest in multifamily residential properties and development projects. Fannie Mae and Freddie Mac are major sources of financing for the multifamily sector. Since 2007, Fannie Mae and Freddie Mac have reported substantial losses and a need for significant amounts of additional capital. In response to the deteriorating financial condition of Fannie Mae and Freddie Mac and the recent credit market disruption, the U.S. Congress and Treasury undertook a series of actions to stabilize these government-sponsored enterprises and the financial markets. Pursuant to legislation enacted in 2008, the U.S. government placed both Fannie Mae and Freddie Mac under its conservatorship.

Currently, Fannie Mae and Freddie Mac remain active multifamily lenders but there is significant uncertainty surrounding their futures. Should Fannie Mae and Freddie Mac have their mandates changed or reduced, be disbanded or reorganized by the government or otherwise discontinue providing liquidity to the multifamily sector, it would significantly reduce our access to such debt capital and/or increase borrowing costs. If new U.S. government regulations heighten Fannie Mae's and Freddie Mac's underwriting standards, adversely affect interest rates and reduce the amount of capital they can make available to the multifamily sector, it could have a material adverse effect on both the multifamily sector and our access to funding in connection with the acquisition and maintenance of such properties. Any potential reduction in loans, guarantees and credit-enhancement arrangements from Fannie Mae and Freddie Mac could jeopardize the effectiveness of the multifamily sector's derivative securities market, potentially causing breaches in loan covenants, and through reduced loan availability, impact the value of multifamily assets, which could impair the value of a significant portion of multifamily communities.

Specifically, the potential for a decrease in liquidity made available to the multifamily sector by Fannie Mae and Freddie Mac could:

- make it more difficult for us to secure new takeout financing for any multifamily development projects we acquire;
- hinder our ability to refinance any completed multifamily assets;
- decrease the amount of available liquidity and credit that could be used to diversify our portfolio through the acquisition of multifamily assets; and
- require us to obtain other sources of debt capital with potentially different terms.

Short-term multifamily community leases associated with any multifamily residential properties we acquire may expose us to the effects of declining market rent and could adversely impact our ability to make cash distributions to our stockholders.

We expect that, to the extent that we invest in any multifamily residential properties, substantially all of our multifamily community leases will be for a term of one year or less. Because these leases generally permit the residents to leave at the end of the lease term without penalty, our rental revenues may be impacted by declines in market rents more quickly than if our leases were for longer terms.

High levels of unemployment could adversely affect the occupancy and rental rates of any multifamily residential properties we acquire, with high quality multifamily communities suffering even more severely.

Increased levels of unemployment in multifamily markets could significantly decrease occupancy and rental rates. In times of increasing unemployment, multifamily occupancy and rental rates have historically been adversely affected by:

- rental residents deciding to share rental units and therefore rent fewer units;
- · potential residents moving back into family homes or delaying leaving family homes;
- a reduced demand for higher-rent units, such as those of high quality multifamily communities;
- a decline in household formation;
- persons enrolled in college delaying leaving college or choosing to proceed to or return to graduate school in the absence of available employment;
- the inability or unwillingness of residents to pay rent increases; and
- increased collection losses.

These factors generally have contributed to lower rental rates. To the extent that we invest in any multifamily residential properties, our results of operations, financial condition and ability to make distributions to our stockholders may be adversely affected if these factors worsen.

In connection with the recent credit market disruptions and economic slowdown, any investments in multifamily residential properties may face increased competition from single-family homes and condominiums for rent, which could limit our ability to retain residents, lease apartment units or increase or maintain rents.

To the extent that we invest in any multifamily residential properties, we will face significant competition with respect to our investments in multifamily residential properties. Any multifamily communities in which we invest may compete with numerous housing alternatives in attracting residents, including single-family homes and condominiums available for rent. Such competitive housing alternatives may become more prevalent in a particular area because of the tightening of mortgage lending underwriting criteria, homeowner foreclosures, the decline in single-family home and condominium sales and the lack of available credit. The number of single-family homes and condominiums for rent in a particular area could limit our ability to retain residents, lease apartment units or increase or maintain rents.

If we acquire hospitality or leisure properties, we will depend on others to manage those facilities.

In order to qualify as a REIT, we will not be able to operate any hospitality or leisure properties that we acquire or participate in the decisions affecting the daily operations of these properties. We will lease any hospitality or leisure properties we acquire to a taxable REIT subsidiary, or TRS, in which we may own up to a 100% interest. Our TRS will enter into management agreements with eligible independent contractors, potentially including Hines or its affiliates, that are not our subsidiaries or otherwise controlled by us to manage these properties. Thus, independent operators, under management agreements with our TRS, will control the daily operations of our hospitality, leisure and healthcare-related properties.

We will depend on these independent management companies to operate our hospitality or leisure properties. We will not have the authority to require these properties to be operated in a particular manner or to govern any particular aspect of the daily operations, such as establishing room rates at our hospitality or leisure properties. Thus, even if we believe our hospitality or leisure properties are being operated inefficiently or in a manner that does not result in satisfactory results, we may not be able to force the management company to change its method of operation of these properties. We can only seek redress if a management company violates the terms of the applicable management agreement with the TRS, and then only to the extent of the remedies provided for under the terms of the management agreement. In the event that we need to replace any management company, we may be required by the terms of the management agreement to pay substantial termination fees and may experience significant disruptions at the affected properties.

The hospitality or leisure industry is seasonal.

The hospitality or leisure industry is seasonal in nature. As a result of the seasonality of the hospitality or leisure industry, there will likely be quarterly fluctuations in results of operations of any hospitality or leisure properties that we may own. Quarterly financial results may be adversely affected by factors outside our control.

The hospitality or leisure market is highly competitive and generally subject to greater volatility than our other market segments.

The hospitality or leisure business is highly competitive and influenced by factors such as location, room rates, quality, service levels, reputation and reservation systems, among many other factors. There are many competitors in this market, and these competitors may have substantially greater marketing and financial resources than those available to us. This competition, along with other factors, such as over-building in the hospitality or leisure industry and certain deterrents to traveling, may increase the number of rooms available and may decrease the average occupancy and room rates of our hospitality or leisure properties. The demand for rooms at any hospitality or leisure properties that we may acquire will change much more rapidly than the demand for space at other properties that we acquire. This volatility in room demand and occupancy rates could have a material adverse effect on our financial condition, results of operations and ability to pay distributions to our stockholders.

If we purchase assets at a time when the commercial real estate market is experiencing substantial influxes of capital investment and competition for properties, the real estate we purchase may not appreciate or may decrease in value.

Real estate investment transaction volume has increased and estimated going-in capitalization rates, or cap rates (ratio of the net projected operating income of a property in its initial fiscal year divided by the net purchase price), have fallen relative to their post-recession peaks in late 2009. There continues to be a significant amount of investment capital pursuing high-quality, well-located assets that generate stable cash flows, causing aggressive competition and pricing for assets which match our investment strategy. This may continue to drive prices higher, resulting in lower cap rates and returns. To the extent we purchase real estate in the future in this environment, we are subject to the risks that the value of our assets may not appreciate or may decrease significantly below the amount we paid for such assets if the real estate market ceases to attract the same level of capital investment in the future as it has recently attracted, or if the number of companies seeking to acquire such assets decreases. If any of these circumstances occur or the values of our investments are otherwise negatively affected, the value of our stockholders' investment may be lower.

Risks Related to Investments in Debt

Hines does not have substantial experience investing in mortgage, mezzanine, bridge or construction loans, B Notes, securitized debt or other debt related to properties in which we may invest, which could adversely affect our return on our loan investments.

We may make investments in mortgage, mezzanine, bridge or construction loans, B-Notes, securitized debt or other debt related to properties if our Advisor determines that it is advantageous to us due to the state of the real estate market or in order to diversify our investment portfolio. However neither our Advisor nor any of its affiliates has any substantial experience investing in these types of loans and we may not have the expertise necessary to maximize the return on our investment in these types of loans.

If we make or invest in loans, our loans may be impacted by unfavorable real estate market conditions, which could decrease the value of our loan investments.

If we make or invest in loans, we will be at risk of default by the borrowers on those loans. These defaults may be caused by many conditions beyond our control, including interest rate levels and local and other economic conditions affecting real estate values. We may invest in unsecured loans. Even with respect to loans secured by real property, we will not know whether the values of the properties securing the loans will remain at the levels existing on the dates of origination of the loans. If the values of such underlying properties drop, our risk will increase with respect to secured loans because of the lower value of the security associated with such loans.

If we make or invest in loans, our loans will be subject to interest rate fluctuations, which could reduce our returns as compared to market interest rates as well as the value of the loans in the event we sell the loans.

If we invest in fixed-rate, long-term loans and interest rates rise, the loans could yield a return that is lower than thencurrent market rates. If interest rates decrease, we will be adversely affected to the extent that loans are prepaid, because we may not be able to make new loans at the previously higher interest rate. If we invest in variable interest rate loans, if interest rates decrease, our revenues will likewise decrease. Finally, if interest rates increase, the value of fixed-rate loans we own at such time would decrease which would lower the proceeds we would receive in the event we sell such assets.

Delays in liquidating defaulted loans could reduce our investment returns.

If there are defaults under our loans secured by real property, we may not be able to repossess and sell the underlying properties quickly. The resulting time delay could reduce the value of our investment in the defaulted loans. An action to foreclose on a property securing a loan is regulated by state statutes and rules and is subject to many of the delays and expenses of other lawsuits if the defendant raises defenses or counterclaims. In the event of default by a borrower, these restrictions, among other things, may impede our ability to foreclose on or sell the secured property or to obtain proceeds sufficient to repay all amounts due to us on the loan.

We may make or invest in mezzanine loans, which involve greater risks of loss than senior loans secured by real properties.

We may make or invest in mezzanine loans that generally take the form of subordinated loans secured by second mortgages on the underlying real property or loans secured by a pledge of the ownership interests of an entity that directly or indirectly owns real property. These types of investments involve a higher degree of risk than long-term senior mortgage loans secured by real property because the investment may become unsecured as a result of foreclosure by the senior lender. In the event of a bankruptcy of the entity providing the pledge of its ownership interests as security, we may not have full recourse to the assets of such entity, or the assets of the entity may not be sufficient to satisfy our mezzanine loan. If a borrower defaults on our mezzanine loan or debt senior to our mezzanine loan, or in the event of a borrower bankruptcy, our mezzanine loan will be satisfied only after the senior debt. As a result, we may not recover some or all of our investment. In addition, mezzanine loans may have higher loan-to-value ratios than traditional mortgage loans, resulting in less equity in the real property and increasing our risk of loss of principal.

We may invest in B-Notes, which are subject to additional risks as a result of the privately negotiated structure and terms of such transactions which may result in losses.

We may invest in B-Notes, which are typically secured by a first mortgage on a single large commercial property or group of related properties and subordinated to an A-Note secured by the same first mortgage on the same collateral. If a borrower defaults on a B-Note, A-Note holders would be paid first and there may not be sufficient funds remaining to repay us and other B-Note holders. B-Notes can vary in their structural characteristics and risks because each transaction is privately negotiated. For example, the rights of holders of B-Notes to control the process following a borrower default may be limited in certain investments. We cannot predict the terms of each B-Note investment. Moreover, because B-Notes are typically secured by a single property or group of related properties, such investments may not be as diversified as investments secured by a pool of properties and therefore may be subject to increased risks.

Bridge loans may involve a greater risk of loss than conventional mortgage loans.

We may provide bridge loans secured by first lien mortgages on properties to borrowers who are typically seeking shortterm capital in connection with acquisitions, developments or refinancings of real estate. In connection with such investments, there is a risk that the borrower may not achieve its investment objectives and that we may therefore not recover some or all of our investment in such bridge loans. For example, if we provide a bridge loan to a borrower who has identified an undervalued asset, either due to mismanagement of the underlying assets or as a result of what the borrowers deem to be a recovering market, and the market in which such asset is located fails to recover according to the borrower's projections, or if the borrower fails to improve the quality of the asset's management or the value of the asset, the borrower may not receive a sufficient return on the asset to satisfy the bridge loan.

In addition, owners usually borrow funds under a conventional mortgage loan to repay a bridge loan. If the borrower is unable to obtain permanent financing to repay our bridge loan, we may lose some or all of our investment. Bridge loans are also subject to risks of borrower defaults, bankruptcies, fraud, losses and special hazard losses that are not covered by standard hazard insurance. In the event we make a bridge loan to a borrower who defaults, we bear the risk of loss of principal and nonpayment of interest and fees to the extent of any deficiency between the value of the mortgage collateral and the principal amount of the bridge loan. To the extent we suffer such losses with respect to our investments in bridge loans, it could adversely impact our business, results of operations, cash flows and financial ability and our ability to make distributions to our stockholders and value of their investment.

Non-conforming and non-investment grade loans are subject to an increased risk of loss.

Loans we may acquire or originate may not conform to conventional loan criteria applied by traditional lenders and may not be rated or may be rated as "non-investment grade." Non-investment grade ratings for these loans typically result from the overall leverage of the loans, the lack of a strong operating history for the properties underlying the loans, the borrowers' credit history, the properties' underlying cash flow or other factors. Therefore, non-conforming and investment loans we acquire or originate may have a higher risk of default and loss than conventional loans. Any loss we incur may adversely impact our business, results of operations, cash flows and financial ability and our ability to make distributions to our stockholders and value of their investment.

We may invest in commercial mortgage-backed securities, or CMBS, which are subject to all of the risks of the underlying mortgage loans and the additional risks of the securitization process.

CMBS are securities that evidence interests in, or are secured by, a single commercial mortgage loan or a pool of commercial mortgage loans. In a rising interest rate environment, the value of CMBS may be adversely affected when payments on underlying mortgages do not occur as anticipated, resulting in the extension of the security's effective maturity and the related increase in interest rate sensitivity of a longer-term instrument. The value of CMBS may also change due to shifts in the market's perception of issuers and regulatory or tax changes adversely affecting the mortgage securities market as a whole. In addition, CMBS are subject to the credit risk associated with the performance of the underlying mortgage properties.

The securitization process CMBS go through may also result in additional risks. Generally, CMBS are issued in classes similar to mortgage loans. To the extent that we invest in a subordinate class, we will be paid interest only to the extent that there are funds available after paying the senior classes. To the extent the collateral pool includes delinquent loans, subordinate classes will likely not be fully paid and may not be paid at all. Subordinate CMBS are also subject to greater credit risk than those CMBS that are more highly rated. Further, the ratings assigned to any particular class of CMBS may not ultimately prove to be accurate. Thus, any particular class of CMBS may be riskier and more volatile than the rating assigned to such security which may result in the returns on any such CMBS investment to be less than anticipated.

Our debt investments may be considered illiquid and we may not be able to adjust our portfolio in response to changes in economic and other conditions.

The debt investments we may make in connection with privately negotiated transactions may not be registered under the relevant securities laws, resulting in a prohibition against their transfer, sale, pledge or other disposition except in a transaction that is exempt from the registration requirements of, or is otherwise in accordance with, those laws. As a result, our ability to vary our portfolio in response to changes in economic and other conditions may be relatively limited. The mezzanine loans we may purchase in the future will be particularly illiquid investments due to their short life, their unsuitability for securitization and the greater difficulty of recoupment in the event of a borrower's default.

Risks Related to International Investments

We are subject to additional risks from our international investments.

We expect to continue to purchase real estate investments located in, or related to assets located in, the United States and internationally, and may make or purchase loans or participations in loans secured by property located outside the United

States. These investments may be affected by factors particular to the laws and business practices of the jurisdictions in which the properties are located. These laws and business practices may expose us to risks that are different from and in addition to those commonly found in the United States. Foreign investments are subject to risk, including the following risks:

- the burden of complying with a wide variety of foreign laws;
- changing governmental rules and policies, including changes in land use and zoning laws, more stringent environmental laws or changes in such environmental laws;
- existing or new laws relating to the foreign ownership of real property or loans and laws restricting the ability of
 foreign persons or companies to remove profits earned from activities within the country to the person's or company's
 country of origin;
- the potential for expropriation;
- possible currency transfer restrictions;
- imposition of adverse or confiscatory taxes;
- · changes in real estate and other tax rates and changes in other operating expenses in particular countries;
- possible challenges to the anticipated tax treatment of the structures that allow us to acquire and hold investments;
- adverse market conditions caused by terrorism, civil unrest and changes in national or local governmental or economic conditions;
- the willingness of domestic or foreign lenders to make loans in certain countries and changes in the availability, cost and terms of loan funds resulting from varying national economic policies;
- general political and economic instability in certain regions;
- the potential difficulty of enforcing obligations in other countries; and
- Hines' limited experience and expertise in foreign countries relative to its experience and expertise in the United States.

Investments in properties or other real estate investments outside the United States subject us to foreign currency risks, which may adversely affect distributions and our REIT status.

Revenues generated from any properties or other real estate investments we acquire or ventures we enter into relating to transactions involving assets located in markets outside the United States likely will be denominated in the local currency. Therefore, any investments we make outside the United States may subject us to foreign currency risk due to potential fluctuations in exchange rates between foreign currencies and the U.S. dollar. As a result, changes in exchange rates of any such foreign currency to the U.S. dollar may affect our revenues, operating margins and distributions and may also affect the book value of our assets and the amount of stockholders' equity.

Changes in foreign currency exchange rates used to value a REIT's foreign assets may be considered changes in the value of the REIT's assets. These changes may adversely affect our status as a REIT. Further, bank accounts in foreign currency which are not considered cash or cash equivalents may adversely affect our status as a REIT.

Inflation in foreign countries, along with government measures to curb inflation, may have an adverse effect on our investments.

Certain countries have in the past experienced extremely high rates of inflation. Inflation, along with governmental measures to curb inflation, coupled with public speculation about possible future governmental measures to be adopted, has had significant negative effects on these international economies in the past and this could occur again in the future. The introduction of governmental policies to curb inflation can have an adverse effect on our business. High inflation in the countries in which we purchase real estate or make other investments could increase our expenses and we may not be able to pass these increased costs on to our tenants.

Lack of compliance with the United States Foreign Corrupt Practices Act, or FCPA, could subject us to penalties and other adverse consequences.

We are subject to the FCPA, which generally prohibits United States companies from engaging in bribery or other prohibited payments to foreign officials for the purpose of obtaining or retaining business. Foreign companies, including potential competitors, are not subject to these prohibitions. Fraudulent practices, including corruption, extortion, bribery, payoffs, theft and others, occur from time-to-time in countries in which we may do business. If people acting on our behalf or at our request are found to have engaged in such practices, severe penalties and other consequences could be imposed on us that may have a material adverse effect on our business, results of operations, cash flows and financial condition and our ability to make distributions to our stockholders and the value of their investment.

Risks Related to Organizational Structure

Any interest in us will be diluted by the Special OP Units and any other OP Units in the Operating Partnership and any interest in us may be diluted if we issue additional shares.

We owned a 95.2% general partner interest in the Operating Partnership as of December 31, 2014. Hines Global REIT II Associates Limited Partnership owns the remaining interest in the Operating Partnership, including the Special OP Units in the Operating Partnership, which were issued as consideration for an obligation by Hines and its affiliates to perform future services in connection with our real estate operations. Please see "Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations—Critical Accounting Policies—Treatment of Management Compensation and Expense Reimbursements" for a summary of these interests. Payments with respect to these interests will reduce the amount of distributions that would otherwise be payable to our stockholders in the future.

Stockholders do not have preemptive rights to acquire any shares issued by us in the future. Therefore, investors purchasing our common shares may experience dilution of their equity investment if we:

- sell shares in the Offering or sell additional shares in the future, including those issued pursuant to our distribution reinvestment plan;
- sell securities that are convertible into shares, such as OP Units;
- at the option of our Advisor, issue OP Units to pay for certain fees;
- issue OP Units or common shares to our Advisor or affiliates in exchange for advances or deferrals of fees;
- issue shares in a private offering; or
- issue shares to sellers of properties acquired by us in connection with an exchange of partnership units from the Operating Partnership.

The repurchase of interests in the Operating Partnership held by Hines and its affiliates (including the Special OP Units and other OP Units) as required in our Advisory Agreement may discourage a takeover attempt.

Under certain circumstances, including a merger, consolidation or sale of substantially all of our assets or any similar transaction, a purchase, tender or exchange offer that results in the tender or exchange of more than 50% of our outstanding common shares, a transaction pursuant to which a majority of our board of directors then in office are replaced or removed, or the termination or non-renewal of our Advisory Agreement, including termination by us for cause or non-renewal for poor performance, the Operating Partnership is, at the election of Hines or its affiliates, required to purchase the Special OP Units and any OP Units that Hines or its affiliates own for cash (or, in certain cases, a promissory note) or our shares, at the election of the holder. These rights may deter these types of transactions which may limit the opportunity for stockholders to receive a premium for their common shares that might otherwise exist if an investor attempted to acquire us.

Hines' ability to cause the Operating Partnership to purchase the Special OP Units and any other OP Units that it or its affiliates hold in connection with the termination of our Advisory Agreement may deter us from terminating our Advisory Agreement.

Under certain circumstances, if we are not advised by an entity affiliated with Hines, including under circumstances where the Advisory Agreement is terminated by us for cause or not renewed due to poor performance, Hines or its affiliates may cause the Operating Partnership to purchase some or all of the Special OP Units or any other OP Units then held by such entities. Under these circumstances if the amount necessary to purchase Hines' and its affiliates' interests in the Operating Partnership is substantial, these rights could discourage or deter us from terminating our Advisory Agreement under circumstances in which we would otherwise do so.

We have authorized stock dividends and may issue preferred shares or separate classes or series of common shares, which issuance could adversely affect the holders of our common shares.

Holders of our common stock do not have preemptive rights to any shares issued by us in the future. With the authorization of our board of directors, we declared special daily stock dividends for the period from October 1, 2014 through March 31, 2015, which may dilute the value of our shares. In addition, we may issue, without stockholder approval, preferred shares or a class or series of common shares with rights that could adversely affect the holders of our shares. Upon the affirmative vote of a majority of our directors (including, in the case of preferred shares, a majority of our independent directors), our charter authorizes our board of directors (without any further action by our stockholders) to issue preferred shares or common shares in one or more classes or series, and to fix the voting rights (subject to certain limitations), liquidation preferences, distribution rates, conversion rights, redemption rights and terms, including sinking fund provisions, and certain

other rights and preferences with respect to such classes or series of shares. If we ever create and issue preferred shares with a distribution preference over common shares, payment of any distribution preferences of outstanding preferred shares would reduce the amount of funds available for the payment of distributions on the common shares. Further, holders of preferred shares are normally entitled to receive a preference payment in the event we liquidate, dissolve or wind up before any payment is made to the common stockholders, likely reducing the amount common stockholders would otherwise receive upon such an occurrence. We could also designate and issue shares in a class or series of common shares with similar rights. In addition, under certain circumstances, the issuance of preferred shares or a separate class or series of common shares may render more difficult or tend to discourage:

- a merger, tender offer or proxy contest;
- the assumption of control by a holder of a large block of our securities; and/or
- the removal of incumbent management.

Our board of directors determines our major policies and operations which increases the uncertainties faced by our stockholders.

Our board of directors determines our major policies, including our policies regarding acquisitions, dispositions, financing, growth, debt capitalization, REIT qualification, redemptions and distributions. Our board of directors may amend or revise these and other policies without a vote of the stockholders. Under the Maryland General Corporation Law and our charter, our stockholders have a right to vote only on limited matters. Our board of directors' broad discretion in setting policies and the inability of our stockholders to exert control over those policies increases the uncertainty and risks they face, especially if our board of directors and our stockholders disagree as to what course of action is in the best interests of our stockholders.

The ownership limit in our charter may discourage a takeover attempt.

Our charter provides that no holder of shares, other than any person to whom our board of directors grants an exemption, may directly or indirectly own more than 9.9% of the number or value, whichever is more restrictive, of the aggregate of our outstanding shares or more than 9.9% of the number or value, whichever is more restrictive, of the outstanding shares of any class or series of our outstanding securities. This ownership limit may deter tender offers for our common shares, which offers may be attractive to our stockholders, and thus may limit the opportunity for stockholders to receive a premium for their common shares that might otherwise exist if an investor attempted to assemble a block of common shares in excess of 9.9% of the number or value, whichever is more restrictive, of the aggregate of our outstanding shares, or 9.9% in number or value, whichever is more restrictive, of the outstanding common shares or otherwise to effect a change of control in us.

We will not be afforded the protection of the Maryland General Corporation Law relating to business combinations.

Provisions of the Maryland General Corporation Law prohibit business combinations, unless prior approval of the board of directors is obtained before the person seeking the combination became an interested stockholder, with:

- any person who beneficially owns 10% or more of the voting power of our outstanding voting shares (an "interested stockholder");
- any of our affiliates or associates who, at any time within the two-year period prior to the date in question, was the beneficial owner of 10% or more of the voting power of our then outstanding shares (also an "interested stockholder"); or
- an affiliate of an interested stockholder.

These prohibitions are intended to prevent a change of control by interested stockholders who do not have the support of our board of directors. Because our charter contains limitations on ownership of more than 9.9% of our common shares, our board of directors has adopted a resolution presently opting out of the business combinations statute. Therefore, we will not be afforded the protections of this statute and, accordingly, there is no guarantee that the ownership limitations in our charter will provide the same measure of protection as the business combinations statute and prevent an undesired change of control by an interested stockholder.

We are not registered as an investment company under the Investment Company Act of 1940, as amended, or the Investment Company Act, and therefore we will not be subject to the requirements imposed on an investment company by the Investment Company Act which may limit or otherwise affect our investment choices.

Hines Global II, the Operating Partnership, and our subsidiaries will conduct our businesses so that none of such entities are required to register as "investment companies" under the Investment Company Act. The operation of a business in a manner so as not to be subject to regulation as an investment company requires an analysis of and compliance with complex laws, regulations and SEC staff interpretations, not all of which are summarized herein. Although we could modify our business methods at any time, at the present time we expect that the focus of our activities will involve investments in real estate, buildings, and other assets that can be referred to as "sticks and bricks" and therefore we will not be an investment company under Section 3(a)(1)(A) of the Investment Company Act. We may invest in other real estate investments and will otherwise be considered to be in the real estate business.

Companies subject to the Investment Company Act are required to comply with a variety of substantive requirements such as requirements relating to:

- limitations on the capital structure of the entity;
- restrictions on certain investments;
- prohibitions on transactions with affiliated entities; and
- public reporting disclosures, record keeping, voting procedures, proxy disclosure and similar corporate governance rules and regulations.

These and other requirements are intended to provide benefits and/or protections to security holders of investment companies. Because we and our subsidiaries do not expect to be subject to these requirements, our stockholders will not be entitled to these benefits or protections. It is our policy to operate in a manner that will not require us to register as an investment company, and we do not expect or intend to register as an "investment company" under the Investment Company Act.

We do not expect that we, the Operating Partnership, or other subsidiaries will be an investment company because we intend to ensure that holdings of investment securities by each such entity will not exceed 40% of the total assets of that entity and that no such entity holds itself out as being engaged primarily in the business of investing in securities. If an entity were to hold investment securities having a value exceeding 40% of the value of the entity's total assets, and no other exclusion from registration was available, that entity might be required to register as an investment company. In order to avoid such a result, we, the Operating Partnership, or a subsidiary may be unable to sell assets we would otherwise want to sell or may need to sell assets we would otherwise wish to retain. In addition, we may also have to forgo opportunities to acquire certain investments or interests in companies or entities that we would otherwise want to acquire, or acquire assets we might otherwise not select for purchase.

If Hines Global II, the Operating Partnership or any subsidiary owns assets that qualify as "investment securities" and the value of such assets exceeds 40% of the value of its total assets, the entity would be deemed to be an investment company absent another exclusion from the Investment Company Act. Certain of the subsidiaries that we may form in the future could seek to rely upon the exclusion provided by Section 3(c)(5)(C) of that Act, which is available for, among other things, entities "primarily engaged in the business of purchasing or otherwise acquiring mortgages and other liens on and interests in real estate." This exclusion, as interpreted by the staff of the SEC, generally requires that at least 55% of an entity's portfolio be comprised of qualifying interests and the remaining 45% of the entity's portfolio be comprised primarily of real estate-type interests (as such terms have been interpreted by the staff of the SEC). SEC staff no-action letters have indicated that the foregoing real estate-type interests test will be met if at least 25% of such entity's assets are invested in real estate-type interests, which threshold is subject to reduction to the extent that the entity invested more than 55% of its total assets in qualifying interests, and no more than 20% of such entity's total assets are invested in miscellaneous investments. Qualifying interests for this purpose include actual interests in real estate, certain mortgage loans and other assets as interpreted in a manner consistent with SEC staff guidance. We intend to treat the following as real estate-type interests: non-agency RMBS; CMBS, debt and equity securities of companies primarily engaged in real estate businesses; agency partial pool certificates and securities issued by pass-through entities of which substantially all of the assets consist of qualifying interests; and/or real estate-related assets. Due to the factual nature of this test, we, the Operating Partnership, or a subsidiary may be unable to sell assets we would otherwise want to sell or may need to sell assets we would otherwise wish to retain, if we deem it necessary to remain in compliance with the foregoing standards. In addition, we may have to forgo opportunities to acquire certain investments or interests in companies or entities that we would otherwise want to acquire, or acquire assets we might otherwise not select for purchase, if we deem it necessary to remain in compliance with the foregoing standards.

In addition, we, the Operating Partnership and/or our subsidiaries may rely upon other exclusions, including the exclusion provided by Section 3(c)(6) of the Investment Company Act (which excludes, among other things, parent entities whose primary business is conducted through majority-owned subsidiaries relying upon the exclusion provided by Section 3(c)(5)(C) (discussed above)), from the definition of an investment company and the registration requirements under the Investment Company Act.

There can be no assurance that the laws and regulations governing the Investment Company Act status of REITs (and/or their subsidiaries), including actions by the SEC or its staff providing more specific or different guidance regarding these exclusions, will not change in a manner that adversely affects our operations. For example, on August 31, 2011, the SEC issued a concept release requesting comments regarding a number of matters relating to the exclusion provided by Section 3(c)(5)(C) of the Investment Company Act, including the nature of assets that qualify for purposes of the exclusion and whether mortgage REITs should be regulated in a manner similar to investment companies. To the extent that the SEC or the SEC staff provides more specific guidance regarding any of the matters bearing upon the exclusions discussed above or other exclusions from the definition of an investment company under the Investment Company Act upon which we may rely, we may be required to change the way we conduct our business or adjust our strategy accordingly. Any additional guidance from the SEC staff could provide additional flexibility to us, or it could further inhibit our ability to pursue the strategies we have chosen. If we meet the definition of an investment company under the Investment Company Act and we fail to qualify for an exclusion therefrom, our ability to use leverage and other business strategies would be substantially reduced. Our business will be materially and adversely affected if we fail to qualify for an exclusion from regulation under the Investment Company Act.

If Hines Global II or the Operating Partnership is required to register as an investment company under the Investment Company Act, the additional expenses and operational limitations associated with such registration may reduce our stockholders investment return or impair our ability to conduct our business as planned.

If we become an investment company or are otherwise required to register as an investment company, we might be required to revise some of our current policies, or substantially restructure our business, to comply with the Investment Company Act. This would likely require us to incur the expense and delay of holding a stockholder meeting to vote on proposals for such changes. Further, if we were required to register as an investment company, but failed to do so, we would be prohibited from engaging in our business, criminal and civil actions could be brought against us, some of our contracts might be unenforceable, unless a court were to direct enforcement, and a court could appoint a receiver to take control of us and liquidate our business.

If we internalize our management functions, we could incur adverse effects on our business and financial condition, including significant costs associated with becoming and being self-managed and the percentage of our outstanding common stock owned by our stockholders could be reduced.

If we seek to list our shares on an exchange as a way of providing our stockholders with a liquidity event, we may consider internalizing the functions performed for us by our Advisor. An internalization could take many forms, for example, we may hire our own group of executives and other employees or we may acquire our Advisor or its respective assets including its existing workforce. Any internalization could result in significant payments, including in the form of our stock, to the owners of our Advisor as compensation, which could reduce the percentage ownership of our then existing stockholders and concentrate ownership in Hines. In addition, there is no assurance that internalizing our management functions will be beneficial to us and our stockholders. For example, we may not realize the perceived benefits because of: (i) the costs of being self-managed; (ii) our inability to effectively integrate a new staff of managers and employees; or (iii) our inability to properly replicate the services provided previously by our Advisor or its affiliates. Additionally, internalization transactions have also, in some cases, been the subject of litigation and even if these claims are without merit, we could be forced to spend significant amounts of money defending claims which would reduce the amount of funds available for us to invest in real estate investments or to pay distributions. In connection with any such internalization transaction, a special committee consisting of our independent directors will be appointed to evaluate the transaction and to determine whether a fairness opinion should be obtained.

Risks Related to Potential Conflicts of Interest

We compete with affiliates of Hines for real estate investment opportunities and some of these affiliates have preferential rights to accept or reject certain investment opportunities in advance of our right to accept or reject such opportunities.

Hines has existing real estate joint ventures, funds and programs, which we collectively refer to as real estate investment vehicles, with investment objectives and strategies similar to ours, including Hines REIT and Hines Global I. Because we

compete with these real estate investment vehicles for investment opportunities, Hines faces conflicts of interest in allocating investment opportunities between us and these other real estate investment vehicles. We have limited rights to specific investment opportunities located by Hines. Some of these entities have a priority right over other Hines real estate investment vehicles, including us, to accept investment opportunities that meet certain defined investment criteria. Because we and other Hines real estate investment vehicles rely on Hines to present us with investment opportunities, these rights will reduce our investment opportunities. Please see "Conflicts of Interest—Competitive Activities of Hines and its Affiliates" for a description of some of these entities and priority rights. We therefore may not be able to invest in, or we may only invest indirectly with or through another Hines affiliated real estate investment vehicle in, certain investments we otherwise would make directly. To the extent we invest in opportunities with another real estate investment vehicles affiliated with Hines, we may not have the control over such investment we would otherwise have if we owned all of or otherwise controlled such assets.

We do not have priority rights to specific investment opportunities located by Hines. Our right to participate in Hines' investment allocation process will terminate once we have fully invested the proceeds of our public offerings or if we are no longer advised by an affiliate of Hines. For investment opportunities not covered by this allocation procedure, Hines will decide in its discretion, subject to any priority rights it grants or has granted to other Hines-managed or otherwise affiliated real estate investment vehicles, how to allocate such opportunities among us, Hines and other real estate investment vehicles. Because we do not have a right to accept or reject any investment opportunities before Hines or one or more Hines real estate investment vehicles have the right to accept such opportunities, and are otherwise subject to Hines' discretion as to the investment opportunities we will receive, we may not be able to review and/or invest in opportunities which we would otherwise pursue if we were the only real estate investment vehicle sponsored by Hines or had a priority right in regard to such investments. We are subject to the risk that, as a result of the conflicts of interest between Hines, us and other real estate investment vehicles sponsored or managed by or affiliated with Hines, and the priority rights Hines has granted or may in the future grant to any such other real estate investment vehicles, we may not be offered favorable investment opportunities located by Hines when it would otherwise be in our best interest to accept such investment opportunities, and our business, results of operations, cash flows and financial condition and our ability to make distributions to our stockholders and the value of their investment may be adversely impacted thereby.

We may compete with other investment vehicles affiliated with Hines for tenants.

Hines and its affiliates are not prohibited from engaging, directly or indirectly, in any other business or from possessing interests in any other business venture or ventures, including businesses and ventures involved in the acquisition, development, ownership, management, leasing or sale of real estate projects. Hines or its affiliates own and/or manage properties in most, if not all, geographical areas in which we expect to acquire interests in real estate assets. Therefore, our properties may compete for tenants with other properties owned and/or managed by Hines and its affiliates, including Hines REIT and Hines Global I. Hines may face conflicts of interest when evaluating tenant opportunities for our properties and other properties owned and/or managed by Hines and its affiliates and these conflicts of interest may have a negative impact on our ability to attract and retain tenants.

Employees of our Advisor and Hines will face conflicts of interest relating to time management and allocation of resources and investment opportunities.

We do not have employees. Pursuant to a contract with Hines, we rely on employees of Hines and its affiliates to manage and operate our business and they are contractually bound to devote the time and attention reasonably necessary to conduct our business in an appropriate manner. Our officers and the officers and employees of our Advisor, Hines and its affiliates hold similar positions in numerous entities and they may from time to time allocate more of their time to service the needs of such entities than they allocate to servicing our needs. Hines is not restricted from acquiring, developing, operating, managing, leasing or selling real estate through entities other than us and Hines will continue to be actively involved in real estate operations and activities other than our operations and activities. Hines currently controls and/or operates other entities that own properties in many of the markets in which we will seek to invest. Hines spends a material amount of time managing these properties and other assets unrelated to our business. We lack the ability to manage it without the time and attention of Hines' employees.

Hines and its affiliates are general partners and sponsors of other real estate investment vehicles having investment objectives and legal and financial obligations similar to ours, including Hines REIT and Hines Global I. Because Hines and its affiliates have interests in other real estate investment vehicles and also engage in other business activities, they may have conflicts of interest in allocating their time and resources among our business and these other activities. Our officers and directors, as well as those of our Advisor, own equity interests in entities affiliated with Hines from which we may buy properties. These individuals may make substantial profits in connection with such transactions, which could result in conflicts of interest. Likewise, such individuals could make substantial profits as the result of investment opportunities allocated to

entities affiliated with Hines other than us. As a result of these interests, they could pursue transactions that may not be in our best interest.

Hines may face conflicts of interest if it sells properties it acquires or develops to us.

We may in the future acquire properties from Hines and affiliates of Hines. We may acquire properties Hines currently owns or hereafter acquires from third parties. Hines may also develop properties and then sell the completed properties to us. Similarly, we may provide development loans to Hines in connection with these developments. Hines, its affiliates and its employees (including our officers and directors) may make substantial profits in connection with such transactions. We must follow certain procedures when purchasing assets from Hines and its affiliates. Hines may owe fiduciary and/or other duties to the selling entity in these transactions and conflicts of interest between us and the selling entities could exist in such transactions. Because we are relying on Hines, these conflicts could result in transactions based on terms that are less favorable to us than we would receive from a third party.

Hines may face conflicts of interest when determining whether we should dispose of any property we own that is managed by Hines because Hines may lose fees associated with the management of the property.

We expect that Hines will manage many of the properties we acquire directly as well as many of the properties in which we acquire an indirect interest, should we invest in other Hines affiliated entities. Because Hines receives significant fees for managing these properties, it may face a conflict of interest when determining whether we should sell properties under circumstances where Hines would no longer manage the property after the transaction. As a result of this conflict of interest, we may not dispose of properties when it would be in our best interests to do so.

Hines may face conflicts of interest in connection with the management of our day-to-day operations and in the enforcement of agreements between Hines and its affiliates.

Hines and our Advisor manage our day-to-day operations and properties pursuant to an advisory agreement. This agreement was not negotiated at arm's length and certain fees payable by us under such agreement are paid regardless of our performance. Hines and its affiliates may encounter conflicts of interest with respect to position as to matters relating to this agreement. Areas of potential conflict include the computation of fees and reimbursements under such agreements, the enforcement, renewal and/or termination of the agreements and the priority of payments to third parties as opposed to amounts paid to affiliates of Hines. These fees may be higher than fees charged by third parties in an arm's-length transaction as a result of these conflicts.

Certain of our officers and directors face conflicts of interest relating to the positions they hold with other entities.

All of our officers and non-independent directors are also officers and directors of our Advisor and/or other entities controlled by Hines. Some of these entities, including Hines REIT and Hines Global I, may compete with us for investment and leasing opportunities. These personnel owe fiduciary duties to these other entities and their security holders and these duties may from time to time conflict with the fiduciary duties such individuals owe to us and our stockholders. For example, conflicts of interest adversely affecting our investment decisions could arise in decisions or activities related to:

- the allocation of new investments among us and other entities operated by Hines;
- the allocation of time and resources among us and other entities operated by Hines;
- the timing and terms of the investment in or sale of an asset;
- investments with Hines and affiliates of Hines;
- the compensation paid to our Advisor; and
- our relationship with Hines in the management of our properties.

These conflicts of interest may also be impacted by the fact that such individuals may have compensation structures tied to the performance of such other entities controlled by Hines and these compensation structures may potentially provide for greater remuneration in the event an investment opportunity is presented to a Hines affiliate rather than us.

Our officers and directors have limited liability.

Generally, we are obligated under our charter to indemnify our officers and directors against certain liabilities incurred in connection with their services. We enter into indemnification agreements with each of our officers and directors pursuant to which we generally agree to indemnify our officers and directors for any such liabilities that they incur. These indemnification agreements, as well as the indemnification provisions in our charter, could limit our ability and the ability of our stockholders to effectively take action against our officers and directors arising from their service to us. In addition, there could be a potential reduction in distributions resulting from our payment of premiums associated with insurance or payments of a defense, settlement or claim.

Our UPREIT structure may result in potential conflicts of interest.

Persons holding OP Units have the right to vote on certain amendments to the Agreement of Limited Partnership of the Operating Partnership, as well as on certain other matters. Persons holding such voting rights may exercise them in a manner that conflicts with the interests of our stockholders. As general partner of the Operating Partnership, we will be obligated to act in a manner that is in the best interest of all partners of the Operating Partnership. Circumstances may arise in the future when the interests of limited partners in the Operating Partnership may conflict with the interests of our stockholders.

Risks Related to Taxes

If we fail to qualify as a REIT, our operations and our ability to pay distributions to our stockholders would be adversely impacted.

We intend to qualify as a REIT for U.S. federal income tax purposes commencing with the taxable year ending on December 31, 2015. We have received the opinion of our U.S. federal income tax counsel, Greenberg Traurig, LLP, in connection with the Offering and with respect to our qualification as a REIT. The opinion of Greenberg Traurig, LLP represents only the view of our counsel based on our counsel's review and analysis of existing law and on certain representations as to factual matters and covenants made by us, including representations relating to the values of our assets and the sources of our income and is not binding on the Internal Revenue Service or any court. Greenberg Traurig, LLP has no obligation to advise us or the holders of our common stock of any subsequent change in the matters stated, represented or assumed in its opinion or of any subsequent change in applicable law. Furthermore, both the validity of the opinion of Greenberg Traurig LLP and our qualification as a REIT will depend on our satisfaction of numerous requirements (some on an annual and quarterly basis) established under highly technical and complex provisions of the Code, for which there are only limited judicial or administrative interpretations, and involves the determination of various factual matters and circumstances not entirely within our control. The complexity of these provisions and of the applicable income tax regulations that have been promulgated under the Code is greater in the case of a REIT that holds its assets through a partnership, as we do. Moreover, no assurance can be given that legislation, new regulations, administrative interpretations or court decisions will not change the tax laws with respect to qualification as a REIT or the U.S. federal income tax consequences of that qualification.

If we were to fail to qualify as a REIT in any taxable year:

- we would not be allowed to deduct our distributions to our stockholders when computing our taxable income;
- we would be subject to federal income tax (including any applicable alternative minimum tax) on our taxable income at regular corporate rates;
- we would be disqualified from being taxed as a REIT for the four taxable years following the year during which qualification was lost, unless entitled to relief under certain statutory provisions;
- our cash available for distribution would be reduced and we would have less cash to distribute to our stockholders; and
- we might be required to borrow additional funds or sell some of our assets in order to pay corporate tax obligations we may incur as a result of our disqualification.

We may be required to defer repatriation of cash from foreign jurisdictions in order to qualify as a REIT.

Investments in foreign real property may be subject to foreign currency gains and losses. Certain foreign currency gains will generally be excluded from income for purposes of determining our satisfaction of one or both of the REIT gross revenue tests; however, under certain circumstances (for example, if we regularly trade in foreign securities) such gains will be treated as non-qualifying income. To reduce the risk of foreign currency gains adversely affecting our REIT qualification, we may be required to defer the repatriation of cash from foreign jurisdictions or to employ other structures that could affect the timing, character or amount of income we receive from our foreign investments. No assurance can be given that we will be able to manage our foreign currency gains in a manner that enables us to qualify as a REIT or to avoid U.S. federal and other taxes on our income as a result of foreign currency gains.

If we were considered to actually or constructively pay a "preferential dividend' to certain of our stockholders, our status as a REIT could be adversely affected.

In order to qualify as a REIT, we must generally distribute to our stockholders at least 90% of our annual REIT taxable income, determined without regard to the deduction for dividends paid. In order for distributions to be counted as satisfying the annual distribution requirements for REITs, and to provide us with a REIT-level tax deduction, the distributions must not be "preferential dividends." A dividend is not a preferential dividend if the distribution is pro rata among all outstanding shares of stock within a particular class, and in accordance with the preferences among different classes of stock as set forth in our organizational documents. If the IRS were to take the position that we paid a preferential dividend, and if a court were to sustain such a position, we might be deemed to fail the 90% distribution test, and our status as a REIT could be terminated.

If the Operating Partnership is classified as a "publicly traded partnership" under the Code, our operations and our ability to pay distributions to our stockholders could be adversely affected.

We believe that the Operating Partnership will be treated as a partnership, and not as an association or a publicly traded partnership for federal income tax purposes. In this regard, the Code generally classifies "publicly traded partnerships" (as defined in Section 7704 of the Code) as associations taxable as corporations (rather than as partnerships), unless substantially all of their taxable income consists of specified types of passive income. In order to minimize the risk that the Code would classify the Operating Partnership as a "publicly traded partnership" for tax purposes, we placed certain restrictions on the transfer and/or repurchase of partnership units in the Operating Partnership. However, if the Internal Revenue Service, or IRS, successfully determines that the Operating Partnership should be taxed as a corporation, the Operating Partnership would be required to pay U.S. federal income tax at corporate rates on its net income, its partners would be treated as stockholders of the Operating Partnership and distributions to partners would constitute non-deductable distributions in computing the Operating Partnership's taxable income. In addition, we could fail to qualify as a REIT and the imposition of a corporate tax on the Operating Partnership would reduce our amount of cash available for distribution to our stockholders.

Distributions to tax-exempt investors may be classified as unrelated business taxable income.

Neither ordinary nor capital gain distributions with respect to our common shares nor gain from the sale of common shares should generally constitute unrelated business taxable income to a tax-exempt investor. However, there are certain exceptions to this rule. In particular:

- part of the income and gain recognized by certain qualified employee pension trusts with respect to our common shares may be treated as unrelated business taxable income if our stock is predominately held by qualified employee pension trusts, we are required to rely on a special look through rule for purposes of meeting one of the REIT stock ownership tests, and we are not operated in such a manner as to otherwise avoid treatment of such income or gain as unrelated business taxable income;
- part of the income and gain recognized by a tax exempt investor with respect to our common shares would constitute unrelated business taxable income if such investor incurs debt in order to acquire the common shares; and
- part or all of the income or gain recognized with respect to our common shares by social clubs, voluntary employee benefit associations, supplemental unemployment benefit trusts and qualified group legal services plans which are exempt from federal income taxation under Sections 501(c)(7), (9), (17), or (20) of the Code may be treated as unrelated business taxable income.

Stockholders who participate in our distribution reinvestment plan may realize taxable income without receiving cash distributions.

If our stockholders participate in the distribution reinvestment plan, they generally will be required to take into account, in computing their taxable income, ordinary and capital gain distributions allocable to shares that they own, even though they receive no cash because such distributions are reinvested. In addition, the difference between the public offering price of our shares and the amount paid for shares purchased pursuant to our distribution reinvestment plan may be deemed to be taxable as income to participants in the plan.

Foreign investors may be subject to FIRPTA tax on sale of common shares if we are unable to qualify as a "domestically controlled" REIT.

A foreign person disposing of a U.S. real property interest, including shares of a U.S. corporation whose assets consist principally of U.S. real property interests, is generally subject to a tax, known as FIRPTA tax, on the gain recognized on the disposition. Such FIRPTA tax does not apply, however, to the disposition of stock in a REIT if the REIT is "domestically controlled." A REIT is "domestically controlled" if less than 50% of the REIT's capital stock, by value, has been owned directly or indirectly by persons who are not qualifying U.S. persons during a continuous five-year period ending on the date of disposition or, if shorter, during the entire period of the REIT's existence.

We cannot assure our stockholders that we will qualify as a "domestically controlled" REIT. If we were to fail to so qualify, gain realized by foreign investors on a sale of our common shares would be subject to FIRPTA tax (unless our common shares were traded on an established securities market and the foreign investor did not at any time during a specified testing period directly or indirectly own more than 5% of the value of our outstanding common shares). Our common shares are not currently traded on an established securities market.

In certain circumstances, we may be subject to federal and state income taxes as a REIT or other state or local income taxes, which would reduce our cash available to pay distributions to our stockholders.

Even if we qualify and maintain our status as a REIT, we may be subject to federal income taxes or state taxes. For example, if we have net income from a "prohibited transaction," such income will be subject to a 100% tax. We may not be able to make sufficient distributions to avoid paying federal income tax and/or the 4% excise tax that generally applies to income retained by a REIT. We may also decide to retain income we earn from the sale or other disposition of our property and pay income tax directly on such income. In that event, our stockholders would be treated as if they earned that income and paid the tax on it directly. However, stockholders that are tax-exempt, such as charities or qualified pension plans, would have no benefit from their deemed payment of such tax liability. We may also be subject to state and local taxes on our income or property, either directly or at the level of the Operating Partnership or at the level of the other companies through which we indirectly own our assets.

We may enter into certain hedging transactions which may have a potential impact on our REIT status.

We may enter into certain hedging transactions with respect to one or more of our assets or liabilities. Our hedging activities may include entering into interest rate and/or foreign currency swaps, caps and floors, options to purchase these items, and futures and forward contracts. The gross income tests generally exclude any income or gain from a hedging or similar transaction entered into by the REIT primarily to manage the risk of interest rate, price changes or currency fluctuations with respect to borrowings made or to be made to acquire or carry real estate assets or to manage the risk of currency fluctuations with respect to an item of income or gain that would be qualifying income under the 75% or 95% gross income test (or any property which generates such income or gain), provided we properly identify such hedges and other transactions in the manner required by the Code. To the extent that we do not properly identify such transactions as hedges or we hedge with other types of financial instruments, or hedge other types of indebtedness, the income from those transactions is likely to be treated as non-qualifying income for purposes of the gross income tests and may affect our ability to qualify as a REIT.

Entities through which we hold foreign real estate investments are, in most cases, subject to foreign taxes, notwithstanding our status as a REIT.

Even if we maintain our status as a REIT, entities through which we hold investments in assets located outside the United States will, in most cases, be subject to income taxation by jurisdictions in which such assets are located. Our cash available for distribution to our stockholders will be reduced by any such foreign income taxes.

Dividends payable by REITs do not qualify for the reduced tax rates available for some dividends.

The maximum tax rate applicable to income from "qualified dividends" payable to U.S. stockholders that are individuals, trusts or estates is currently 20%. Dividends payable by REITs, however, generally are not eligible for the reduced rates. The more favorable rates applicable to regular corporate qualified dividends could cause investors who are individuals, trusts or estates to perceive investments in our common shares to be relatively less attractive than investments in the stocks of non-REIT corporations that pay dividends, which could adversely affect the value of our common shares.

Recharacterization of sale-leaseback transactions may cause us to lose our REIT status.

We may purchase real properties and lease them back to the sellers of such properties. We will use commercially reasonable efforts to structure any such sale-leaseback transaction such that the lease will be characterized as a "true lease," thereby allowing us to be treated as the owner of the property for federal income tax purposes, but cannot assure our stockholders that the IRS will not challenge such characterization. In the event that any such sale-leaseback transaction is challenged and recharacterized as a financing transaction or loan for U.S. federal income tax purposes, deductions for depreciation and cost recovery relating to such property would be disallowed. We might fail to satisfy the REIT qualification "asset tests" or the "income tests" and, consequently, lose our REIT status effective with the year of recharacterization if a sale-leaseback transaction were so recharacterized. Alternatively, the amount of our REIT taxable income could be recalculated which might also cause us to fail to meet the distribution requirement for a taxable year.

Investments in other REITs and real estate partnerships could subject us to the tax risks associated with the tax status of such entities.

We may invest in the securities of other REITs and real estate partnerships. Such investments are subject to the risk that any such REIT or partnership may fail to satisfy the requirements to qualify as a REIT or a partnership, as the case may be, in any given taxable year. In the case of a REIT, such failure would subject such entity to taxation as a corporation. Failure to qualify as a REIT may require such REIT to incur indebtedness to pay its tax liabilities, may reduce its ability to make distributions to us, and may render it ineligible to elect REIT status prior to the fifth taxable year following the year in which it fails to so qualify. In the case of a partnership, such failure could subject such partnership to an entity level tax and reduce the entity's ability to make distributions to us. In addition, such failures could, depending on the circumstances, jeopardize our ability to qualify as a REIT.

Complying with the REIT requirements may cause us to forego otherwise attractive opportunities.

To qualify as a REIT for U.S. federal income tax purposes, we must continually satisfy tests concerning, among other things, the sources of our income, the nature and diversification of our assets, the amounts we distribute to our stockholders and the ownership of shares of our common stock. We may be required to forego otherwise attractive investments or make distributions to stockholders at disadvantageous times or when we do not have funds readily available for distribution. Thus, compliance with the REIT requirements may hinder our ability to operate solely on the basis of maximizing profits.

Complying with the REIT requirements may force us to liquidate otherwise attractive investments.

We must ensure that at the end of each calendar quarter, at least 75% of the value of our assets consists of cash, cash items, government securities and qualified REIT real estate assets in order to ensure our qualification as a REIT. The remainder of our investments (other than governmental securities and qualified real estate assets) generally cannot include more than 10% of the outstanding voting securities of any one issuer or more than 10% of the total value of the outstanding securities and qualified real estate assets) concerning securities and qualified real estate assets (other than government securities and qualified real estate assets) concerning securities and qualified real estate assets) can consist of the securities of any one issuer, and no more than 25% of the value of our total securities can be represented by securities of one or more taxable REIT subsidiaries. If we fail to comply with these requirements at the end of any calendar quarter, we must correct such failure within 30 days after the end of the calendar quarter to avoid losing our REIT status and suffering adverse tax consequences. As a result, we may be required to liquidate otherwise attractive investments.

The failure of a mezzanine loan or any other loan which is not secured by a mortgage on real property to qualify as a real estate asset could adversely affect our ability to qualify as a REIT.

The IRS has issued Revenue Procedure 2003-65, which provides a safe harbor pursuant to which a mezzanine loan that is secured by interests in a pass-through entity will be treated by the IRS as a real estate asset for purposes of the REIT tests, and interest derived from such loan will be treated as qualifying mortgage interest for purposes of the REIT 75% income test. Although the Revenue Procedure provides a safe harbor on which taxpayers may rely, it does not prescribe rules of substantive

tax law. We may make investments in loans secured by interests in pass-through entities in a manner that complies with the various requirements applicable to our qualification as a REIT. To the extent, however, that any such loans do not satisfy all of the requirements for reliance on the safe harbor set forth in the Revenue Procedure, there can be no assurance that the IRS will not challenge the tax treatment of such loans, which could jeopardize our ability to qualify as a REIT. Similarly any other loan which we make which is not secured by a mortgage on real property may fail to qualify as a real estate asset for purposes of the Federal Income tax REIT qualification tests and therefore could adversely affect our ability to qualify as a REIT.

Legislative or regulatory action could adversely affect us and/or our investors.

In recent years, numerous legislative, judicial and administrative changes have been made to the U.S. federal income tax laws applicable to the qualification and taxation of REITs and to investments in REITs and similar entities. Additional changes to tax laws are likely to continue to occur in the future and may be given retroactive or prospective effect, and we cannot assure our stockholders that any such changes will not adversely affect how we are taxed or the taxation of a stockholder. Any such changes could have an adverse effect on us and on an investment in shares of our common stock. We urge our stockholders to consult with their own tax advisors with respect to the status of legislative, regulatory or administrative developments and proposals and their potential effect on an investment in shares of our common stock.

Risks Related to ERISA

If our assets are deemed to be plan assets under the Employee Retirement Income Security Act of 1974, as amended, or ERISA, we, our Advisor and the fiduciaries of investing ERISA plans may be exposed to liabilities under Title I of ERISA and the Internal Revenue Code.

In some circumstances where an ERISA plan holds an interest in an entity, an undivided interest in the assets of the entity attributable to that interest are deemed to be ERISA plan assets unless an exception applies. This is known as the "look-through rule." Under those circumstances, the obligations and other responsibilities of plan sponsors, plan fiduciaries and plan administrators, and of parties in interest and disqualified persons, under Title I of ERISA and Section 4975 of the Code, as applicable, may be applicable, and there may be liability under these and other provisions of ERISA and the Code. We believe that our assets should not be treated as plan assets because the shares should qualify as "publicly-offered securities" that are exempt from the look-through rules under applicable regulations of the U.S. Department of the Treasury, because we have 100 stockholders that are independent of us and one another. We note, however, that because certain limitations are imposed upon the transferability of shares so that we may qualify as a REIT, and perhaps for other reasons, it is possible that this exemption may not apply. If that is the case, and if our Advisor or we are exposed to liability under ERISA or the Code, our performance and results of operations could be adversely affected. In addition, if that were the case, an investment in our common shares might constitute an ineffective delegation of fiduciary responsibility to our Advisor, and expose the fiduciary of the benefit plan to co-fiduciary liability under ERISA for any breach by our Advisor of the fiduciary duties mandated under ERISA. If our Advisor or we are exposed to liability under ERISA or the Code, our performance and results of operations could be adversely affected. Prior to making an investment in us, potential investors should consult with their legal and other advisors concerning the impact of ERISA and the Code on such investors' investment and our performance.

There are special considerations that apply to pension or profit sharing trusts or individual retirement accounts, or IRAs, investing in our common stock.

If a stockholder is investing the assets of an IRA, pension, profit sharing, 401(k), Keogh or other qualified retirement plan, they should satisfy themselves that:

- their investment is consistent with fiduciary obligations under ERISA and the Code;
- their investment is made in accordance with the documents and instruments governing their plan or IRA, including their plan's investment policy;
- their investment satisfies the prudence and diversification requirements of Sections 404(a)(1)(B) and 404(a)(1)(C) of ERISA;
- their investment will not impair the liquidity of the plan or IRA;
- their investment will not produce "unrelated business taxable income" for the plan or IRA;
- they will be able to value the assets of the plan annually in accordance with ERISA requirements; and
- their investment will not constitute a prohibited transaction under Section 406 of ERISA or Section 4975 of the Code.

Item 1B. Unresolved Staff Comments

Not applicable.

Item 2. Properties

As of December 31, 2014, we owned one real estate investment, 2819 Loker Avenue East. Additionally, we acquired our second real estate investment, Bishop's Square, in March 2015. The following table provides additional information regarding each of these properties:

Property ⁽¹⁾	Location	Investment Type	Date Acquired/ Net Purchase Price (in millions)	Estimated Going-in Capitalization Rate ⁽³⁾	Leasable Square Feet	Percent Leased
2819 Loker Avenue East	Carlsbad, California	Industrial	12/2014; \$25.4	6.5%	161,310	100%
Bishop's Square	Dublin, Ireland	Office	3/2015; \$103.5 (2)	6.1%	153,529	100%
Total for All Investments					314,839	100%

- (1) On December 31, 2014, the Company effectively owned a 95.2% interest in these properties through the ownership interest in the Operating Partnership as its sole general partner. Hines Global REIT II Associates Limited Partnership ("HALP II"), an affiliate of Hines, owned the remaining 4.8% interest in the Operating Partnership.
- (2) The net purchase price for Bishop's Square was denominated in Euros and has been translated at an exchange rate based on the rate in effect on the acquisition date.
- (3) The estimated going-in capitalization rate is determined as of the date of acquisition by dividing the projected property revenues in excess of expenses for the first fiscal year by the net purchase price (excluding closing costs and taxes). Property revenues in excess of expenses includes all projected operating revenues (rental income, tenant reimbursements, parking and any other property-related income) less all projected operating expenses (property operating and maintenance expenses, property taxes, insurance and property management fees). The projected property revenues in excess of expenses of expenses includes assumptions which may not be indicative of the actual future performance of the property, including the assumption that the tenants will perform under their lease agreements during the 12 months following our acquisition of the properties.

2819 Loker Avenue East

On December 17, 2014, we acquired 2819 Loker Avenue East, a Class A industrial building located in Carlsbad, California. 2819 Loker Avenue East was constructed in 1998 and consists of 161,310 square feet of net rentable area that is 100% leased to the Acushnet Company, the parent company of Titleist, FootJoy, BV-Vokey Design, Scotty Cameron, Pinnacle, and the Titleist Performance Institute, under a lease that expires in July 2019.

Bishop's Square

On March 3, 2015, we acquired Bishop's Square, a Class A office building located in Dublin, Ireland. Bishop's Square was constructed in 2001 and consists of 153,529 square feet of net rentable area that is 100% leased to five tenants. The Commissioners of Public Works in Ireland, a state agency of Ireland, leases 90,017 square feet or approximately 59%, of the net rentable area of Bishop's Square, under a lease that expires in January 2028. Further, International Financial Data Services, an investor record-keeping and transfer agency provider, leases 32,916 square feet or approximately 21%, of the net rentable area of Bishop's Square, under a lease that expires in June 2024. Additionally, News Corp UK & Ireland Limited, a diversified media and information services company, leases 16,153 square feet, or approximately 11%, of the net rentable area of Bishop's Square, under a lease that expires in August 2027. The remaining space is leased to two tenants, neither of which leases more than 10% of the net rentable area of Bishop's Square.

Lease Expirations

The following table lists the scheduled lease expirations and related expiring base rents of our properties for each of the years ending December 31, 2015 through December 31, 2024. The table also shows the approximate leasable square feet represented by the applicable lease expirations:

		Leasabl	e Area					
Number of Year Leases		Approximate Percent of Total Square Feet Leasable Area			Annual Base Ital Income of piring Leases	Percent of Total Annual Base Rental Income		
2015	1	2,048	0.7%	\$	11,673	0.2%		
2016			%	\$	—	%		
2017	—	—	%	\$	—	%		
2018	—		%	\$		%		
2019	1	161,310	51.2%	\$	1,845,225	23.0%		
2020	—		%	\$		%		
2021	—	_	%	\$		%		
2022	1	12,395	3.9%	\$	663,128	8.3%		
2023	—		<u> %</u>	\$		%		
2024	1	32,916	10.5%	\$	1,117,704	13.9%		
Thereafter	2	106,170	33.7%	\$	4,383,586	54.6%		

Industry Concentration

The following table provides a summary of the industry concentration of the tenants of our properties based on their leased square footage:

Industry	Industry Concentration
Manufacturing	51%
Government	33%
Finance and Insurance	10%
Information	5%
Administrative and Support Services	1%

Item 3. Legal Proceedings

From time to time in the ordinary course of business, the Company or its subsidiaries may become subject to legal proceedings, claims or disputes. As of March 25, 2015, neither the Company nor any of its subsidiaries was a party to any material pending legal proceedings.

Item 4. Mine Safety Disclosures

Not applicable.

PART II

Item 5. Market for Registrant's Common Equity, Related Stockholder Matters and Issuer Purchases of Equity Securities

Market Information

As of December 31, 2014, we had 420,541 common shares outstanding, held by a total of 41 stockholders. The number of stockholders is based on the records of our registrar and transfer agent. There is no established public trading market for our common stock. Therefore, there is a risk that a stockholder may not be able to sell our stock at a time or price acceptable to the stockholder. Pursuant to the terms of the Offering, we are selling shares of our common stock to the public at a price of \$10.00 per share and at a price of \$9.50 per share pursuant to our distribution reinvestment plan.

To assist the Financial Industry Regulatory Authority ("FINRA") members and their associated persons that participate in our public offerings in their effort to comply with National Association of Securities Dealers ("NASD") Rule 2340, we disclose in each annual report distributed to investors a per share estimated value of the shares, the method by which it was developed and the date of the data used to develop the estimated value. In addition, we plan to prepare annual statements of estimated share values to assist fiduciaries of retirement plans subject to the annual reporting requirements of ERISA in the preparation of their reports relating to an investment in our shares and such statements should not be used for any other purpose. For these purposes, the estimated value of a share of our common stock is estimated to be \$10.00 per share as of the date of filing of this Annual Report on Form 10-K, as that is the price at which we are currently offering and selling shares of our common stock in our Offering. Although this estimated value is the price currently paid to acquire a share in our Offering, this estimated value is not likely to reflect the proceeds our stockholders would receive upon our liquidation or upon the sale of our stockholders' shares because (i) this estimate is not designed to reflect the price at which properties and other assets can be sold; (ii) our proceeds invested in real estate are reduced by fees such as the commissions, dealer manager fee and issuer costs associated with the Offering; (iii) no public market for our shares exists or is likely to develop; and (iv) the per share valuation method is not designed to arrive at a valuation that is related to any individual or aggregated value estimates or appraisals of the value of our assets. We are not required to obtain and did not obtain appraisals for our assets or third-party valuations or opinions for the specific purpose of determining this estimated value as of December 31, 2014.

The SEC has approved an amendment to NASD Rule 2340, which sets forth the obligations of FINRA members to provide per share values in customer account statements. We will disclose an estimated per share NAV no later than 150 days following the second anniversary of the date on which we break escrow in the offering, which was September 26, 2014, although we may determine to provide an estimated per share NAV earlier. Thereafter, in accordance with the amended rule, we intend to undertake a valuation process to provide an estimated per share value annually. In connection with the disclosure of a new estimated per share value of our common stock, our board of directors may determine to modify the offering price for each class of our shares, if we are engaged in an offering at that time and the purchase price stockholders pay for shares of our common stock may be higher than such estimated per share value. Further, the amendment to NASD Rule 2340 takes effect on April 11, 2016, prior to the conclusion of the Offering, and if we have not yet disclosed an estimated per share value of our common stock by the time the amended rule takes effect, our stockholders' customer account statements will include a value per share that is less than the offering price for such class of shares of our common stock in the offering, because the amendment requires the "value" on the customer account statement to be equal to the offering price less up-front underwriting compensation and organization and offering expenses. Once the amended rule takes effect, we expect to use this "value" from customer account statements as the estimated per share value for purposes of reports to fiduciaries.

Distributions

In order to meet the requirements for being treated as a REIT under the Internal Revenue Code, we must pay distributions to our stockholders each taxable year in an amount equal to at least 90% of our net ordinary taxable income (capital gains are not required to be distributed). With the authorization of our board of directors, we declared distributions for the period from October 1, 2014 through March 31, 2015. These distributions were calculated based on stockholders of record for each day in an amount equal to \$0.001575342 per share, per day. All distributions were or will be paid in cash or reinvested in shares of our common stock for those participating in our distribution reinvestment plan and will be paid or issued, respectively, on the first business day following the completion of the month to which they relate.

In addition to the distributions described above, our board of directors authorized special stock dividends as of daily record dates for the period from October 1, 2014 through March 31, 2015. Stock dividends for our common shares were calculated based on stockholders of record each day in an amount equal to 0.0000273973 of a share of common stock per share, per day. Shares issued related to our stock dividends were or will be issued on the first business day of the month following the quarter to which they relate.

In our initial quarters of operations, and from time to time thereafter, we may continue to be unable to generate sufficient cash flow from operations to fully fund distributions paid. Therefore, particularly in the earlier part of the Offering, some or all of our distributions may continue to be paid from other sources, such as proceeds from our debt financings, proceeds from the Offering, cash advances by our Advisor, cash resulting from a waiver or deferral of fees and/or proceeds from the sale of assets. For example, for the year ended December 31, 2014, 100% of our distributions were funded with cash flows from financing activities, which includes offering proceeds and our Advisor waived \$16,258 of asset management fees otherwise payable to it, as described below. We have not placed a cap on the amount of our distributions that may be paid from any of these sources. See "Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations — Financial Condition, Liquidity and Capital Resources" for additional information regarding our distributions.

In September 2014, the Advisor agreed to waive the asset management fees otherwise payable to it for the quarter ended December 31, 2014, to the extent that our modified funds from operations for the quarter ended December 31, 2014, as disclosed in this Annual Report on Form 10-K, amounts to less than 100% of the aggregate distributions declared to our stockholders for the quarter ended December 31, 2014. As a result of this waiver, the Advisor waived \$16,258 of asset management fees payable to it for the year ended ended December 31, 2014.

Furthermore, the Advisor also agreed to waive the asset management fees otherwise payable to it for the quarters ended March 31, 2015 and June 30, 2015, to the extent that our MFFO for such quarters, as disclosed in our Quarterly Report on Form 10-Q, amounts to less than 100% of the aggregate distributions declared to our stockholders for each respective quarter. As a result of these fee waivers, cash flows from operations that would have been paid to the Advisor for asset management fees may be available to pay distributions to stockholders. These fee waivers are not deferrals and accordingly, any fees that are waived will not be paid to the Advisor in cash at any time in the future.

The following table outlines our total cash distributions declared to stockholders and noncontrolling interests (HALP II) for the year ended December 31, 2014, including the breakout between the distributions declared in cash and those reinvested pursuant to our distribution reinvestment plan.

	Stockholders						ntrolling erests	
Distributions for the Year Ended		Cash ributions		ibutions vested		Total eclared	Total	Declared
December 31, 2014	\$	40,377	\$	1,841	\$	42,218	\$	3,060

Stock Dividends

The following table outlines our total stock dividends declared for the year ended December 31, 2014.

Dividends for the Year Ended	Issuance Date	Shares	Amo	ount ⁽¹⁾		
December 31, 2014	1/1/2015	737	\$	7,367		

(1) Amount based on \$10.00 per share offering price.

Distributions to stockholders are characterized for federal income tax purposes as ordinary income, capital gains, nontaxable return of capital or a combination of the three. Distributions that exceed our current and accumulated earnings and profits (calculated for tax purposes) constitute a return of capital for tax purposes rather than a distribution and reduce the stockholders' basis in our common shares. To the extent that a distribution exceeds both current and accumulated earnings and profits and the stockholders' basis in the common shares, it will generally be treated as a capital gain. We annually notify stockholders of the taxability of distributions paid during the preceding year.

For the year ended December 31, 2014, 100% of the distributions paid were treated as a return of capital for federal income tax purposes. The amount of distributions paid and taxable portion are not indicative or predictive of amounts anticipated in future periods.

Recent Sales of Unregistered Securities

We issued 1,111.111 common shares to Hines Global REIT II Investor Limited Partnership, in exchange for an investment of \$10,000 in connection with our formation on September 6, 2013 in an offering exempt from registration under Section 4(2) of the Securities Act of 1933, as amended. There have been no other sales of unregistered securities within the past three years.

Use of Proceeds from Registered Securities

On August 20, 2014, the Registration Statement on Form S-11 (File No. 333-191106) for the Offering, was declared effective under the Securities Act of 1933. The Offering commenced on August 20, 2014 and is currently expected to terminate on or before August 20, 2016, unless extended by our board of directors.

From August 20, 2014 through December 31, 2014, we received gross proceeds of approximately \$3.9 million through the sale of 419,430 shares to the public in connection with the Offering. Since August 20, 2014, we have used proceeds from the Offering to pay \$77,885 of selling commissions and dealer manager fees and \$17,250 of issuer costs related to the Offering. The selling commissions and dealer manager fees were not paid with respect to the shares sold through our distribution reinvestment plan. The selling commissions and dealer manager fees were paid to our dealer manager, which is an affiliate of Hines and is wholly-owned, indirectly, by, or for the benefit of, our Chairman, Jeffrey C. Hines and his father, Gerald D. Hines.

Net proceeds available for investment after the payment of the costs described above were approximately \$3.7 million. Approximately \$1.0 million was used in the acquisition of 2819 Loker Avenue East. We also used \$153,166 in offering proceeds to fund general and administrative expenses and to provide working capital for our real estate investments. We had approximately \$2.5 million in uninvested offering proceeds as of December 31, 2014.

Additionally, we may not generate sufficient cash flow from operations to fully fund distributions paid. As of December 31, 2014, as discussed in "Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations — Financial Condition, Liquidity and Capital Resources — Distributions" all of our distributions have been funded with cash flows from financing activities which includes offering proceeds.

Share Redemption Program

We offer a share redemption program that may allow stockholders who have purchased shares from us or received their shares through a non-cash transaction, not in the secondary market, and who have held their shares for a minimum of one year, to have their shares redeemed subject to certain limitations and restrictions. We initially intend to allow redemptions of our shares on a monthly basis. No fees will be paid to Hines in connection with any redemption. Our board of directors may terminate, suspend or amend the share redemption program upon 30 days' written notice without stockholder approval, which notice shall take the form of a report on Form 8-K filed at least 30 days prior to the effective date of such termination, suspension or amendment.

Subject to the limitations of and restrictions on the program, and subject to funds being available as described below, the number of shares repurchased during any consecutive 12-month period will be limited to no more than 5% of the number of outstanding shares of common stock at the beginning of that 12-month period. Unless our board of directors determines otherwise, the funds available for redemptions in each month will be limited to the funds received from the distribution reinvestment plan in the prior month. Our board of directors has complete discretion to determine whether all of such funds from the prior month's distribution reinvestment plan will be applied to redemptions in the following month, whether such funds are needed for other purposes or whether additional funds from other sources may be used for redemptions.

Shares that are redeemed will be redeemed at the following prices, unless such shares are redeemed in connection with the death or disability of a stockholder, as described below: (i) the lower of 92.5% of the then-current offering price or 92.5% of the price paid to acquire the shares from us for stockholders who have held their shares continuously for at least one year; (ii) the lower of 95.0% of the then-current offering price or 95.0% of the price paid to acquire the shares from us for stockholders who have held their shares continuously for at least two years; (iii) the lower of 97.5% of the then-current offering price or 97.5% of the price paid to acquire the shares from us for stockholders who have held their shares continuously for at least two years; (iii) the lower of 97.5% of the then-current offering price or 97.5% of the price paid to acquire the shares from us for stockholders who have held their shares continuously for at least three years; and (iv) the lower of 100% of the then-current offering price or 100% of the price paid to acquire the shares from us for stockholders, splits, recapitalizations or similar actions with respect to our common stock. However, our board of directors, in its sole discretion, may determine at any time to amend the share redemption program to redeem shares at a price that is higher or lower than the price paid for the shares by the redeeming stockholder. Any such price modification may be arbitrarily determined, or may be determined on a different basis, as determined by our board of

directors. For example, in the event that we are not engaged in a public offering of our primary shares, our board of directors may determine to amend the share redemption program to redeem shares at a price that is at or below the then-current net asset value per share.

In the event a stockholder is having all his shares redeemed, we may waive the one-year holding requirement for shares purchased under our distribution reinvestment plan. In addition, we may waive the one-year holding requirement in the event of a stockholder's bankruptcy. If we determine to waive the one-year holding requirement in these circumstances, then, for purposes of determining the applicable redemption price, the stockholder will be deemed to have held his shares continuously for one year. In the event of the death or disability of the holder, upon request, we will waive the one-year holding requirement. Except as noted below, shares that are redeemed in connection with the death or disability of a stockholder will be redeemed at a purchase price equal to the price paid to acquire such shares from us; provided, that, the redemption price cannot exceed the then-current offering price and, as described below, the redemption price will be reduced as necessary to equal the then-current offering price.

During the period of any public offering, the redemption price will be equal to or less than the price of the shares offered in the relevant offering. If we are engaged in a public offering and the redemption price calculated in accordance with the guidelines set forth above would result in a price that is higher than the then-current public offering price of our common shares, then the redemption price will be reduced and will be equal to the then-current public offering price of our common shares.

In the event that funds are insufficient to repurchase all of the shares for which repurchase requests have been submitted in a particular month and our board of directors determines that we will redeem shares in that month, then shares will be repurchased on a pro rata basis and the portion of any unfulfilled repurchase request will be held and considered for redemption until the next month unless withdrawn by the stockholder.

Issuer Redemptions of Equity Securities

None. We have not received any requests for redemption.

Item 6. Selected Financial Data

The following selected consolidated financial data for the year ended December 31, 2014 and the period from July 31, 2013 (date of inception) through December 31, 2013 are not indicative of what is expected in future periods as we did not meet our minimum offering requirements until September 2014 (and have not met our minimum offering requirements with respect to Pennsylvania and Washington) and did not make our first real estate investment until December 2014. Further, the following selected consolidated financial data are qualified by reference to and should be read in conjunction with our Consolidated Financial Statements and Notes thereto and "Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations" below.

	2014		2013 ⁽¹⁾
Operating Data:			
Revenues	\$	94,327	\$
Depreciation and amortization	\$	49,288	\$ —
Asset management and acquisition fees	\$	570,375	\$ —
Organizational expenses	\$	63,164	\$
General and administrative	\$	554,476	\$ —
Net income (loss)	\$	(1,327,663)	\$ _
Net (income) loss attributable to noncontrolling interests	\$	186,940	\$
Net income (loss) attributable to common stockholders	\$	(1,140,723)	\$ _
Basic and diluted income (loss) per Class A common share	\$	(14.67)	\$
Distributions declared per common share	\$	0.15	\$
Weighted average common shares outstanding - basic and diluted		77,779	1,113
Balance Sheet Data:			
Total investment property	\$	21,355,421	\$ —
Cash and cash equivalents	\$	2,726,742	\$ 200,000
Total assets	\$	28,550,814	\$ 200,000
Long-term obligations	\$	24,200,000	\$ _

(1) For the period from July 31, 2013 (date of inception) through December 31, 2013 for operating data and as of December 31, 2013 for balance sheet data.

Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations

You should read the following discussion and analysis together with our consolidated financial statements and notes thereto included in this Annual Report on Form 10-K. The following information contains forward-looking statements, which are subject to risks and uncertainties. Should one or more of these risks or uncertainties materialize, actual results may differ materially from those expressed or implied by the forward-looking statements. Please see "Special Note Regarding Forward-Looking Statements" above for a description of these risks and uncertainties.

Executive Summary

Hines Global REIT II, Inc. ("Hines Global II" and, together with its consolidated subsidiaries, "we," "us" or the "Company") was formed in July 2013 to invest in a diversified portfolio of quality commercial real estate properties and other real estate investments located throughout the United States and internationally. In August 2014, we commenced a \$2.5 billion initial public offering of our common stock (the "Offering"). On September 26, 2014, we received an investment of \$2.0 million from an affiliate of the Advisor and achieved the minimum offering requirements pursuant to the terms of the Offering in all states except for Pennsylvania and Washington, which have higher minimum offering requirements. As of March 18, 2015, we had received gross offering proceeds of \$18.3 million from the sale of approximately 1.9 million common shares.

We intend to meet our primary investment objectives by investing in a portfolio of real estate properties and other real estate investments that relate to properties that are generally diversified by geographic area, lease expirations and tenant industries. These investments could include a variety of asset types in the US and internationally such as office, retail, industrial, multi-family residential and hospitality or leisure. In addition, we may invest in operating properties, properties under development, and undeveloped properties such as land. Other real estate investments may include equity or debt interests, including securities, in other real estate entities and debt related to properties such as mortgages, mezzanine loans, B-notes, bridge loans, construction loans and securitized debt. We expect to fund these acquisitions primarily with proceeds from the Offering and debt financing.

In December 2014, we acquired 2819 Loker Avenue East, a 100% leased, Class A industrial building in Carlsbad, California for \$25.4 million. In March 2015, we closed on our second real estate investment, Bishop's Square, a 100% leased, Class A office building in Dublin, Ireland for \notin 92.0 million (approximately \$103.5 million assuming a rate of \$1.12 per EUR as of the transaction date). As we are in the early stages of our offering, these real estate investments were primarily purchased with proceeds from a \$75.0 million unsecured credit facility with Hines (the "Hines Credit Facility") and a \notin 55.2 million secured credit facility (\$62.1 million using \$1.12 per Euro as of the transaction date) secured by our interest in Bishop's Square.

Critical Accounting Policies

Our discussion and analysis of financial condition and results of operations is based on our consolidated financial statements statements, which have been prepared in accordance with GAAP. The preparation of the consolidated financial statements requires us to make estimates and judgments that affect the reported amounts of assets, liabilities and contingencies as of the date of the financial statements and the reported amounts of revenues and expenses during the reporting periods. We evaluate our assumptions and estimates on an ongoing basis. We base our estimates on historical experience and on various other assumptions that we believe to be reasonable under the circumstances. Additionally, application of our accounting policies involves exercising judgments regarding assumptions as to future uncertainties. Actual results may differ from these estimates under different assumptions or conditions. The following is a discussion of our critical accounting policies, for a discussion of all of our significant accounting policies, see <u>Note 2</u> — Summary of Significant Accounting Policies, to the accompanying consolidated financial statements.

Basis of Presentation

Our financial statements are expected to include the accounts of Hines Global II and the Operating Partnership (over which we exercise financial and operating control). All intercompany balances and transactions are eliminated in consolidation.

We may enter into various joint venture agreements with unrelated third parties to hold or develop real estate assets. We will evaluate the need to consolidate joint ventures and will consolidate those that are determined to be variable interest entities for which we are the primary beneficiary. We will also consolidate joint ventures that are not determined to be variable interest entities, but for which we exercise control over major operating decisions through substantive participation rights, such as approval of budgets, selection of property managers, asset management, investment activity and changes in financing. The analysis required to identify VIEs and primary beneficiaries is complex and requires substantial judgment. Accordingly, we

believe the decisions made to choose an appropriate accounting framework are critical. To date, we have not entered into any such joint ventures.

Investment Property and Lease Intangibles

When we acquire a property, we allocate the purchase price of the acquisition based upon our assessment of the fair value of various components, including to land and building, land, building and improvements, and intangible lease assets and liabilities. Fair value determinations are based on estimated cash flow projections that utilize discount and/or capitalization rates, as well as certain available market information. The fair value of land, building and improvements considers the value of the property as if it were vacant. The fair value of intangible lease assets is based on our evaluation of the specific characteristics of each lease. Factors considered include estimates of carrying costs during hypothetical expected lease-up periods, current market conditions and market rates, the customer's credit quality and costs to execute similar leases. The fair value of out-of-market leases is calculated as the present value (using a discount rate that reflects the risks associated with the leases) of the difference between the contractual amounts to be paid pursuant to each in-place lease and our estimate of fair market rates during the expected lease-up periods, depending on local market conditions. In estimating costs to execute similar leases are subject to change until such information is finalized, which will occur no later than 12 months after the acquisition date.

Real estate assets are reviewed for impairment each reporting period if events or changes in circumstances indicate that the carrying amount of the individual property may not be recoverable. In such an event, a comparison will be made of the current and projected operating cash flows and expected proceeds from the eventual disposition of each property on an undiscounted basis to the carrying amount of such property. If the carrying amount exceeds the undiscounted cash flows, it would be written down to the estimated fair value to reflect impairment in the value of the asset. The determination of whether investment property is impaired requires a significant amount of judgment by management and is based on the best information available to management at the time of the evaluation.

Deferred Leasing Costs

We consider a number of different factors to evaluate whether we or the lessee is the owner of the tenant improvements for accounting purposes. These factors include: (i) whether the lease stipulates how and on what a tenant improvement allowance may be spent; (ii) whether the tenant or landlord retains legal title to the improvements; (iii) the uniqueness of the improvements; (iv) the expected economic life of the tenant improvements relative to the term of the lease; and (v) who constructs or directs the construction of the improvements. The determination of who owns the tenant improvements for accounting purposes is subject to significant judgment. In making that determination, we consider all of the above factors. No one factor, however, necessarily establishes any determination.

Revenue Recognition and Valuation of Receivables

We are required to recognize minimum rent revenues on a straight-line basis over the terms of tenant leases, including rent holidays and bargain renewal options, if any. Revenues associated with tenant reimbursements are recognized in the period in which the expenses are incurred based upon the tenant's lease provision. Leases are not uniform in dealing with such cost reimbursements and there are many variations in the computation. We make quarterly accrual adjustments, positive or negative, to tenant reimbursement revenue to adjust the recorded amounts to our best estimate of the final amounts to be billed and collected with respect to the cost reimbursements. Revenues relating to lease termination fees are recognized on a straightline basis amortized from the time that a tenant's right to occupy the leased space is modified through the end of the revised lease term and are included in other revenue in the accompanying consolidated statements of operations. To the extent our leases provide for rental increases at specified intervals, we will record a receivable for rent not yet due under the lease terms. Accordingly, our management must determine, in its judgment, to what extent the unbilled rent receivable applicable to each specific tenant is collectible. We review unbilled rent receivables on a quarterly basis and take into consideration the tenant's payment history, the financial condition of the tenant, business conditions in the industry in which the tenant operates and economic conditions in the area in which the property is located. In the event that the collectability of unbilled rent with respect to any given tenant is in doubt, we would be required to record an increase in our allowance for doubtful accounts or record a direct write-off of the specific rent receivable, which would have an adverse effect on our net income for the year in which the reserve is increased or the direct write-off is recorded and would decrease our total assets and stockholders' equity.

Recent Accounting Pronouncements

See <u>Note 2</u> — Summary of Significant Accounting Policies for a discussion regarding recent accounting pronouncements and the potential impact, if any, on our financial statements.

Financial Condition, Liquidity and Capital Resources

Our principal demands for funds are to purchase real estate properties and make other real estate investments, for the payment of operating expenses and distributions, and for the payment of principal and interest on any indebtedness we incur. Generally, we expect to meet operating cash needs from our cash flows from operating activities, and we expect to meet cash needs for acquisitions and investments from the net proceeds of the Offering and from debt proceeds.

We expect that once we have fully invested the proceeds of the Offering and other potential subsequent offerings, our debt financing, including our pro rata share of the debt financing of entities in which we invest, will be in the range of approximately 40% to 60% of the aggregate value of our real estate investments and other assets. Financing for acquisitions and investments may be obtained at the time an asset is acquired or an investment is made or at such later time as determined to be appropriate. In addition, debt financing may be used from time to time for property improvements, lease inducements, tenant improvements and other working capital needs. Additionally, the amount of debt placed on an individual property or related to a particular investment, including our pro rata share of the amount of debt incurred by an individual entity in which we invest, may be less than 40% or more than 60% of the value of such property/investment or the value of the assets owned by such entity, depending on market conditions and other factors. Our aggregate borrowings, secured and unsecured, must be reasonable in relation to our net assets and must be reviewed by our board of directors at least quarterly.

Our charter limits our borrowing to 300% of our net assets (which approximates 75% of the cost of our assets) unless any excess borrowing is approved by a majority of our independent directors and is disclosed to our stockholders in our next quarterly report along with justification for the excess. In November 2014, our board of directors, including all of our independent directors, approved the \$75.0 million Hines Credit Facility to be used to fund acquisitions and other working capital needs. Additionally, in December 2014, our board of directors, including all of our independent directors, approved the \$75.0 million Hines Credit Facility to be used to fund acquisitions and other working capital needs. Additionally, in December 2014, our board of directors, including all of our independent directors, approved a \$24.2 million borrowing under this credit facility to fund our acquisition of 2819 Loker Avenue East, an industrial building located in Carlsbad, California. This borrowing represents approximately 95% of the contract purchase price of 2819 Loker Avenue East. Further, subsequent to December 31, 2014, our board of directors, including all of our independent directors, approved \$45.2 million of additional borrowings under the Hines Credit Facility and a €55.2 million secured credit facility (\$62.1 million using \$1.12 per Euro as of the transaction date) for the purchase of Bishop's Square in March 2015. In total, these borrowings represent approximately 104% of the cost of Bishop's Square based on the contract purchase price. In March 2015, we repaid \$17.0 million in borrowings under the Hines Credit Facility using proceeds from the Offering.

Notwithstanding the above, depending on market conditions and other factors, we may choose not to place debt on our portfolio or our assets and may choose not to borrow to finance our operations or to acquire properties. Any indebtedness we do incur will likely be subject to continuing covenants, and we will likely be required to make continuing representations and warranties about our Company in connection with such debt. Moreover, some or all of our debt may be secured by some or all of our assets. If we default in the payment of interest or principal on any such debt, breach any representation or warranty in connection with any borrowing or violate any covenant in any loan document, our lender may accelerate the maturity of such debt requiring us to immediately repay all outstanding principal. If we are unable to make such payment, our lender could foreclose on our assets that are pledged as collateral to such lender. The lender could also sue us or force us into bankruptcy. Any such event would have a material adverse effect on the value of an investment in our common shares.

The discussions below provide additional details regarding our cash flows.

Cash Flows from Operating Activities

Our real estate property generates cash flow in the form of rental revenues, which are used to pay direct leasing costs, property-level operating expenses and interest payments. Property-level operating expenses consist primarily of salaries and wages of property management personnel, utilities, cleaning, insurance, security and building maintenance costs, property management and leasing fees, and property taxes. Additionally, we incur general and administrative expenses, acquisition fees and expenses and asset management fees.

Net cash used in operating activities for the year ended December 31, 2014 was \$241,298, which was primarily related to general and administrative expenses paid in 2014.

Cash Flows from Investing Activities

Net cash used in investing activities was primarily due to the payment of \$25.1 million related to the acquisition of 2819 Loker Avenue East and its related lease intangibles.

Cash Flows from Financing Activities

Initial Public Offering

We commenced the Offering in August 2014 and met our minimum offering requirements for every state, except Pennsylvania and Washington, in September 2014. During the year ended December 31, 2014, we raised proceeds of \$3.8 million from the Offering, excluding proceeds from the distribution reinvestment plan.

In addition to the investing activities described previously, we use proceeds from the Offering to make certain payments to the Advisor, our Dealer Manager and Hines and their affiliates during the various phases of our organization and operation. During the organization and offering stage, these include payments to our Dealer Manager for selling commissions and the dealer manager fee and payments to our Advisor for reimbursement of issuer costs. During the year ended December 31, 2014, we made payments of \$95,135 for selling commissions, dealer manager fees and issuer costs related to the Offering.

Distributions

Our board of directors authorized us to declare distributions for the period from October 1, 2014 through March 31, 2015. Distributions for shares of our common shares will be calculated based on stockholders of record each day in an amount equal to \$0.001575342 per share, per day. All distributions were or will be paid in cash or reinvested in shares of our common stock for those participating in our distribution reinvestment plan and will be paid or issued, respectively, on the first business day following the completion of the month to which they relate.

In addition to the distributions described above, our board of directors has authorized special stock dividends as of daily record dates for the period from October 1, 2014 through March 31, 2015. Stock dividends for our common shares will be calculated based on stockholders of record each day in an amount equal to 0.0000273973 of a share of common stock per share, per day. Shares issued related to our stock dividends were or will be issued on the first business day of the month following the quarter to which they relate.

We have not generated sufficient cash flow from operations to fully fund distributions paid. Therefore, particularly in the earlier part of the Offering, some or all of our distributions have been and may continue to be paid from other sources, such as proceeds from our debt financings, proceeds from the Offering, cash advances by our Advisor, cash resulting from a waiver or deferral of fees and/or proceeds from the sale of assets. For example, for the year ended December 31, 2014, 100% of our distributions were funded with cash flows from financing activities, which includes offering proceeds. In addition, the Advisor waived \$16,258 of asset management fees otherwise payable to it for the three months ended December 31, 2014, in order to more closely align the amount of distributions paid with our operations. We have not placed a cap on the amount of our distributions that may be paid from any of these sources.

Cash Distributions

The following table contains additional information regarding cash distributions declared to our stockholders and noncontrolling interests (HALP II) for each of the quarters during 2014, as well as the sources of payments:

			Stock	holders				ncontrolling Interests		Sour	ces	
Distributions for the Quarters Ended ⁽¹⁾	Dist	Cash ributions	Distributions Total Reinvested Declared			Tot	Total Declared		ash Flows From Operating Activities	Cash Flow Financ Activitie	ing	
December 31, 2014	\$	40,377	\$	1,841	\$	42,218	\$	3,060	\$	%	\$ 45,278	100%

(1) There were not any cash dividends declared for any other quarters during 2014.

(2) Cash flows from financing activities primarily includes proceeds from the Offering.

Stock Dividends

The following table outlines our total stock dividends declared for the quarter ended December 31, 2014.

	Stockh	olders	
Dividends for the Quarter Ended ⁽¹⁾	Issuance Date	Shares	Amount ⁽²⁾
December 31, 2014	1/1/2015	737	\$ 7,367

(1) There were not any stock dividends declared for any other quarters during 2014.

(2) Amount based on our \$10.00 per share offering price.

Debt Financings

In December 2014, in connection with the acquisition of 2819 Loker Avenue East, we borrowed \$24.2 million under the Hines Credit Facility.

Contributions From Noncontrolling Interests

In September 2013, HALP II made its initial capital contribution to the Operating Partnership of \$190,000 and, accordingly, owned a 95.0% limited partner interest in the Operating Partnership. As of December 31, 2014, HALP II owned a 4.8% limited partner interest in the Operating Partnership.

Results of Operations

Our results of operations for the year ended December 31, 2014 and the period from July 31, 2013 (date of inception) through December 31, 2013 are not indicative of those expected in future periods as we did not meet our minimum offering requirements until September 2014 (and have not met our minimum offering requirements with respect to Pennsylvania and Washington) and did not make our first real estate investment until December 2014. Amounts recorded in our consolidated statements of operations for the year ended December 31, 2014 are due to the following:

- Total revenues, property operating expenses, real property taxes, property management fees, depreciation and amortization, and interest expense relate to the operation of 2819 Loker Avenue East.
- Acquisition-related expenses represent costs incurred on 2819 Loker Avenue East and properties which we may acquire in future periods.
- We expect to pay monthly asset management fees to our Advisor based on an annual fee equal to 0.75% of the cost of our real estate investments. As described previously, our Advisor has agreed to waive asset management fees for the quarter ended December 31, 2014 to the extent that our MFFO does not exceed our distributions for the quarter ended December 31, 2014. Accordingly, our Advisor waived the entire asset management fee for the quarter ended December 31, 2014 totaling \$16,258.
- We pay our Advisor acquisition fees equal to 2.25% of the purchase price of our real estate investments. As a result, we incurred a \$570,375 acquisition fee related to our acquisition of 2819 Loker Avenue East in December 2014.
- General and administrative expenses for the year ended December 31, 2014 primarily consist of legal and accounting fees, costs and expenses associated with our board of directors, and insurance costs. Certain of these costs are variable and will increase in the future as we continue to raise capital and make additional real estate investments.
- Total loss attributable to noncontrolling interests for the year ended December 31, 2014 relates to our allocation of our net loss to HALP II based on its ownership in the Operating Partnership.

Funds from Operations and Modified Funds from Operations

Funds from Operations ("FFO") is a non-GAAP financial performance measure defined by the National Association of Real Estate Investment Trusts ("NAREIT") and is widely recognized by investors and analysts as one measure of operating performance of a real estate company. FFO excludes items such as real estate depreciation and amortization. Depreciation and amortization, as applied in accordance with GAAP, implicitly assumes that the value of real estate assets diminishes predictably

over time and also assumes that such assets are adequately maintained and renovated as required in order to maintain their value. Since real estate values have historically risen or fallen with market conditions such as occupancy rates, rental rates, inflation, interest rates, the business cycle, unemployment and consumer spending, it is management's view, and we believe the view of many industry investors and analysts, that the presentation of operating results for real estate companies using historical cost accounting alone is insufficient. In addition, FFO excludes gains and losses from the sale of real estate and impairment charges related to depreciable real estate assets and in-substance real estate equity investments, which we believe provides management and investors with a helpful additional measure of the historical performance of our real estate portfolio, as it allows for comparisons, year to year, that reflect the impact on operations from trends in items such as occupancy rates, rental rates, operating costs, general and administrative expenses and interest costs. A property will be evaluated for impairment if events or circumstances indicate that the carrying amount may not be recoverable (i.e. the carrying amount exceeds the total estimated undiscounted future cash flows from the property). Undiscounted future cash flows are based on anticipated operating performance, including estimated future net rental and lease revenues, net proceeds on the sale of the property, and certain other ancillary cash flows. While impairment charges are excluded from the calculation of FFO as described above, stockholders are cautioned that due to the limited term of our operations, it could be difficult to recover any impairment charges.

In addition to FFO, management uses MFFO, as defined by the Investment Program Association, (the "IPA"), as a non-GAAP supplemental financial performance measure to evaluate our operating performance. The IPA has recommended the use of MFFO as a supplemental measure for publicly registered, non-listed REITs to enhance the assessment of the operating performance of a non-listed REIT. MFFO is not equivalent to our net income or loss as determined under GAAP, and MFFO may not be useful as a measure of the long-term operating performance of our investments or as a comparative measure to other publicly registered, non-listed REITs if we do not continue to operate with a limited life and targeted exit strategy, as currently intended and described herein. MFFO includes funds generated by the operations of our real estate investments and funds used in our corporate-level operations. MFFO is based on FFO, but includes certain additional adjustments which we believe are appropriate. Such items include reversing the effects of straight-line rent revenue recognition, fair value adjustments to derivative instruments that do not qualify for hedge accounting treatment and certain other items as described below. Some of these adjustments are necessary to address changes in the accounting and reporting rules under GAAP such as the accounting for acquisition-related expenses from a capitalization/depreciation model to an expensed-as-incurred model that were put into effect in 2009 and other changes to GAAP rules for real estate subsequent to the establishment of NAREIT's definition of FFO. These changes in the accounting and reporting rules under GAAP affected all industries, and as a result of these changes, acquisition fees and expenses are typically accounted for as operating expenses under GAAP. Management believes these fees and expenses do not affect our overall long-term operating performance. These changes also have prompted a significant increase in the magnitude of non-cash and non-operating items included in FFO, as defined. Such items include amortization of out-of-market lease intangible assets and liabilities and certain tenant incentives.

Other adjustments included in MFFO are necessary to address issues that are common to publicly registered, non-listed REITs. Publicly registered, non-listed REITs typically have a significant amount of acquisition activity and are substantially more dynamic during their initial years of investment and operations. While other start-up entities may also experience significant acquisition activity during their initial years, we believe that non-listed REITs like us are unique in that they have a limited life with targeted exit strategies within a relatively limited time frame after the acquisition activity ceases. We will use the proceeds raised in our offerings to make real estate investments, and intend to begin the process of considering our alternatives for the execution of a Liquidity Event (i.e., a sale of our assets, our sale or merger, a listing of our shares on a national securities exchange, or another similar transaction) eight to ten years following the commencement of the Offering. Thus, as a limited life REIT we will not continuously purchase assets and will have a limited life.

The purchase of properties, and the corresponding expenses associated with that process, including acquisition fees and expenses, is a key operational feature of our business plan to generate operational income and cash flows in order to make distributions to our stockholders. MFFO excludes acquisition fees payable to our Advisor and acquisition expenses. Under GAAP, acquisition fees and expenses are characterized as operating expenses in determining operating net income. These expenses are paid in cash by us, and therefore such funds will not be available to distribute to our stockholders. All paid and accrued acquisition fees and expenses with respect to the acquisition of a property negatively impact our operating performance during the period in which the property is acquired and will have negative effects on returns to our stockholders, the potential for future distributions, and future cash flows, unless earnings from operations or net sales proceeds from the disposition of other properties are generated to cover the purchase price of the property, the related acquisition fees and expenses and other costs related to such property. In addition, if we acquire a property after all offering proceeds from the Offering have been invested, there will not be any offering proceeds to pay the corresponding acquisition-related costs. Accordingly, unless our Advisor determines to waive the payment of any then-outstanding acquisition-related costs otherwise payable to the Advisor, such costs will be paid from additional debt, operational earnings or cash flow, net proceeds from the sale of properties, or ancillary cash flows. Therefore, MFFO may not be an accurate indicator of our operating performance, especially during

periods in which properties are being acquired. Since MFFO excludes acquisition fees and expenses, MFFO would only be comparable to the operations of non-listed REITs that have completed their acquisition activity and have other similar operating characteristics.

Management uses MFFO to evaluate the financial performance of our investment portfolio, including the impact of potential future investments. In addition, management uses MFFO to evaluate and establish our distribution policy and the sustainability thereof. Further, we believe MFFO is one of several measures that may be useful to investors in evaluating the potential performance of our portfolio following the conclusion of the acquisition phase, as it excludes acquisition fees and expenses, as described herein.

MFFO has limitations as a performance measure in an offering such as ours where the price of a share of common stock is a stated value and there is no net asset value determination during the offering stage and for a period thereafter. MFFO is useful in assisting management and investors in assessing the sustainability (that is, the capacity to continue to be maintained) of operating performance in future operating periods, and in particular, after the offering and acquisition stages are complete and net asset value is disclosed. MFFO is not a useful measure in evaluating net asset value because impairments are taken into account in determining net asset value but not in determining MFFO.

FFO and MFFO should not be construed to be more relevant or accurate than the current GAAP methodology in calculating net income or in its applicability in evaluating our operating performance. In addition, FFO and MFFO should not be considered as alternatives to net income (loss) or income (loss) from continuing operations as an indication of our performance or as alternatives to cash flows from operating activities as an indication of our liquidity, but rather should be reviewed in conjunction with these and other GAAP measurements. Further, FFO and MFFO are not intended to be used as liquidity measures indicative of cash flow available to fund our cash needs, including our ability to make distributions to our stockholders. Please see the limitations listed below associated with the use of MFFO:

- As we are currently in the acquisition phase of our life cycle, acquisition costs and other adjustments that are increases to MFFO are, and may continue to be, a significant use of cash and dilutive to the value of an investment in our shares.
- MFFO excludes acquisition fees payable to our Advisor and acquisition expenses. Although these amounts reduce net income, we generally fund such costs with proceeds from our offering and acquisition-related indebtedness and do not consider these fees and expenses in the evaluation of our operating performance and determining MFFO.
- We utilize the definition of FFO as set forth by NAREIT and the definition of MFFO as set forth by the IPA. Our FFO and MFFO as presented may not be comparable to amounts calculated by other REITs, if they use different approaches.
- Our business is subject to volatility in the real estate markets and general economic conditions, and adverse changes in those conditions could have a material adverse impact on our business, results of operations and MFFO. Accordingly, the predictive nature of MFFO is uncertain and past performance may not be indicative of future results.

Neither the SEC, NAREIT nor any regulatory body has passed judgment on the acceptability of the adjustments that we use to calculate FFO or MFFO. In the future, the SEC, NAREIT or a regulatory body may decide to standardize the allowable adjustments across the non-listed REIT industry and we would have to adjust our calculation and characterization of FFO or MFFO.

The following section presents our calculation of FFO and MFFO and provides additional information related to our operations for the year ended December 31, 2014 and the period from inception through December 31, 2014. As we are in the capital raising and acquisition phase of our operations, FFO and MFFO are not useful in comparing operations for the two periods presented below. We expect revenues and expenses to increase in future periods as we raise additional offering proceeds and use them to acquire additional investments.

	De	Year Ended cember 31, 2014	(date)	d from July 31, 2013 of inception) through ecember 31, 2014
Net income (loss)	\$	(1,327,663)	\$	(1,327,663)
Depreciation and amortization ⁽¹⁾		49,288		49,288
Adjustments for noncontrolling interests ⁽²⁾		184,584		184,584
Funds from operations		(1,093,791)		(1,093,791)
Other components of revenues and expenses ⁽³⁾		(5,895)		(5,895)
Acquisition fees and expenses ⁽⁴⁾		698,829		698,829
Adjustments for noncontrolling interests ⁽²⁾		(33,122)		(33,122)
Modified Funds From Operations	\$	(433,979)	\$	(433,979)
Basic and Diluted Loss Per Common Share Attributable to Common Stockholders	\$	(14.67)	\$	(19.29)
Funds From Operations Per Common Share	\$	(14.06)	\$	(18.50)
Modified Funds From Operations Per Common Share	\$	(5.58)	\$	(7.34)
Weighted Average Shares Outstanding		77,779		59,139

Notes to the table:

- (1) Represents the depreciation and amortization of real estate assets. Historical cost accounting for real estate assets in accordance with GAAP implicitly assumes that the value of real estate assets diminishes predictably over time. Since real estate values have historically risen or fallen with market conditions, we believe that such depreciation and amortization may be of limited relevance in evaluating current operating performance and, as such, these items are excluded from our determination of FFO.
- (2) Includes income attributable to noncontrolling interests and all adjustments to eliminate the noncontrolling interests' share of the adjustments to convert our net loss to FFO and MFFO.
- (3) Includes the following components of revenues and expenses that we do not consider in evaluating our operating performance and determining MFFO for the year ended December 31, 2014 and the period from inception through December 31, 2013:

	 ar Ended ber 31, 2014	(date	od from July 31, 2013 of inception) through becember 31, 2014
Straight-line rent adjustment ^(a)	\$ (5,895)	\$	(5,895)
	\$ (5,895)	\$	(5,895)

(a) Represents the adjustments to rental revenue as required by GAAP to recognize minimum lease payments on a straightline basis over the respective lease terms. We have excluded these adjustments from our evaluation of our operating performance and in determining MFFO because we believe that the rent that is billable during the current period is a more relevant measure of our operating performance for such period.

(4) Represents acquisition expenses and acquisition fees paid to our Advisor that are expensed in our consolidated statements of operations. We fund such costs with proceeds from our offering, and therefore do not consider these expenses in evaluating our operating performance and determining MFFO.

As noted previously, our cash flows from operations have been and may continue to be insufficient to fund distributions to stockholders. We may continue to choose to use proceeds from our debt financings, proceeds from the Offering, cash advances by our Advisor, cash resulting from a waiver or deferral of fees and/or proceeds from the sale of assets to fund distributions to

our stockholders. We funded 100% of distributions for the year ended December 31, 2014 with cash flows from financing activities, which includes offering proceeds. The Advisor waived \$16,258 of asset management fees payable to it for the three months ended December 31, 2014. We have not placed a cap on the amount of our distributions that may be paid from sources other than cash flows from operations, including proceeds from our debt financings, proceeds from our public offerings, cash advances by our Advisor and cash resulting from a waiver or deferral of fees.

Related-Party Transactions and Agreements

We have entered into agreements with the Advisor, Dealer Manager and Hines or its affiliates, whereby we pay certain fees and reimbursements to these entities during the various phases of our organization and operation. During the organization and offering stage, these include payments to our Dealer Manager for selling commissions, the dealer manager fee and payments to our Advisor for reimbursement of issuer costs. During the acquisition and operational stages, these include payments for certain services related to acquisitions, financing and management of our investments and operations provided to us by our Advisor and Hines and its affiliates pursuant to various agreements we have entered into with these entities. See <u>Note 7</u> — Related Party Transactions in the Notes to the Consolidated Financial Statements and Part III, Item 13—Certain Relationships and Related Transactions, and Director Independence contained elsewhere in this Annual Report on Form 10-K for additional information concerning our related-party transactions and agreements.

Off-Balance Sheet Arrangements

As of December 31, 2014 and 2013, we had no off-balance sheet arrangements that have or are reasonably likely to have a current or future effect on our financial condition, changes in financial condition, revenues or expenses, results of operations, liquidity, capital expenditures or capital resources.

Contractual Obligations

The following table lists our known contractual obligations as of December 31, 2014. Specifically included are our obligations under long-term debt agreements:

		Payments due by Period										
Contractual Obligations	Less Than 1 Year	1-3 Years	3-5 Years	More Than 5 Years	Total							
Hines Credit Facility ⁽¹⁾	\$ 24,461,360	\$ —	\$ _	\$	\$ 24,461,360							

(1) Includes principal and interest payments outstanding as of December 31, 2014.

Recent Developments and Subsequent Events

Bishop's Square

In March 2015, we acquired Bishop's Square, a Class A office building, located in the central business district of Dublin, Ireland. The contract purchase price was \notin 92.0 million (approximately \$103.5 million assuming a rate of \$1.12 per EUR as of the transaction date), exclusive of transaction costs and working capital reserves. The building contains approximately 153,529 square feet of net rentable area that is 100% leased to five tenants. As a result of the acquisition of Bishop's Square, we incurred approximately \$2.3 million in acquisition fees payable to our Advisor.

We have not concluded our accounting for this recent acquisition, but we expect that the purchase price will primarily be allocated to building, land and intangible lease assets and liabilities.

In connection with the acquisition of Bishop's Square, we, through a wholly-owned subsidiary, entered into a secured facility agreement (the "Bishop's Square Facility") with DekaBank Deutsche Girozentrale totaling €55.2 million (approximately \$62.1 million assuming a rate of \$1.12 per EUR as of the transaction date). The Bishop's Square Facility requires quarterly interest payments beginning in July 2015 and repayment of principal upon the maturity of the Bishop's Square Facility on March 2, 2022. The Bishop's Square Facility has a floating interest rate of EURIBOR plus 1.30%. At the date of the initial draw, the interest rate was 1.38%. The Bishop's Square Facility may be repaid in full prior to maturity, subject to a prepayment premium if it is repaid in the first three years. In accordance with the terms of the facility agreement,

we entered into a \notin 55.2 million (approximately % 2.1 million assuming a rate of \$1.12 per EUR as of the transaction date) three-year interest rate cap at 2% to limit exposure to interest rate fluctuations.

Item 7A. Quantitative and Qualitative Disclosures About Market Risk

Market risk includes risks that arise from changes in interest rates, foreign currency exchange rates, commodity prices, equity prices and other market changes that affect market-sensitive instruments. In pursuing our business plan, we believe that interest rate risk, currency risk and real estate valuation risk are the primary market risks to which we are exposed. As of December 31, 2014, we were exposed to the market risk listed below.

Interest Rate Risk

We are exposed to the effects of interest rate changes primarily as a result of debt used to maintain liquidity and fund expansion of our real estate investment portfolio and operations. As of December 31, 2014, we had \$24.2 million of debt outstanding under the Hines Credit Facility, which requires interest at a variable rate. If interest rates were to increase by 1%, we would incur an additional \$242,000 in interest expense. See <u>Note 5</u> — Debt Financing in the Notes to the Consolidated Financial Statements for more information concerning our outstanding debt.

Other Risks

In our initial quarters of operations, and from time to time thereafter, our cash flow from operations may be insufficient to fund distributions to stockholders. We may choose to use advances, deferrals or waivers of fees, if available, from our Advisor or affiliates, borrowings and/or proceeds of the Offering or other sources to fund distributions to our stockholders. As described elsewhere in this Annual Report on Form 10-K, our Advisor has agreed to waive \$16,258 of asset management fees payable to it for the year ended December 31, 2014, in order to more closely align the amount of distributions paid with our operations. Our Advisor has also agreed to waive the asset management fees otherwise payable to it for each of the quarters ended March 31, 2015 and June 30, 2015, to the extent that our MFFO for such quarters, as disclosed in our Quarterly Report on Form 10-Q, amounts to less than 100% of the aggregate distributions declared to our stockholders for each respective quarter. There can be no assurances that our Advisor will continue this waiver, and if not, cash available to pay distributions in future periods may be reduced.

We invest proceeds we receive from our public offerings in short-term, highly-liquid investments until we use such funds to make real estate investments. Although we do not expect that income we earn on these temporary investments will be substantial, our earnings will be subject to the fluctuations of interest rates and their effect on these investments.

Item 8. Financial Statements and Supplementary Data

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors and Stockholders of Hines Global REIT II, Inc. Houston, Texas

We have audited the accompanying consolidated balance sheets of Hines Global REIT II, Inc. and subsidiaries (the "Company") as of December 31, 2014 and 2013, and the related consolidated statements of operations, equity, and cash flows for the year ended December 31, 2014 and for the period from July 31, 2013 (date of inception) through December 31, 2013. Our audits also included the financial statement schedule listed in the Index at Item 15. These financial statements and financial statement schedule are the responsibility of the Company's management. Our responsibility is to express an opinion on the financial statements and financial statement schedule based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. Our audits included consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, such consolidated financial statements present fairly, in all material respects, the financial position of Hines Global REIT II, Inc. and subsidiaries as of December 31, 2014 and 2013, and the results of their operations and their cash flows for the year ended December 31, 2014 and for the period from July 31, 2013 (date of inception) through December 31, 2013, in conformity with accounting principles generally accepted in the United States of America. Also, in our opinion, such financial statement schedule, when considered in relation to the basic consolidated financial statements taken as a whole, presents fairly in all material respects the information set forth therein.

/s/ Deloitte & Touche LLP

Houston, Texas March 25, 2015

HINES GLOBAL REIT II, INC. CONSOLIDATED BALANCE SHEETS As of December 31, 2014 and 2013

		2014		2013
ASSETS				
Investment property, net	\$	21,355,421	\$	
Cash and cash equivalents		2,726,742		200,000
Tenant and other receivables, net		68,160		
Intangible lease assets, net		3,945,291		—
Deferred financing costs, net		8,925		
Other assets		446,275		_
Total assets	\$	28,550,814	\$	200,000
LIABILITIES AND EQUITY	_		_	
Liabilities:				
Accounts payable and accrued expenses	\$	259,111	\$	_
Due to affiliates		5,213,634		—
Distributions payable		18,371		_
Note payable to affiliate		24,200,000		—
Total liabilities	\$	29,691,116	\$	
Commitments and contingencies (Note 9)		—		_
Equity:				
Stockholders' equity (deficit):				
Preferred shares, \$0.001 par value; 500,000,000 preferred shares authorized, none issued or outstanding as of December 31, 2014 and 2013		_		
Class A common stock, \$0.001 par value; 1,500,000,000 authorized; 420,541 and 1,111 issued and outstanding as of December 31, 2014 and 2013, respectively		421		1
Additional paid-in capital				9,999
Accumulated deficit		(1,140,723)		
Total stockholders' equity (deficit)		(1,140,302)	_	10,000
Noncontrolling interests				190,000
Total equity (deficit)		(1,140,302)	_	200,000
Total liabilities and equity	\$	28,550,814	\$	200,000

HINES GLOBAL REIT II, INC. CONSOLIDATED STATEMENTS OF OPERATIONS For the Year Ended December 31, 2014 and for the Period from July 31, 2013 (date of inception) through December 31, 2013

Revenues:Rental revenue§94,327§—Total revenues94,327—Expenses: $Property operating expenses21,919—Real property taxes9,460—Property management fees1,769—Depreciation and amortization49,288—Acquisition related expenses131,454—Asset management and acquisition fees570,375—Organizational expenses63,164—General and administrative expenses554,476—Income (loss) before other income (expenses)(1,307,578)—Interest income243—Net income (loss)(1,327,663)—Net (income) loss attributable to noncontrolling interests186,940—Net income (loss) per Class A common shares§(1,407,223)§Weighted average number of common shares outstanding77,7791,113$		 r Ended ber 31, 2014	Period from July 31, 2013 (date of inception) through December 31, 2013		
Total revenues94,327—Expenses:—Property operating expenses21,919—Real property taxes9,460—Property management fees1,769—Depreciation and amortization49,288—Acquisition related expenses131,454—Asset management and acquisition fees570,375—Organizational expenses63,164—General and administrative expenses554,476—Total expenses1,401,905—Income (loss) before other income (expenses)(1,307,578)—Interest income243—Net income (loss)(1,327,663)—Net (income) loss attributable to noncontrolling interests186,940—Net income (loss) per Class A common share\$ (14.67)\$	Revenues:				
Expenses:Property operating expenses $21,919$ —Real property taxes $9,460$ —Property management fees $1,769$ —Depreciation and amortization $49,288$ —Acquisition related expenses $131,454$ —Asset management and acquisition fees $570,375$ —Organizational expenses $63,164$ —General and administrative expenses $554,476$ —Total expenses $1,401,905$ —Income (loss) before other income (expenses) $(1,307,578)$ —Other income (expenses): $(1,327,663)$ —Interest income 243 —Net income (loss) attributable to noncontrolling interests $186,940$ —Net income (loss) attributable to common stockholders§ $(1,467)$ § —Basic and diluted income (loss) per Class A common share§ (14.67) § —	Rental revenue	\$ 94,327	\$		
Property operating expenses21,919—Real property taxes9,460—Property management fees1,769—Depreciation and amortization49,288—Acquisition related expenses131,454—Asset management and acquisition fees570,375—Organizational expenses63,164—General and administrative expenses554,476—Total expenses1,401,905—Income (loss) before other income (expenses)(1,307,578)—Other income (expenses):11,307,578—Interest expense243—Net income (loss)(1,327,663)—Net (income) loss attributable to noncontrolling interests186,940—Net income (loss) per Class A common share§ (11,40,723)§ —Basic and diluted income (loss) per Class A common share§ (14.67)§ —	Total revenues	94,327		—	
Real property taxes9,460-Property management fees1,769-Depreciation and amortization49,288-Acquisition related expenses131,454-Asset management and acquisition fees570,375-Organizational expenses63,164-General and administrative expenses554,476-Total expenses1,401,905-Income (loss) before other income (expenses)(1,307,578)-Other income (expenses):Interest expense243-Net income (loss)1186,940-Net (income) loss attributable to noncontrolling interests186,940-Net income (loss) per Class A common share\$(14.67)\$	Expenses:				
Property management fees $1,769$ $-$ Depreciation and amortization $49,288$ $-$ Acquisition related expenses $131,454$ $-$ Asset management and acquisition fees $570,375$ $-$ Organizational expenses $63,164$ $-$ General and administrative expenses $554,476$ $-$ Total expenses $1,401,905$ $-$ Income (loss) before other income (expenses) $(1,307,578)$ $-$ Other income (expenses): $ 243$ $-$ Interest expense $(20,328)$ $ -$ Net income (loss) $(1,327,663)$ $ -$ Net (income) loss attributable to noncontrolling interests $186,940$ $-$ Net income (loss) per Class A common share $$ (14.67)$ $$ -$	Property operating expenses	21,919		—	
Depreciation and amortization49,288—Acquisition related expenses131,454—Asset management and acquisition fees570,375—Organizational expenses63,164—General and administrative expenses554,476—Total expenses1,401,905—Income (loss) before other income (expenses)(1,307,578)—Other income (expenses):—Interest expense243—Interest income243—Net income (loss) attributable to noncontrolling interests186,940—Net income (loss) attributable to common stockholders\$ (1,140,723) \$ —Basic and diluted income (loss) per Class A common share\$ (14.67) \$ —	Real property taxes	9,460		—	
Acquisition related expenses $131,454$ $-$ Asset management and acquisition fees $570,375$ $-$ Organizational expenses $63,164$ $-$ General and administrative expenses $554,476$ $-$ Total expenses $1,401,905$ $-$ Income (loss) before other income (expenses) $(1,307,578)$ $-$ Other income (expenses): $(20,328)$ $-$ Interest expense 243 $-$ Net income (loss) $(1,327,663)$ $-$ Net (income) loss attributable to noncontrolling interests $186,940$ $-$ Net income (loss) per Class A common share $$(114.67)$ $$$ Basic and diluted income (loss) per Class A common share $$(14.67)$ $$$	Property management fees	1,769		—	
Asset management and acquisition fees $570,375$ $-$ Organizational expenses $63,164$ $-$ General and administrative expenses $554,476$ $-$ Total expenses $1,401,905$ $-$ Income (loss) before other income (expenses) $(1,307,578)$ $-$ Other income (expenses): $(20,328)$ $-$ Interest expense $(20,328)$ $-$ Interest income 243 $-$ Net income (loss) $(1,327,663)$ $-$ Net (income) loss attributable to noncontrolling interests $186,940$ $-$ Net income (loss) per Class A common share $$ (1140,723)$ $$ -$	Depreciation and amortization	49,288			
Organizational expenses63,164General and administrative expenses554,476Total expenses1,401,905Income (loss) before other income (expenses)(1,307,578)Other income (expenses):(1,307,578)Interest expense(20,328)Interest income243Net income (loss)(1,327,663)Net (income) loss attributable to noncontrolling interests186,940Net income (loss) attributable to common stockholders\$(1,140,723)\$Basic and diluted income (loss) per Class A common share\$(14.67)\$	Acquisition related expenses	131,454		—	
General and administrative expenses554,476—Total expenses1,401,905—Income (loss) before other income (expenses)(1,307,578)—Other income (expenses):Interest expense(20,328)—Interest income243—Net income (loss)(1,327,663)—Net (income) loss attributable to noncontrolling interests186,940—Net income (loss) attributable to common stockholders\$ (1,140,723)\$ —Basic and diluted income (loss) per Class A common share\$ (14.67)\$ —	Asset management and acquisition fees	570,375			
Total expenses1,401,905—Income (loss) before other income (expenses)(1,307,578)—Other income (expenses):—Interest expense(20,328)—Interest income243—Net income (loss)(1,327,663)—Net (income) loss attributable to noncontrolling interests186,940—Net income (loss) attributable to common stockholders\$ (1,140,723)\$ —Basic and diluted income (loss) per Class A common share\$ (14.67)\$ —	Organizational expenses	63,164		—	
Income (loss) before other income (expenses)(1,307,578)—Other income (expenses):(1,307,578)—Interest expense(20,328)—Interest income243—Net income (loss)(1,327,663)—Net (income) loss attributable to noncontrolling interests186,940—Net income (loss) attributable to common stockholders\$ (1,140,723)\$ —Basic and diluted income (loss) per Class A common share\$ (14.67)\$ —	General and administrative expenses	554,476			
Other income (expenses):Interest expense(20,328)—Interest income243—Net income (loss)(1,327,663)—Net (income) loss attributable to noncontrolling interests186,940—Net income (loss) attributable to common stockholders\$ (1,140,723)\$ —Basic and diluted income (loss) per Class A common share\$ (14.67)\$ —	Total expenses	 1,401,905			
Interest expense(20,328)—Interest income243—Net income (loss)(1,327,663)—Net (income) loss attributable to noncontrolling interests186,940—Net income (loss) attributable to common stockholders\$ (1,140,723)\$ —Basic and diluted income (loss) per Class A common share\$ (14.67)\$ —	Income (loss) before other income (expenses)	(1,307,578)			
Interest income243—Net income (loss)(1,327,663)—Net (income) loss attributable to noncontrolling interests186,940—Net income (loss) attributable to common stockholders\$ (1,140,723)\$ —Basic and diluted income (loss) per Class A common share\$ (14.67)\$ —	Other income (expenses):				
Net income (loss)(1,327,663)—Net (income) loss attributable to noncontrolling interests186,940—Net income (loss) attributable to common stockholders\$ (1,140,723)\$ —Basic and diluted income (loss) per Class A common share\$ (14.67)\$ —	Interest expense	(20,328)			
Net (income) loss attributable to noncontrolling interests186,940—Net income (loss) attributable to common stockholders\$ (1,140,723)\$ —Basic and diluted income (loss) per Class A common share\$ (14.67)\$ —	Interest income	243		_	
Net income (loss) attributable to common stockholders\$ (1,140,723)\$Basic and diluted income (loss) per Class A common share\$ (14.67)\$	Net income (loss)	(1,327,663)		_	
Basic and diluted income (loss) per Class A common share	Net (income) loss attributable to noncontrolling interests	186,940		_	
	Net income (loss) attributable to common stockholders	\$ (1,140,723)	\$		
Weighted average number of common shares outstanding77,7791,113	Basic and diluted income (loss) per Class A common share	\$ (14.67)	\$		
	Weighted average number of common shares outstanding	77,779		1,113	

HINES GLOBAL REIT II, INC. CONSOLIDATED STATEMENTS OF EQUITY For the Year Ended December 31, 2014 and for the Period from July 31, 2013 (date of inception) through December 31, 2013

Hines Global REIT II, Inc. Stockholders													
	Commo	n Sh	ares										
	Cla	ass A		Class A		— Additional Paid-		Accumulated		Total Stockholders'		Noncontrolling	
	Shares	A	mount		in Capital Deficit			Equity (Deficit)		Interests			
Balance at July 31, 2013 (date of inception)		\$		\$		\$	_	\$		\$	_		
Issuance of common shares	1,111		1		9,999		—		10,000		—		
Contributions from noncontrolling interest			_		_		_		_		190,000		
Balance as of December 31, 2013	1,111	\$	1	\$	9,999	\$		\$	10,000	\$	190,000		

	······································												
	Commo	n Sh	ares		Additional Paid- A in Capital								
	Cla	ss A		Ad					Accumulated	Total Stockholders'		Noncontrolling	
	Shares	Ar	nount	110					Equity (Deficit)		Interests		
Balance as of January 1, 2014	1,111	\$	1	\$	9,999	\$	_	\$	10,000	\$	190,000		
Issuance of common shares	419,430		420		3,863,201		—		3,863,621		_		
Distributions declared	—				(42,218)		—		(42,218)		(3,060)		
Selling commissions and dealer manager fees	_				(88,208)		_		(88,208)		_		
Issuer costs	—				(3,742,774)		—		(3,742,774)				
Net income (loss)	_						(1,140,723)		(1,140,723)		(186,940)		
Balance as of December 31, 2014	420,541	\$	421	\$		\$	(1,140,723)	\$	(1,140,302)	\$	_		

Hines Global REIT II, Inc. Stockholders

HINES GLOBAL REIT II, INC. CONSOLIDATED STATEMENTS OF CASH FLOWS For the Year Ended December 31, 2014 and for the Period from July 31, 2013 (date of inception) through December 31, 2013

	ear Ended mber 31, 2014	Period from July 31, 2013 (date of inception) through December 31, 2013		
CASH FLOWS FROM OPERATING ACTIVITIES:				
Net income (loss)	\$ (1,327,663)	\$		
Adjustments to reconcile net income (loss) to net cash from operating activities:				
Depreciation and amortization	49,288	—		
Changes in assets and liabilities:				
Change in other assets	(71,052)	—		
Change in tenant and other receivables	(5,960)	—		
Change in accounts payable and accrued expenses	106,685	—		
Change in other liabilities	(88,432)	—		
Change in due to affiliates	1,095,836	—		
Net cash from operating activities	(241,298)			
CASH FLOWS FROM INVESTING ACTIVITIES:				
Investments in acquired properties and lease intangibles	 (25,110,781)			
Net cash from investing activities	(25,110,781)			
CASH FLOWS FROM FINANCING ACTIVITIES:				
Proceeds from issuance of common stock	3,800,833	10,000		
Contribution from noncontrolling interest		190,000		
Payment of issuer costs	(17,250)	—		
Payment of selling commissions and dealer manager fees	(77,885)	—		
Distributions paid to stockholders and noncontrolling interests	(26,877)	—		
Proceeds from related party note payable	 24,200,000			
Net cash from financing activities	27,878,821	200,000		
Net change in cash and cash equivalents	2,526,742	200,000		
Cash and cash equivalents, beginning of period	 200,000			
Cash and cash equivalents, end of period	\$ 2,726,742	\$ 200,000		
Supplemental Schedule of Non-Cash Financing Activities				
Distributions declared and unpaid	\$ 18,371	\$		
Offering proceeds due from transfer agent	\$ 62,200	\$		
Non-cash net liabilities acquired	\$ 239,219	\$		
Offering costs payable to the Advisor	\$ 4,107,475	\$		

HINES GLOBAL REIT II, INC.

NOTES TO THE CONSOLIDATED FINANCIAL STATEMENTS

1. ORGANIZATION

Hines Global REIT II, Inc. (the "Company"), was formed as a Maryland corporation on July 31, 2013 for the purpose of engaging in the business of investing in and owning commercial real estate properties and other real estate investments. The business of the Company is managed by Hines Global REIT II Advisors LP (the "Advisor"), an affiliate of Hines Interests Limited Partnership ("Hines"), pursuant to the Advisory Agreement, dated as of August 15, 2014, between the Company, the Advisor and the Operating Partnership (defined below).

On July 31, 2013, the Company and Hines Global REIT II Associates Limited Partnership ("HALP II"), an affiliate of the Advisor, formed Hines Global REIT II Properties, LP (the "Operating Partnership"). On September 6, 2013, Hines Global REIT II Investor Limited Partnership, an affiliate of the Advisor, purchased 1,111.111 shares of common stock for \$10,000 and was admitted as the initial stockholder of the Company. Additionally, HALP II made initial capital contributions to the Operating Partnership of \$190,000 and, accordingly, owned a 95.0% limited partner interest in the Operating Partnership. As of December 31, 2014, HALP II owned a 4.8% limited partner interest in the Operating Partnership.

On August 20, 2014, the Company commenced an offering of up to \$2,500,000,000 of Class A common stock, at a price of \$10.00 per share for sale to the public (the "Offering"). The Company engaged Hines Securities, Inc. ("HSI"), an affiliate of the Advisor, to serve as the dealer manager for the Offering and market its shares. On September 26, 2014, the Company received an investment of \$2.0 million from Hines Global REIT II Investor Limited Partnership and achieved the minimum offering requirements pursuant to the terms of the Offering in all states except for Pennsylvania and Washington, which have higher minimum offering requirements. As of March 18, 2015, the Company had received gross offering proceeds of \$18.3 million from the sale of 1.9 million common shares.

The Company intends to invest the net proceeds from the Offering in a diversified portfolio of quality commercial real estate properties and other real estate investments throughout the United States and internationally. Properties purchased by the Company may have varying uses, including office, retail, industrial, multi-family residential and hospitality or leisure. The Company may invest in operating properties, properties under development, and undeveloped properties such as land. In addition, the Company may also make other real estate investments including equity or debt interests, which may include securities, in other real estate entities and debt related to real estate. The Company made its initial real estate investment in December 2014 and, as a result, owned one property, 2819 Loker Avenue East, as of December 31, 2014. The Class A industrial building, located in Carlsbad, California, consists of 161,310 square feet of leasable space. In March 2015, the Company acquired Bishop's Square, a Class A office building in Dublin, Ireland. See <u>Note 11</u> — Subsequent Events for additional information.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Use of Estimates

The Company's consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP"). The preparation of the consolidated financial statements requires the Company to make estimates and judgments that affect the reported amounts of assets, liabilities and contingencies as of the date of the financial statements and the reported amounts of revenues and expenses during the reporting periods. The Company evaluates its assumptions and estimates on an ongoing basis. The Company bases its estimates on historical experience and on various other assumptions that the Company believes to be reasonable under the circumstances. Additionally, application of the Company's accounting policies involves exercising judgments regarding assumptions as to future uncertainties. Actual results may differ from these estimates under different assumptions or conditions.

Basis of Presentation

The consolidated financial statements of the Company include the accounts of Hines Global REIT II, Inc. and the Operating Partnership (over which the Company exercises financial and operating control). All intercompany balances and transactions have been eliminated in consolidation.

Investment Property and Lease Intangibles

Real estate assets acquired by the Company are stated at fair value at the date of acquisition less accumulated depreciation. Depreciation is computed using the straight-line method. The estimated useful lives for computing depreciation are generally 10 years for furniture and fixtures, 15-20 years for electrical and mechanical installations and 40 years for buildings. Major replacements that extend the useful life of the assets are capitalized and maintenance and repair costs are expensed as incurred.

Acquisitions of properties are accounted for utilizing the acquisition method and, accordingly, are recorded at the estimated fair values of the assets acquired and liabilities assumed. The results of operations of acquired properties are included in the Company's results of operations from their respective dates of acquisition. Estimates of fair values are based upon estimates of future cash flows and other valuation techniques that the Company believes are similar to those used by market participants and are used to record the purchase of identifiable assets acquired, such as land, buildings and improvements, equipment and identifiable intangible assets related to in-place leases and liabilities assumed, such as amounts related to acquired out-of-market leases, asset retirement obligations, and mortgage notes payable. Values of buildings and improvements are determined on an as-if-vacant basis. Initial valuations are subject to change until such information is finalized, which will occur no later than 12 months after the acquisition date. Acquisition-related costs such as transaction costs and acquisition fees paid to the Advisor are expensed as incurred.

The estimated fair value of acquired in-place leases are the costs the Company would have incurred to lease the properties to the occupancy level of the properties at the date of acquisition. Such estimates include the fair value of leasing commissions, legal costs and other direct costs that would be incurred to lease the properties to such occupancy levels. Additionally, the Company evaluates the time period over which such occupancy levels would be achieved. Such evaluation will include an estimate of the net market-based rental revenues and net operating costs (primarily consisting of real estate taxes, insurance and utilities) that would be incurred during the lease-up period. Acquired in-place leases as of the date of acquisition are amortized over the remaining lease terms. Should a tenant terminate its lease, the unamortized portion of the in-place lease value is charged to amortization expense.

Acquired out-of-market lease values (including ground leases) are recorded based on the present value (using a discount rate that reflects the risks associated with the lease acquired) of the difference between the contractual amounts paid pursuant to the in-place leases and management's estimate of fair market value lease rates for the corresponding in-place leases. The capitalized out-of-market lease values are amortized as adjustments to rental revenue (or ground lease expense, as applicable) over the remaining terms of the respective leases, which include periods covered by bargain renewal options. Should a tenant terminate its lease, the unamortized portion of the out-of-market lease value is charged to rental revenue.

Management estimates the fair value of assumed mortgage notes payable based upon indications of then-current market pricing for similar types of debt with similar maturities. Assumed mortgage notes payable are initially recorded at their estimated fair value as of the assumption date, and the difference between such estimated fair value and the outstanding principal balance of the note will be amortized over the life of the mortgage note payable.

Impairment of Investment Property

Real estate assets are reviewed for impairment each reporting period if events or changes in circumstances indicate that the carrying amount of the individual property may not be recoverable. In such an event, a comparison will be made of the current and projected operating cash flows and expected proceeds from the eventual disposition of each property on an undiscounted basis to the carrying amount of such property. If the carrying amount exceeds the undiscounted cash flows, it would be written down to the estimated fair value to reflect impairment in the value of the asset. The determination of whether investment property is impaired requires a significant amount of judgment by management and is based on the best information available to management at the time of the evaluation. No impairment charges were recorded during the year ended December 31, 2014.

Cash and Cash Equivalents

The Company considers all short-term, highly liquid investments that are readily convertible to cash with a maturity of three months or less at the time of purchase to be cash equivalents.

Concentration of Credit Risk

As of December 31, 2014, the Company had cash and cash equivalents deposited in certain financial institutions in excess of federally insured levels. Management regularly monitors the financial stability of these financial institutions in an effort to manage the Company's exposure to any significant credit risk in cash and cash equivalents.

Tenant and Other Receivables

As of December 31, 2014, tenant and other receivables primarily consisted of a \$62,200 receivable from our transfer agent related to offering proceeds not yet received. Generally, we expect receivable balances to consist primarily of base rents, tenant reimbursements and receivables attributable to straight-line rent. An allowance for the uncollectible portion of tenant and other receivables is determined based upon an analysis of the tenant's payment history, the financial condition of the tenant, business conditions in the industry in which the tenant operates and economic conditions in the area in which the property is located. Tenant and other receivables are shown at cost in the consolidated balance sheets, net of allowance for doubtful accounts, if necessary. As of December 31, 2014, no such allowances have been recorded.

Deferred Leasing Costs

Direct leasing costs, primarily consisting of third-party leasing commissions, tenant inducements and legal costs are capitalized and amortized over the life of the related lease. Tenant inducement amortization is recorded as an offset to rental revenue and the amortization of other direct leasing costs is recorded in amortization expense.

Deferred Financing Costs

Deferred financing costs consist of direct costs incurred in obtaining debt financing. These costs are amortized into interest expense on a straight-line basis, which approximates the effective interest method, over the terms of the obligations. The amortization of these fees was insignificant for the year ended December 31, 2014.

Other Assets

Other assets included the following:

	Decem	ber 31, 2014
Deferred issuer costs	\$	364,701
Other ⁽¹⁾		81,574
Other assets	\$	446,275

(1) As of December 31, 2014, this primarily consisted of excess cash held in an escrow account related to timing differences during the completion of the acquisition of 2819 Loker Avenue East.

Revenue Recognition

The Company recognizes rental revenue on a straight-line basis over the life of the lease including rent holidays, if any. Straight-line rent receivable was \$5,895 as of December 31, 2014. Straight-line rent receivable consists of the difference between the tenants' rents calculated on a straight-line basis from the date of acquisition or lease commencement over the remaining terms of the related leases and the tenants' actual rents due under the lease agreements and is included in tenant and other receivables in the accompanying consolidated balance sheets. Revenues associated with operating expense recoveries are recognized in the period in which the expenses are incurred based upon the tenant lease provisions. Revenues relating to lease termination fees are recognized on a straight-line basis amortized from the time that a tenant's right to occupy the leased space is modified through the end of the revised lease term.

Issuer Costs

The Company reimburses the Advisor and its affiliates for any issuer costs related to the Offering that it pays on the Company's behalf. Such costs consist of, among other costs, expenses of the Company's organization, actual legal, accounting, bona fide out-of-pocket itemized and detailed due diligence costs, printing, filing fees, transfer agent costs, postage, escrow fees, data processing fees, advertising and sales literature and other offering-related costs. The Company did not have an obligation to reimburse the Advisor for any issuer costs until it achieved its minimum offering requirements on September 26,

2014. Therefore, the Company did not record issuer costs within its financial statements until that time. Organizational issuer costs, such as expenses associated with the formation of the Company and its board of directors are expensed as incurred, and offering-related issuer costs are recorded as an offset to additional paid-in capital. We have recorded a \$4.2 million liability to the Advisor for all of the issuer costs incurred by the Advisor, of which \$3.7 million has been recorded to stockholders' equity. Upon receipt of additional offering proceeds, the Company will record the remaining \$364,701 in deferred issuer costs to stockholders' equity.

Income Taxes

The Company intends to make an election to be taxed as a REIT, under Sections 856 through 860 of the Internal Revenue Code of 1986, as amended ("the Code"), and expects it will be taxed as such beginning with its taxable year ended December 31, 2015. In order to qualify as a REIT, an entity must meet certain organizational and operational requirements, including a requirement to distribute at least 90% of its annual ordinary taxable income to stockholders. REITs are generally not subject to federal income tax on taxable income that they distribute to their stockholders. If the Company fails to qualify as a REIT in any taxable year, it will be subject to federal income taxes on its taxable income at regular corporate rates and will not be permitted to qualify for treatment as a REIT for federal income tax purposes for four years following the year during which qualification is lost unless the Internal Revenue Service granted the Company relief under certain statutory provisions. Such an event could materially adversely affect the Company's net income and net cash available for distribution to stockholders. However, the Company believes that it will be organized and operate in such a manner as to qualify for treatment as a REIT for federal income taxed as a REIT for the year ended December 31, 2014. This did not have an impact on the Company's tax liability or the tax liability of the Company's stockholders that invested in the Company's public offering during 2014 since the Company did not have any taxable income for the year ended December 31, 2014.

Reportable Segments

Our current business consists of owning, operating, acquiring, developing, investing in, and disposing of real estate assets. All of our consolidated revenues and property operating expenses are from our one consolidated real estate property that we acquired on December 17, 2014.

Redemption of Common Stock

The Company offers a share redemption program which will allow certain stockholders to have their shares redeemed subject to approval and certain limitations and restrictions. No fees will be paid to Hines in connection with any redemption. The Company's board of directors may terminate, suspend or amend the share redemption program upon 30 days' written notice without stockholder approval.

The Company initially intends to allow redemptions of its shares on a monthly basis. Subject to funds being available as described below, the number of shares repurchased during any consecutive 12-month period will be limited to no more than 5% of the number of outstanding shares of common stock at the beginning of that 12-month period. Unless the Company's board of directors determines otherwise, the funds available for redemptions in each month will be limited to the funds received from the distribution reinvestment plan in the prior month.

The Company complies with FASB Accounting Standards Codification ("ASC") 480 "Distinguishing Liabilities from Equity" which requires, among other things, that financial instruments that represent a mandatory obligation of the Company to repurchase shares be classified as liabilities and reported at settlement value. When approved, the Company will reclassify such obligations from equity to an accrued liability based upon their respective settlement values. The Company had not redeemed any shares of its common stock as of December 31, 2014.

Reclassification of Class T Shares

Previously, the Company intended to offer a combination of Class A shares and Class T shares of common stock for sale to the public. On December 12, 2014, the Company supplemented its Articles of Amendment and Restatement to reclassify 300,000,000 authorized but unissued shares of Class T common stock as Class A common stock. As a result, the Company reclassified 300,000,000 Class T shares authorized on the December 31, 2013 balance sheet into shares of Class A common stock.

Per Share Data

Net income (loss) per common share is calculated by dividing the net income (loss) attributable to common stockholders for each period by the weighted average number of common shares outstanding during such period. Net income (loss) per common share on a basic and diluted basis is the same because the Company has no potentially dilutive common shares outstanding. Earnings per share have been restated for 737 shares in stock dividends through the date of issuance of the financial statements.

Recent Accounting Pronouncements

In August 2014, FASB issued amendments to the Codification to provide guidance on management's responsibility to evaluate whether there are conditions or events that raise substantial doubt about the entity's ability to continue as a going concern within one year after the date that the financial statements are issued or are available to be issued. These amendments are effective for fiscal years, and interim periods within those years, beginning after December 15, 2016 and early adoption is permitted. The Company does not believe the adoption of this guidance will have a material impact on its financial statements.

In January 2015, FASB issued amendments to the Codification to provide guidance on the disclosure of extraordinary items. The amendments eliminate the concept of extraordinary items from GAAP. These amendments are effective for fiscal years, and interim periods within those years, beginning after December 15, 2015 and early adoption is permitted. The Company does not believe the adoption of this guidance will have a material impact on its financial statements.

In February 2015, the FASB issued amendments to the Codification's consolidation guidance which improves targeted areas of the consolidation guidance and reduces the number of consolidation models. The amendments in the ASU are effective for annual and interim periods in fiscal years beginning after December 15, 2015, with early adoption permitted. The Company has not elected to early adopt this standard. The Company does not believe the adoption of this guidance will have a material impact on its financial statements.

3. INVESTMENT PROPERTY

Investment property consisted of the following amounts as of December 31, 2014:

	December 31, 2014
Buildings and improvements	\$ 14,170,000
Less: accumulated depreciation	(14,579)
Buildings and improvements, net	14,155,421
Land	7,200,000
Investment property, net	\$ 21,355,421

As of December 31, 2014, the cost basis and accumulated amortization related to lease intangibles was as follows:

	In-Place Lease
Cost	\$ 3,980,000
Less: accumulated amortization	(34,709)
Net	\$ 3,945,291

Amortization expense of in-place leases was \$34,709 for the year ended December 31, 2014.

Anticipated amortization of the Company's in-place lease for each of the years ending December 31, 2015 through December 31, 2019 are as follows:

	In-Place Lease
2015	\$ 860,851
2016	860,851
2017	860,851
2018	860,851
2019	501,887

Leases

The Company has entered into a non-cancelable lease agreement with a tenant for space. As of December 31, 2014, the approximate fixed future minimum rentals for each of the years ending December 31, 2015 through 2019 and thereafter were as follows:

	Fixed Future Minimum Rentals
2015	\$ 1,845,225
2016	1,900,582
2017	1,957,599
2018	2,016,327
2019	1,207,255
Thereafter	_
Total	\$ 8,926,988

Of the Company's total rental revenue for the year ended December 31, 2014, 100% was earned from Acushnet, a tenant in the manufacturing industry, whose lease expires in 2019.

4. RECENT ACQUISITIONS OF REAL ESTATE

The Company made its initial real estate investment in December 2014 through its acquisition of 2819 Loker Avenue East, a Class A industrial property located in Carlsbad, California. The amounts recognized for major assets acquired as of the acquisition date were determined by allocating the \$25.4 million purchase price of 2819 Loker Avenue East as follows:

Property Name	Acquisition Date	Building and Improvements	Land	In-place Lease Intangibles	Total
	Dutt	impi o vemento	Bunu	Intelligibility	1000
2819 Loker Avenue East	12/17/2014	\$ 14,170,000	\$ 7,200,000	\$ 3,980,000	\$ 25,350,000

Included in the Company's rental revenue and net loss for the year ended December 31, 2014 was \$94,327 in rental revenue and \$79,660 in net loss related to 2819 Loker Avenue East. Additionally, at the date of acquisition, the weighted average useful life of 2819 Loker Avenue East's intangible assets was 4.6 years for in-place lease intangibles.

The following unaudited consolidated information is presented to give effect to the current year acquisition through December 31, 2014 as if the acquisition occurred on July 31, 2013 (date of inception). The pro forma net loss was adjusted to exclude acquisition-related expenses of \$698,829 for the year ended December 31, 2014. For the period from inception through December 31, 2013, the pro forma net loss was adjusted to include acquisition fees and expenses of 698,829 relating to the 2014 acquisition, as if these fees and expenses had been incurred as of July 31, 2013 (date of inception).

The information below is not necessarily indicative of what the actual results of operations would have been had we completed these transactions on July 31, 2013, nor does it purport to represent our future operations:

	Pro Forma for the Year Ended December 31, 2014		Pro Forma for the Period from July 31, 2013 (date of inception) through December 31, 2013		
Revenues	\$ 2,295,290	\$	955,847		
Net income (loss)	\$ (566,958)	\$	(795,192)		
Basic and diluted income (loss) per common share	\$ (4.39)	\$	(6.16)		

See <u>Note 11</u> — Subsequent Events for details regarding our real estate investment that was acquired subsequent to December 31, 2014.

5. DEBT FINANCING

The following table includes all of the Company's debt outstanding as of December 31, 2014:

Description	Origination or Assumption Date	Maturity Date	Interest Rate Description	Interest Rate as of December 31, 2014	Principal Outstanding at December 31, 2014
Affiliates Notes Payable					
Credit Facility with Hines	12/15/2014	12/15/2016 (1)	Variable (2)	2.16%	\$ 24,200,000
Notes Payable to Affiliates					\$ 24,200,000

- (1) Each advance under the credit facility with Hines (the "Hines Credit Facility") must be repaid within six months, subject to one six-month extension and subject to the satisfaction of certain conditions. Notwithstanding that each advance under the Hines Credit Facility matures six months after it is made, Hines Global II is required to repay each advance under the Hines Credit Facility with proceeds from its public offering as such proceeds are raised, unless Hines Global II, through the Operating Partnership, enters into a revolving credit facility (the "OP Facility"), at which point Hines Global II may use such proceeds from its public offering to repay the OP Facility, if any, prior to repaying any advances under its credit facility with Hines. The Hines Credit Facility also permits voluntary prepayment of principal and accrued interest.
- (2) Interest on each advance is charged monthly at a variable rate, which is the greater of (i) Hines' then-current borrowing rate under its revolving credit facility and (ii) if Hines Global II enters into a revolving credit facility through the Operating Partnership, the rate under such facility.

Financial Covenants

The Hines Credit Facility contains customary events of default, with corresponding grace periods, including payment defaults, bankruptcy-related defaults, and customary covenants, including limitations on liens and indebtedness and maintenance of certain financial ratios. The Company is not aware of any instances of noncompliance with financial covenants related to the Hines Credit Facility as of December 31, 2014.

Principal Payments on Debt

The Company is required to make the following principal payments on its outstanding notes payable for each of the years ending December 31, 2015 through December 31, 2019 and for the period thereafter.

	Payments due by Year						
	2015	2016	2017	2018	2019	Thereafter	
Principal payments	\$ 24,200,000	\$ —	\$ —	\$ —	\$ —	\$ —	

6. DISTRIBUTIONS

With the authorization of its board of directors, the Company declared distributions to its stockholders and HALP II for the period from October 1, 2014 through March 31, 2015. These distributions were calculated based on stockholders of record for each day in an amount equal to \$0.001575342 per share, per day. All distributions were or will be paid in cash or reinvested in shares of the Company's common stock for those participating in the Company's distribution reinvestment plan and will be paid or issued, respectively, on the first business day following the completion of the month to which they relate. Some or all of the cash distributions may be paid from sources other than cash flows from operations.

In addition to the distributions described above, the Company's board of directors authorized special stock dividends to its stockholders as of daily record dates for the period from October 1, 2014 through March 31, 2015. Stock dividends for the Company's common shares were calculated based on stockholders of record each day in an amount equal to 0.0000273973 of a share of common stock per share, per day. Shares issued related to the Company's stock dividends were or will be issued on the first business day of the month following the quarter to which they relate.

Cash Distributions

The following table outlines the Company's total cash distributions declared to stockholders and noncontrolling interests (HALP II) for the year ended December 31, 2014, including the breakout between the distributions declared in cash and those reinvested pursuant to the Company's distribution reinvestment plan.

	Stockholders				Noncontrolling Interests		
Distributions for the Year Ended		Cash ributions		ibutions vested	Total eclared	Total I	Declared
December 31, 2014	\$	40,377	\$	1,841	\$ 42,218	\$	3,060

Stock Dividends

The following table outlines the Company's total stock dividends declared to stockholders for the year ended December 31, 2014.

		Stocl	kholdei	rs
Dividends for the Year Ended	Issuance Date	Shares	Amor	unt ⁽¹⁾
December 31, 2014	1/1/2015	737	\$	7,367

(1) Amount based on \$10.00 per share Offering price.

7. RELATED PARTY TRANSACTIONS

The table below outlines fees and expense reimbursements incurred that are payable by the Company to Hines and its affiliates for the year ended December 31, 2014 and amounts unpaid as of December 31, 2014:

Type and Recipient	urred During the Year Ended cember 31, 2014	Unpaid as of December 31, 2014		
Selling Commissions- Dealer Manager	\$ 60,697	\$	—	
Dealer Manager Fee- Dealer Manager	\$ 27,511		10,323	
Issuer and Organization Costs- the Advisor	\$ 4,153,388		4,153,388	
Acquisition Fee- the Advisor	\$ 570,375		570,375	
Asset Management Fee- the Advisor ⁽¹⁾	\$ 			
Other ⁽²⁾ - the Advisor	\$ 365,960		365,960	
Interest Expense- Hines ⁽³⁾	\$ 20,328		20,328	
Property Management Fee- Hines	\$ 1,769			
Expense Reimbursement- Hines (with respect to management and operations of the Company's properties)	\$ 1,260		93,260	
Due to Affiliates		\$	5,213,634	

(1) The Advisor waived \$16,258 in asset management fees payable to it during the year ended December 31, 2014. See "— Advisory Agreement" below for a discussion of the asset management fee waiver.

- (2) Includes amounts the Advisor paid on behalf of the Company such as general and administrative expenses and acquisitionrelated expenses. These amounts are generally reimbursed to the Advisor during the month following the period in which they are incurred.
- (3) Includes amounts paid related to the Hines Credit Facility.

Dealer Manager Agreement

The Dealer Manager receives a selling commission of up to 7.0% of gross offering proceeds and a dealer manager fee of up to 3.0% of gross offering proceeds, both of which are recorded as an offset to additional paid-in-capital in the Company's financial statements. Pursuant to separately negotiated agreements, the Dealer Manager may reallow all of its commission and a portion of its dealer manager fee to broker-dealers participating in the Offering. The Dealer Manager, in its sole discretion, may also pay to participating broker dealers out of its dealer manager fee a marketing fee and may reimburse participating broker dealers for distribution and marketing-related costs and expenses, such as costs associated with attending or sponsoring conferences and technology costs. No selling commissions or dealer manager fees will be paid for sales under the Company's distribution reinvestment plan.

Advisory Agreement

Pursuant to the Advisory Agreement and the Operating Partnership Agreement, the Company is required to pay the following fees and expense reimbursements:

Acquisition Fee – The Advisor receives acquisition fees of 2.25% of (i) the purchase price of real estate investments acquired, including any debt attributable to such investments, or the total principal amounts borrowed under any loans made or acquired directly by the Company, or (ii) when the Company makes an investment or makes or acquires a loan indirectly through another entity, such investment's pro rata share of the gross asset value of real estate investments held by that entity, including any debt attributable to such investments, or the total principal amount borrowed under any loans made or acquired by that entity.

Asset Management Fee – The Advisor also receives asset management fees of 0.0625% per month of the cost of the Company's real estate investments at the end of each month; provided that, if the Company's board of directors has determined an estimated net asset value per share, then, with respect to real estate investments included in the board of director's determination, the asset management fees will be equal to 0.0625% per month of the most recently determined value of such real estate investments at the end of each month.

In September 2014, the Advisor agreed to waive the asset management fees otherwise payable to it for the quarter ended December 31, 2014, to the extent that the Company's modified funds from operations ("MFFO") for the quarter ended December 31, 2014, as disclosed in this Annual Report on Form 10-K for such year, amounts to less than 100% of the aggregate distributions declared to the Company's stockholders for the quarter ended December 31, 2014. As a result of this waiver, the Advisor waived \$16,258 of asset management fees payable to it for the year ended December 31, 2014.

Furthermore, in December 2014, the Advisor also agreed to waive the asset management fees otherwise payable to it for the quarters ended March 31, 2015 and June 30, 2015, to the extent that the Company's MFFO for such quarters, as disclosed in the Company's Quarterly Report on Form 10-Q, amounts to less than 100% of the aggregate distributions declared to the Company's stockholders for each respective quarter. As a result of these fee waivers, cash flows from operations that would have been paid to the Advisor for asset management fees may be available to pay distributions to stockholders. These fee waivers are not deferrals and accordingly, any fees that are waived will not be paid to the Advisor in cash at any time in the future.

Disposition Fee – If the Advisor, its affiliates or related parties provide a substantial amount of services, as determined in good faith by a majority of the Company's independent directors, the Company will pay the Advisor, its affiliates or related parties a disposition fee in an amount equal to (a) 1.0% of the market value determined in connection with a listing of the Company's common stock on a national securities exchange, or 1.0% of the gross consideration received or to be received by the Company or its stockholders upon the occurrence of any other liquidity event involving the Company or the Operating Partnership, pursuant to which the Company's stockholders receive in exchange for their shares of the Company's common stock, cash, listed securities, securities redeemable for cash, or a combination thereof, or (b) 1.0% of the gross sales price upon the sale or transfer of one or more real estate investments (including a sale of all of the Company's real estate investments).

Even if the Advisor, its affiliates or related parties receive a disposition fee, the Company may still be obligated to pay fees or commissions to another third party. However, when a real estate or brokerage fee is payable in connection with a particular transaction, the amount of the disposition fee paid to the Advisor or its affiliates or related parties, as applicable, may not exceed an amount equal to the lesser of (i) one-half of a competitive real estate or brokerage commissions paid to unaffiliated parties, may not exceed the lesser of (x) a competitive real estate or brokerage commission or (y) an amount equal to 6.0% of the gross sales price.

Special OP Units – An affiliate of the Advisor holds special partnership interests in the Operating Partnership ("Special OP Units"), which will entitle them to receive distributions in an amount equal to 15.0% of distributions, including from sales of real estate investments, refinancings and other sources, but only after the Company's stockholders have received, or are deemed to have received, in the aggregate, cumulative distributions equal to their invested capital plus a 6.0% cumulative, non-compounded annual return on such invested capital.

Additionally, at the sole discretion of the Advisor, the acquisition fees, asset management fees or disposition fees are payable, in whole or in part, in cash or units of the Operating Partnership ("OP Units"). In the case of the disposition fee, the Advisor may also elect to be paid, if applicable, in securities issued by another entity. For the purposes of the payment of these fees, each OP Unit will be valued at the per share offering price of the Company's common stock in the Company's most recent public offering less selling commissions and dealer manager fees payable with respect to such common stock, to account for the fact that no selling commissions or dealer manager fees will be paid in connection with any such issuances. Each OP unit will be convertible into one share of the Company's common stock. The Company will recognize the expense related to these OP Units as the related service is performed, as each OP Unit will be fully vested upon issuance.

The Company will reimburse the Advisor for all expenses paid or incurred by the Advisor in connection with the services provided to the Company, subject to the limitation that the Company will not reimburse the Advisor for any amount by which its operating expenses (including the asset management fee) at the end of the four preceding fiscal quarters exceeds the greater of: (A) 2.0% of its average invested assets, or (B) 25.0% of its net income determined without reduction for any additions to reserves for depreciation, bad debts or other similar non-cash reserves and excluding any gain from the sale of the Company's assets for that period. Notwithstanding the above, the Company may reimburse the Advisor for expenses in excess of this limitation if a majority of the independent directors determines that such excess expenses are justified based on unusual and non-recurring factors.

In addition, as described in <u>Note 2</u> — Summary of Significant Accounting Policies – Issuer Costs, the Company reimburses the Advisor for any issuer costs paid on its behalf. However, the total reimbursement related to issuer costs, selling commissions and dealer manager fees may not exceed 15.0% of gross proceeds from the Offering.

Property Management and Leasing Agreements

The Company pays Hines fees for the management and leasing of the Company's properties. Property management fees will be paid in an amount equal to a market-based percentage of the gross revenues of the properties managed by Hines. In addition, if Hines provides leasing services with respect to a property, the Company will pay Hines leasing fees in an amount equal to the leasing fees charged by unaffiliated persons rendering comparable services in the same geographic area of the applicable property. The Company generally will be required to reimburse Hines for certain operating costs incurred in providing property management and leasing services pursuant to the property management and leasing agreements. Included in this reimbursement of operating costs will be the cost of personnel and overhead expenses related to such personnel located at the property as well as off-site personnel located in Hines' headquarters and regional offices, to the extent the same relate to or support the performance of Hines' duties under the agreement.

Hines may perform construction management services for the Company for both re-development activities and tenant construction. These fees are considered incremental to the construction effort and will be capitalized to the associated real estate project as incurred. Costs related to tenant construction will be depreciated over the estimated useful life. Costs related to redevelopment activities will be depreciated over the estimated useful life of the associated project. Leasing activities will generally be performed by Hines on the Company's behalf. Leasing fees will be capitalized and amortized over the life of the related lease.

Transactions with Affiliates

In December 2014, the Company entered into the Hines Credit Facility, which has a maximum commitment amount of \$75.0 million to provide financing for acquisitions and other working capital needs. See <u>Note 5</u> — Debt Financing for additional information regarding this loan agreement.

In addition to its initial capital contribution in September 2013, in September 2014, Hines Global REIT II Investor Limited Partnership, an entity wholly-owned by Jeffrey C. Hines, invested \$2.0 million to purchase shares the Company's common stock through the Offering (no selling commissions or dealer manager fees were incurred). As a result of this investment, the Company satisfied the minimum offering requirements in all states other than Pennsylvania and Washington.

8. FAIR VALUE MEASUREMENTS

Fair values determined by Level 1 inputs utilize quoted prices (unadjusted) in active markets for identical assets or liabilities the Company has the ability to access. Fair values determined by Level 2 inputs utilize inputs other than quoted prices included in Level 1 that are observable for the asset or liability, either directly or indirectly. Level 2 inputs include quoted prices for similar assets and liabilities in active markets and inputs other than quoted prices observable for the asset or liability, such as interest rates and yield curves observable at commonly quoted intervals. Level 3 inputs are unobservable inputs for the asset or liability, and include situations where there is little, if any, market activity for the asset or liability. In instances in which the inputs used to measure fair value may fall into different levels of the fair value hierarchy, the level in the fair value hierarchy within which the fair value measurement in its entirety. The Company's assessment of the significance of a particular input to the fair value measurement in its entirety requires judgment, and considers factors specific to the asset or liability.

As of December 31, 2014, the Company estimated that the book value of its note payable approximates its fair value. Other financial instruments not measured at fair value on a recurring basis include cash and cash equivalents, tenant and other receivables, accounts payable and accrued expenses, due to affiliates and distributions payable. The carrying value of these items reasonably approximates their fair value based on their highly-liquid nature and/or short-term maturities. Due to the short-term nature of these instruments, Level 1 and Level 2 inputs are utilized to estimate the fair value of these financial instruments.

9. COMMITMENTS AND CONTINGENCIES

The Company may be subject to various legal proceedings and claims that arise in the ordinary course of business. These matters are generally covered by insurance. While the resolution of these matters cannot be predicted with certainty,

management believes the final outcome of such matters will not have a material adverse effect on the Company's consolidated financial statements.

10. QUARTERLY FINANCIAL DATA (UNAUDITED)

The following table presents selected unaudited quarterly financial data for each quarter during the year ended December 31, 2014.

	Quarters Ended									
	March 31, 2014		Jun	e 30, 2014	September 30, 2014			December 31, 2014		
Revenues	\$		\$	_	\$	_	\$	94,327		
Net income (loss)	\$	—	\$		\$	(283,389)	\$	(1,044,274)		
Net (income) loss attributable to noncontrolling interests	\$	_	\$	_	\$	190,000	\$	(3,060)		
Net income (loss) attributable to common stockholders	\$		\$	_	\$	(93,389)	\$	(1,047,334)		
Income (loss) per Class A common share, basic and diluted	\$	_	\$	_	\$	(7.07)	\$	(3.57)		

11. SUBSEQUENT EVENTS

Bishop's Square

In March 2015, the Company acquired Bishop's Square, a Class A office building, located in the central business district of Dublin, Ireland. The contract purchase price was €92.0 million (approximately \$103.5 million assuming a rate of \$1.12 per EUR as of the transaction date), exclusive of transaction costs and working capital reserves. The building contains approximately 153,529 square feet of net rentable area and is 100% leased to five tenants. As a result of the acquisition of Bishop's Square, the Company incurred approximately \$2.3 million in acquisition fees payable to the Advisor.

The Company has not concluded its accounting for this recent acquisition, but it expects that the purchase price will primarily be allocated to building, land and intangible assets and liabilities.

In connection with the acquisition of Bishop's Square, a wholly-owned subsidiary of the Company entered into a secured facility agreement (the "Bishop's Square Facility") with DekaBank Deutsche Girozentrale totaling \in 55.2 million (approximately \$62.1 million assuming a rate of \$1.12 per EUR as of the transaction date). The Bishop's Square Facility requires quarterly interest payments beginning in July 2015 and repayment of principal upon the maturity of the Bishop's Square Facility on March 2, 2022. The Bishop's Square Facility has a floating interest rate of EURIBOR plus 1.30%. At the date of the initial draw, the interest rate was 1.38%. The Bishop's Square Facility may be repaid in full prior to maturity, subject to a prepayment premium if it is repaid in the first three years. In accordance with the terms of the facility agreement, the Company entered into a \in 55.2 million (approximately \$62.1 million assuming a rate of \$1.12 per EUR as of the transaction date) three-year interest rate cap at 2% to limit exposure to interest rate fluctuations.

Item 9. Changes in and Disagreements With Accountants on Accounting and Financial Disclosure

None.

Item 9A. Controls and Procedures

Disclosure Controls and Procedures

In accordance with Exchange Act Rules 13a-15 and 15d-15, we carried out an evaluation, under the supervision and with the participation of management, including our Chief Executive Officer and Chief Financial Officer, of the effectiveness of our disclosure controls and procedures as of the end of the period covered by this report. Based on that evaluation, our Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures were effective as of December 31, 2014, to provide reasonable assurance that information required to be disclosed in our reports filed or submitted under the Exchange Act is (i) recorded, processed, summarized and reported within the time periods specified in the Securities and Exchange Commission's rules and forms, and (ii) accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, as appropriate to allow timely decisions regarding required disclosure.

Management's Report on Internal Control Over Financial Reporting

This Annual Report on Form 10-K does not include a report of management's assessment regarding internal control over financial reporting or an attestation report of the Company's registered public accounting firm due to a transition period established by SEC rules for newly public companies.

Change in Internal Controls

No changes have occurred in our internal controls over financial reporting (as defined in Rule 13a-15(f) of the Exchange Act) during the quarter ended December 31, 2014 that has materially affected, or is reasonably likely to materially affect, our internal controls over financial reporting.

Item 9B. Other Information

None.

PART III

Item 10. Directors, Executive Officers and Corporate Governance

As of the date of this report, our directors and executive officers, their ages and their positions and offices are as follows:

<u>Name</u>	Age	Position and Office with Hines Global II
Jeffrey C. Hines	59	Director and Chairman of the board of directors
Charles M. Baughn	60	Director
Humberto "Burt" Cabañas	67	Independent Director
Dougal A. Cameron	57	Independent Director
John O. Niemann, Jr.	58	Independent Director
Sherri W. Schugart	49	President and Chief Executive Officer
Ryan T. Sims	43	Chief Financial Officer and Secretary
David L. Steinbach	38	Chief Investment Officer
Kevin L. McMeans	50	Asset Management Officer
J. Shea Morgenroth	39	Chief Accounting Officer and Treasurer

Jeffrey C. Hines. Mr. Hines joined Hines in 1982. Mr. Hines serves as our Chairman of the board of directors and Chairman of the managers of the general partner of our Advisor. Mr. Hines also serves as the Chairman of the board of directors of Hines Global REIT, Inc. ("Hines Global I") and Chairman of the managers of the general partner of Hines Global REIT Advisors LP, the advisor to Hines Global I. Mr. Hines has been the Chairman of the board of directors of Hines Real Estate Investment Trust, Inc. ("Hines REIT"), Chairman of the managers of the general partner of HALP Associates Limited Partnership, the advisor to Hines REIT and a member of the management board of the Hines US Core Office Fund LP ("Core Fund"), since August 2003. He is also the co-owner and President and Chief Executive Officer of the general partner of Hines in January 2008 and has overseen a major expansion of the firm's personnel, financial resources, domestic and foreign market penetration, products and services. He directed development of the firm's first \$846 million Emerging Markets Fund that provided start-up capital for projects in emerging international markets. He has been a major participant in the development of the Hines domestic and international acquisition program and currently oversees a portfolio with controlled assets valued at approximately \$30.8 billion as of June 30, 2014. Mr. Hines graduated from Williams College with a B.A. in Economics and received his M.B.A. from the Harvard Business School. Mr. Hines is the son of Gerald D. Hines.

We believe that Mr. Hines' career, spanning more than 30 years in the commercial real estate industry, including his service as Chairman of the board of directors of Hines REIT and Hines Global I, his leadership of Hines, his participation in Hines' international acquisition program and the depth of his knowledge of Hines and its affiliates, provide him with the business expertise and leadership experience necessary to serve as Chairman of our board of directors.

Charles M. Baughn. Mr. Baughn joined Hines in 1984. Mr. Baughn serves as a member of our board of directors and as a manager of the general partner of our Advisor. Mr. Baughn also serves as a member of the board of directors of Hines Global I and as a member of the general partner of its Advisor. Mr. Baughn has also been a member of the board of directors of Hines REIT since April 2008 and a manager of the general partner of its Advisor since August 2003. Mr. Baughn also served as Chief Executive Officer of Hines REIT from August 2003 through April 1, 2008. He has served as the Chief Financial Officer of the general partner of Hines since 2012. In this role, he is responsible for overseeing Hines' operating business including central services, balance sheet-related activities and bank and other debt financing. Previously, he has also served as an Executive Vice President and CEO—Capital Markets Group of the general partner of Hines' Executive Committee and the Chief Executive Officer and a director of our Dealer Manager. Mr. Baughn has also been a member of the management board of the Core Fund since 2003. During his tenure at Hines, he also has contributed to the development or redevelopment of over 9 million square feet of office and special use facilities in the southwestern United States. He graduated from the New York State College of Ceramics at Alfred University with a B.A. and received his M.B.A. from the University of Colorado. Mr. Baughn holds Series 7, 24 and 63 securities licenses.

We believe that the depth and breadth of Mr. Baughn's experience in the commercial real estate industry acquired during his 30 year career with Hines, including his familiarity with Hines' financial and investment policies and his experience overseeing the raising, placement and management of equity and debt for Hines' domestic and international projects, well qualifies him to serve on our board of directors.

Humberto "Burt" Cabañas. Mr. Cabañas has been one of our independent directors since July 2014. Mr. Cabañas is the Founder and Chairman of Benchmark Hospitality International, where he oversees Benchmark's diverse portfolio, a position he has held since 1987. He also held the position of Chief Executive Officer at Benchmark from 1987 to 2013. Prior to his current position, he served as a Senior Vice President and in various other positions at The Woodlands Corporation, where he was eventually placed in charge of all hospitality operations. He presently serves as director on the Foundation Board of Florida International University and previously served as Chairman of the Industry Advisory Board for the Chaplin School of Hospitality & Tourism Management at Florida International University. Mr. Cabañas is also a founding board member and a past president of the International Association of Conference Centers (IACC). He is a graduate of Florida International University with a Bachelor's degree in Hotel and Restaurant Management.

We believe Mr. Cabañas' significant and deep experience in the hotel and hospitality industry and the real estate aspects of that industry makes him well-qualified to serve as one of our directors.

Dougal A. Cameron. Mr. Cameron has been one of our independent directors since July 2014. Mr. Cameron is President and Owner of Cameron Management, a position he has held since his founding of the firm in 1995. Cameron Management is a firm that owns, finances, leases, and manages all of its office buildings in an integrated fashion. From 1991-1994, Mr. Cameron was an owner and investment manager of VNSM Inc., an investment management firm with over \$1 billion in assets under management. Prior to this time, he was an accountant with Arthur Young & Company (now Ernst & Young), and he worked as an asset and project manager for Hines Interests Limited Partnership from 1985 to 1991. Additionally, he was on the board of Mosher Inc., a private, closed-end mutual fund from 1992 to 1997. Mr. Cameron holds a bachelor's degree in Accounting from Texas Tech University and an MBA from Harvard Business School.

We believe that Mr. Cameron is well-qualified to serve as one of our directors due to his background in running a real estate firm since 1995 and his extensive experience in owning, leasing and managing office buildings as the company intends to make investments in office buildings along with other real estate product types.

John O. Niemann, Jr. Mr. Niemann has been one of our independent directors since July 2014. In addition, he has served as an independent director of HMS Income Fund, Inc. ("HMS"), which is a non-listed public business development company that is co-sponsored by Hines, since May 2012 and currently serves as the Chairman of the Audit Committees. He previously served as a director and Chairman of the Audit Committee of Gateway Energy Corporation from June 2010 until December 2013 (when the company went private). Since June 2013, he has served as a Managing Director of Andersen Tax LLC (formerly known as WTAS LLC). He is also the president and chief operating officer of Arthur Andersen LLP, and has been since 2003. He previously served on the administrative board of Arthur Andersen LLP and on the board of partners of Andersen Worldwide. He began his career at Arthur Andersen LLP in 1978 and has served in increasing responsibilities in senior management positions, since 1992. Mr. Niemann has served on the board of directors of many Houston area non-profit organizations, including Strake Jesuit College Preparatory School (past chair of the board), The Regis School of the Sacred Heart (past chair of the board), The Houston Symphony, The University of St. Thomas, The Alley Theatre and Taping for the Blind, Inc. He graduated with a bachelor of arts in managerial studies (magna cum laude) and a masters in accounting from Rice University and received a juris doctor (summa cum laude) from the South Texas College of Law.

We believe Mr. Niemann's significant experience in the public accounting industry, including 37 years in various capacities at Arthur Andersen LLP makes him well qualified to serve as one of our directors and as the audit committee financial expert. Drawing on this experience, Mr. Niemann is able to provide valuable insights regarding our investment strategies, internal controls, and financial reporting. In addition, through his experience serving on the board of directors of other public companies, Mr. Niemann has previous experience in the requirements of serving on a public company board.

Sherri W. Schugart. Ms. Schugart joined Hines in 1995. Ms. Schugart serves as President and Chief Executive Officer for us and for the general partner of our Advisor. Ms. Schugart has also served as President and Chief Executive Officer for Hines Global I and for the general partner of its Advisor since March 2013. Additionally, since March 2013, Ms. Schugart has also served as President and Chief Executive Officer for Hines REIT, the general partner of its Advisor and the Core Fund. In these roles, Ms. Schugart is responsible for the overall management of each funds' business strategy and operations in the U.S. and internationally. Also since March 2013, Ms. Schugart has served as the President and Chief Executive Officer of HMS Income Fund, Inc. ("HMS") and HMS Adviser GP LLC, the general partner of the adviser to HMS. Additionally, in February 2014, Ms. Schugart was appointed as the Chairperson of the board of directors of HMS. HMS is a public specialty finance company

sponsored by Hines, which was formed in 2011 and intends to make debt and equity investments in companies with revenues generally between \$10 million and \$3 billion that operate in diverse industries. Prior to March 2013, Ms. Schugart had served as the Chief Operating Officer for Hines Global I and the general partner of its Advisor and as the Chief Operating Officer of Hines REIT, the general partner of its Advisor and the Core Fund since November 2011. In these roles, Ms. Schugart was responsible for the execution of each entity's business plan and oversight of day-to-day business operations, including issues related to portfolio strategy, asset management and all other operational and financial matters of each entity. Ms. Schugart also served as Chief Financial Officer for Hines Global I and the general partner of its Advisor from inception in December 2008 through October 2011. Ms. Schugart also served as the Chief Financial Officer for Hines REIT and the general partner of its Advisor from August 2003 through October 2011 and as the Chief Financial Officer of the Core Fund from July 2004 through October 2011. In these roles, her responsibilities included oversight of financial and portfolio management, equity and debt financing activities, investor relations, accounting, financial reporting, compliance and administrative functions in the U.S. and internationally. She has also been a Senior Managing Director of the general partner of Hines since October 2007 and has served as a director of the Dealer Manager since August 2003. Prior to holding these positions, she was a Vice President in Hines Capital Markets Group raising equity and debt financing for various Hines investment vehicles in the U.S. and internationally. Ms. Schugart has been responsible for arranging and managing more than \$10 billion in equity and debt for Hines' public and private investment funds. Prior to joining Hines, Ms. Schugart spent eight years with Arthur Andersen LLP, where she served both public and private clients in the real estate, construction, finance and banking industries. She holds a Bachelor of Business Administration degree in Accounting from Southwest Texas State University.

David L. Steinbach. Mr. Steinbach joined Hines in 1999. Mr. Steinbach serves as the Chief Investment Officer for us and the general partner for our advisor. Mr. Steinbach also serves as the Chief Investment Officer for Hines Global I and the general partner of its Advisor. In these roles, he is responsible for management of the real estate acquisition program in the U.S. and internationally. He has also served as a Managing Director of the general partner of Hines since February 2011. He has been responsible for the acquisition of over \$4 billion in assets for various Hines affiliates in the U.S. and internationally. Prior to this role, he served in various roles in which he was responsible for acquisitions, asset management and property dispositions on behalf of Hines REIT, Hines Global I and the Core Fund, both in the U.S. and internationally. He graduated from Texas A&M University with a Bachelors and Masters in Business Administration.

Kevin L. McMeans. Mr. McMeans joined Hines in 1992. Mr. McMeans serves as the Chief Asset Management Officer for us and the general partner of our Advisor. Mr. McMeans also serves as Asset Management Officer for Hines Global I and the general partner of its Advisor. Mr. McMeans has also been the Asset Management Officer of Hines REIT and the general partner of its Advisor since April 2008. He has also served as the Asset Management Officer of the Core Fund since January 2005. Since February 2015, he has served as the Senior Managing Director of Investment Management of the general partner of Hines. Prior to February 2015, he also served as a Managing Director of Investment Management of the general partner of Hines. In these roles, he is responsible for overseeing the management of the various investment properties owned by each of the funds in the U.S. and internationally. He previously served as the Chief Financial Officer of Hines Corporate Properties, an investment venture established by Hines with a major U.S. pension fund, from 2001 through June 2004. In this role, Mr. McMeans was responsible for negotiating and closing debt financings, underwriting and evaluating new investments, negotiating and closing sale transactions and overseeing the administrative and financial reporting requirements of the venture and its investors. Before joining Hines, Mr. McMeans spent four and a half years at Deloitte & Touche LLP in the audit department. He graduated from Texas A&M University with a B.S. in Computer Science.

Ryan T. Sims. Mr. Sims joined Hines in August 2003. Mr. Sims serves as Chief Financial Officer and Secretary for us and the general partner of our Advisor. Effective November 1, 2011, Mr. Sims was appointed to serve in the position of Chief Financial Officer and Secretary for Hines Global I and the general partner of its Advisor. Effective November 1, 2011, Mr. Sims was also appointed to the position of Chief Financial Officer and Secretary of Hines REIT, the general partner of its Advisor and the Core Fund. Also since 2011, Mr. Sims has served as the Chief Financial Officer and Secretary of HMS and of the general partner of the adviser to HMS. In these positions, Mr. Sims will be responsible for the oversight of financial operations, equity and debt financing activities, investor relations, accounting, financial reporting, tax, legal, compliance and administrative functions in the U.S. and internationally. Prior to this time, Mr. Sims served as the Chief Accounting Officer for Hines Global I and the general partner of its Advisor since their inception in December 2008. Mr. Sims also served as the Chief Accounting Officer for Hines REIT, the general partner of its Advisor and the Core Fund since April 2008. In these roles, he was responsible for the oversight of the accounting, financial reporting and SEC reporting functions, as well as the Sarbanes-Oxley compliance program in the U.S. and internationally. He was also responsible for establishing the companies' accounting policies and ensuring compliance with those policies in the U.S. and internationally. He has also previously served as a Senior Controller for Hines REIT and the general partner of its Advisor from August 2003 to April 2008 and the Core Fund from July 2004 to April 2008. Prior to joining Hines, Mr. Sims was a manager in the audit practice of Arthur Andersen LLP and Deloitte & Touche LLP, serving clients primarily in the real estate industry. He holds a Bachelor of Business Administration degree in Accounting from Baylor University and is a certified public accountant.

J. Shea Morgenroth. Mr. Morgenroth joined Hines in October 2003. Mr. Morgenroth serves as Chief Accounting Officer and Treasurer for us and the general partner of our Advisor. Effective November 1, 2011, Mr. Morgenroth was appointed to serve in the position of Chief Accounting Officer and Treasurer for Hines Global I and the general partner of its Advisor. Effective November 1, 2011, Mr. Morgenroth was also appointed to the position of Chief Accounting Officer and Treasurer for Hines Global I and the general partner of its Advisor. Effective November 1, 2011, Mr. Morgenroth was also appointed to the position of Chief Accounting Officer and Treasurer of Hines REIT and the general partner of its Advisor. In these roles, Mr. Morgenroth is responsible for the oversight of the treasury, accounting, financial reporting and SEC reporting functions, as well as the Sarbanes-Oxley compliance program in the U.S. and internationally. Prior to his appointment, Mr. Morgenroth served as a Senior Controller for Hines Global I and the general partner of its Advisor since their inception in December 2008 and for Hines REIT and the general partner of its Advisor since January 2008 and as a Controller for Hines REIT and its advisor from October 2003 to January 2008. In these roles, he was responsible for the management of the accounting, financial reporting and SEC reporting functions. Prior to joining Hines, Mr. Morgenroth was a manager in the audit practice of Arthur Andersen LLP and Deloitte & Touche LLP, serving clients primarily in the real estate industry. He holds a Bachelor of Business Administration degree in Accounting from Texas A&M University and is a certified public accountant.

Code of Business Conduct and Ethics

Our board of directors has adopted a Code of Business Conduct and Ethics, which is applicable to our directors and officers, including our principal executive officer, principal financial officer, principal accounting officer or controller and other persons performing similar functions, whether acting in their capacities as our officers or in their capacities as officers of our Advisor or its general partner. The Code of Business Conduct and Ethics covers topics including conflicts of interest, confidentiality of information, full and fair disclosure, reporting of violations and compliance with laws and regulations. Our Code of Business Conduct and Ethics is available, free of charge, on the Corporate Governance section of our website, http:// www.hinessecurities.com/reits/hines-global-reit-2/corporate-governance/. You may also obtain a copy of this code by writing to: Hines Global REIT II Investor Relations, 2800 Post Oak Boulevard, Suite 5000, Houston, Texas 77056-6118. Waivers from our Code of Business Conduct and Ethics are discouraged, but any waivers from the Code of Business Conduct and Ethics that relate to any executive officer or director must be approved by our Nominating and Corporate-governance/ within four business days of any such waiver.

Item 11. Executive Compensation

Compensation of Directors

The following table sets forth information regarding compensation to our directors during 2014:

2014 Director Compensation

Name	-	es Earned r Paid in Cash	Option Awards	Non-Equity Incentive Plan Compensation	Change in Pension Value and Non- Qualified Deferred Compensation Earnings	All Other Compensation	Сог	Total npensation
Humberto "Burt" Cabañas	\$	45,220	\$—	\$—	\$—	\$—	\$	45,220
Dougal A. Cameron	\$	46,837	\$—	\$—	\$—	\$—	\$	46,837
John O. Niemann, Jr.	\$	46,837	\$—	\$—	\$—	\$—	\$	46,837
Jeffery C. Hines and Charles M. Baughn ⁽¹⁾	\$		\$—	\$ —	\$—	\$ —	\$	_

(1) Messrs. Hines and Baughn, who are employees of Hines, receive no compensation for serving as members of our board of directors.

We paid our independent directors an annual fee of \$40,000 (prorated for a partial term), and a fee of \$2,000 for each meeting of the board (or any committee thereof) attended in person. If a committee meeting was held on the same day as a meeting of our board of directors, each independent director received \$1,500 for each committee meeting attended in person on

such day. We also paid each of our independent directors a fee of \$750 for each board or committee meeting attended via teleconference, regardless of its length.

We paid the following annual retainers (prorated for a partial term) to the Chairpersons of our board committees for 2014:

- \$7,500 to the Chairperson of the Conflicts Committee of the board;
- \$10,000 to the Chairperson of the Audit Committee of the board;
- \$5,000 to the Chairperson of the Compensation Committee of the board; and
- \$5,000 to the Chairperson of the Nominating and Corporate Governance Committee of the board.

All directors are reimbursed by us for reasonable out-of-pocket expenses incurred in connection with attendance at board or committee meetings.

Executive Compensation

We have no employees. Our day-to-day management functions are performed by our Advisor and its affiliates. All of our executive officers are employed by and receive compensation from our Advisor or its affiliates, for all of their services to the Hines organization, including their service as our executive officers. The compensation received by our executive officers is not paid or determined by us, but rather by our Advisor or affiliates of our Advisor based on all the services provided by these individuals to the Hines organization, including us. As a result, we do not have and our compensation committee has not considered, a compensation policy or program for our executive officers and have not included a "Compensation Discussion and Analysis," or "Compensation Committee Report" in this Annual Report on Form 10-K.

Item 12. Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters

The following table shows, as of March 2, 2015, the amount of our common stock beneficially owned (unless otherwise indicated) by (1) any person who is known by us to be the beneficial owner of more than 5% of our outstanding common shares, (2) our directors, (3) our executive officers, and (4) all of our directors and executive officers as a group. Except as otherwise indicated, all shares are owned directly, and the owner of such shares has the sole voting and investment power with respect thereto.

		Common Shares Owned	Beneficially
Name of Beneficial Owner ⁽¹⁾	Position	Number of Common Shares	Percentage of Class
Jeffrey C. Hines	Chairman of the Board	223,896.256	18.9%
Humberto "Burt" Cabañas	Independent Director	11,111.111	*
Dougal A. Cameron	Independent Director	_	%
John O. Niemann, Jr.	Independent Director	11,135.160	*
Charles M. Baughn	Director	11,267.779	1.0%
Sherri W. Schugart	President and Chief Executive Officer	11,136.682	*
Ryan T. Sims	Chief Financial Officer and Secretary	2,777.930	*
David L. Steinbach	Chief Investment Officer	2,778.387	*
Kevin L. McMeans	Asset Management Officer	2,778.463	*
J. Shea Morgenroth	Chief Accounting Officer and Treasurer	2,781.568	*
Hines Global REIT II Associates Limited Partnership ⁽⁴⁾		(4	⁴⁾ % (4
All directors and executive officers as a group		279,663.336	23.6%

* Amount represents less than 1%

- (1) The address of each person listed is c/o Hines Global REIT II, Inc., 2800 Post Oak Boulevard, Suite 5000, Houston, Texas 77056-6618.
- (2) For purposes of this table, "beneficial ownership" is determined in accordance with Rule 13d-3 under the Exchange Act, pursuant to which a person is deemed to have "beneficial ownership" of shares of our stock that the person has the right to acquire within 60 days. For purposes of computing the percentage of outstanding shares of the Company's stock held by each person or group of persons named in the table, any shares that such person or persons have the right to acquire within 60 days of March 2, 2015 are deemed to be outstanding, but are not deemed to be outstanding for the purpose of computing the percentage ownership of any other persons.
- (3) Includes all common shares owned directly by Hines Global REIT II Investor Limited Partnership. Mr. Hines is deemed to be the beneficial owner of the shares owned by Hines Global REIT II Investor Limited Partnership. Mr. Hines may also be deemed to be the beneficial owner of interests held by HALP II.
- (4) Hines Global REIT II Associates Limited Partnership owns: (i) 21,111.111 OP Units in the Operating Partnership and (ii) the Special OP Units, if any. Limited partners in the Operating Partnership may request repurchase of their OP Units for cash or, at our option, common shares on a one-for-one basis, beginning one year after such OP Units were issued. The holder of the Special OP Units is entitled to distributions from the Operating Partnership under certain circumstances. In addition, under the Advisory Agreement, if we are not advised by an entity affiliated with Hines, Hines or its affiliates may cause the Operating Partnership to purchase some or all of the Special OP Units or any other OP Units then held by such entities for cash (or in certain cases, a promissory note) or our shares as determined by the seller.

Item 13. Certain Relationships and Related Transactions, and Director Independence

Our Advisor

We do not have employees. Subject to the supervision of our board of directors, our day-to-day operations are conducted by our Advisor in accordance with the Advisory Agreement. Our Advisor is an affiliate of Hines and is wholly-owned, indirectly, by, or for the benefit of, the Chairman of our board of directors, Jeffrey C. Hines, and his father, Gerald D. Hines. All of our executive officers are employed by, and all of our executive officers actively participate in, the management of our Advisor and its affiliates. Jeffrey C. Hines serves as the Chairman of the Managers of the general partner of our Advisor and Charles M. Baughn serves as a Manager of the general partner of our Advisor.

Our executive officers have control and primary responsibility for the management decisions of our Advisor, including the selection of investment properties to be recommended to our board of directors, the negotiations for these investments, and the property management and leasing of properties we acquire directly. Our Advisory Agreement commenced on August 15, 2014 and has a one year term that may be renewed for an unlimited number of successive periods (up to one year at a time) upon the mutual consent of the parties. Renewals of the agreement must be approved by the Conflicts Committee. The Advisory Agreement may be terminated:

- immediately by us (i) in the event our Advisor commits fraud, criminal conduct, willful misconduct or negligent breach of fiduciary duty, (ii) upon the bankruptcy of our Advisor or its involvement in similar insolvency proceedings or (iii) in the event of a material breach of the Advisory Agreement by our Advisor that remains uncured after 10 days' written notice;
- without cause or penalty by us or by our Advisor upon 60 days' written notice; or
- immediately by our Advisor upon our bankruptcy or involvement in similar insolvency proceedings or any material breach of the Advisory Agreement by us that remains uncured after 10 days' written notice.

The Advisor and its affiliates receive compensation and are reimbursed for certain expenses in connection with services provided to us. These payments are summarized below. In the event the Advisory Agreement is terminated, our Advisor will be paid all earned, accrued and unpaid compensation and expense reimbursements within 30 days. Upon termination, we may also be obligated to purchase certain ownership interests owned by our Advisor or other affiliates of Hines under certain circumstances.

The following summarizes fees our Advisor earned under the Advisory Agreement during 2014:

- Under the Advisory Agreement, we reimbursed our Advisor for all costs incurred by our Advisor or its affiliates in connection with our initial public offering up to a limit (when combined with commissions and the Dealer Manager fee paid in connection with the offering) of 15% of aggregate gross proceeds. Such costs consist of, among other costs, expenses of the Company's organization, actual legal, accounting, bona fide out-of-pocket itemized and detailed due diligence costs, printing, filing fees, transfer agent costs, postage, escrow fees, data processing fees, advertising and sales literature and other offering-related costs. We incurred approximately \$4.2 million to our Advisor or its affiliates during 2014 as reimbursement for issuer and organizational costs it incurred as a result of our initial public offering.
- Under the Advisory Agreement, we pay our Advisor an acquisition fee in connection with investments we make, equal to 2.25% of (i) the purchase price of real estate investments acquired, including any debt attributable to such investments, or the total principal amounts borrowed under any loans made or acquired directly by us, or (ii) when we make an investment or make or acquire a loan indirectly through another entity, such investment's pro rata share of the gross asset value of real estate investments held by that entity, including any debt attributable to such investments, or the total principal amount borrowed under any loans made or acquired by that entity. Under the Advisory Agreement, we also reimburse our Advisor for certain acquisition expenses incurred in connection with the purchase of real estate investments. The Advisor earned \$570,375 in acquisition fees during the year ended December 31, 2014, and we will reimburse our Advisor approximately \$33,581 related to acquisition expenses incurred in 2014.
- Under the Advisory Agreement, we pay our Advisor a monthly asset management fee of 0.0625% per month of the cost of our real estate investments at the end of each month; provided that, if our board of directors has determined an estimated net asset value per share, then, with respect to real estate investments included in the board of director's determination, the asset management fees will be equal to 0.0625% per month of the most recently determined value of such real estate investments at the end of each month. In September 2014, our Advisor agreed to waive the asset management fees otherwise payable to it for the quarter ended December 31, 2014, to the extent that our modified funds from operations ("MFFO") for the quarter ended December 31, 2014, as disclosed in our Annual Report on Form 10-K for such year, amounts to less than 100% of the aggregate distributions declared to our stockholders for the quarter ended December 31, 2014. As a result of this waiver, the Advisor waived \$16,258 of asset management fees payable to it for the three months ended December 31, 2014. Furthermore, our Advisor also agreed to waive the asset management fees otherwise payable to it for each of the quarters ended March 31, 2015 and June 30, 2015, to the extent that our MFFO for such quarters, as disclosed in our Quarterly Report on Form 10-Q, amounts to less than 100% of the aggregate distributions declared to our stockholders for each respective quarter. As a result of these fee waivers, cash flows from operations that would have been paid to the Advisor for asset management fees may be available to pay distributions to stockholders. These fee waivers are not deferrals and accordingly, any fees that are waived will not be paid to the Advisor in cash at any time in the future.
- Under the Advisory Agreement, we pay our Advisor a disposition fee in an amount equal to (a) 1.0% of the market value determined in connection with a listing of our common stock on a national securities exchange, or 1.0% of the gross consideration received or to be received by us or our stockholders upon the occurrence of any other liquidity event involving us or the Operating Partnership, pursuant to which our stockholders receive in exchange for their shares of our common stock, cash, listed securities, securities redeemable for cash, or a combination thereof, or (b) 1.0% of the gross sales price upon the sale or transfer of one or more real estate investments (including a sale of all of our real estate investments). Even if the Advisor, its affiliates or related parties receive a disposition fee, we may still be obligated to pay fees or commissions to another third party. However, when a real estate or brokerage fee is payable in connection with a particular transaction, the amount equal to the lesser of (i) one-half of a competitive real estate or brokerage commissions and (ii) 1.0% of the gross sales price and, when added to the sum of all real estate or brokerage commissions paid to unaffiliated parties, may not exceed the lesser of (x) a competitive real estate or brokerage commission or (y) an amount equal to 6.0% of the gross sales price. The advisor did not earn any disposition fees during the year ended December 31, 2014.
- Likewise, under the Advisory Agreement, we will reimburse our Advisor for all expenses paid or incurred by the
 Advisor in connection with the services provided to us, subject to the limitation that we will not reimburse the Advisor
 for any amount by which our operating expenses (including the asset management fee) at the end of the four preceding
 fiscal quarters exceeds the greater of: (A) 2.0% of our average invested assets, or (B) 25.0% of our net income
 determined without reduction for any additions to reserves for depreciation, bad debts or other similar non-cash
 reserves and excluding any gain from the sale of our assets for that period. Notwithstanding the above, we may

reimburse the Advisor for expenses in excess of this limitation if a majority of the independent directors determines that such excess expenses are justified based on unusual and non-recurring factors. We did not reimburse the Advisor and its affiliates for such expenses during the year ended December 31, 2014.

We also agreed to indemnify our Advisor against losses it incurs in connection with its performance of its obligations under the Advisory Agreement, subject to terms and conditions in the Advisory Agreement.

The holder of the Special OP Units in the Operating Partnership will be entitled to receive distributions from the Operating Partnership in an amount equal to 15% of distributions, including from sales of real estate investments, refinancings and other sources, but only after our stockholders have received (or are deemed to have received), in the aggregate, cumulative distributions equal to their invested capital plus a 6.0% cumulative, non-compounded annual pre-tax return on such invested capital. The Special OP Units may be converted into OP Units that, at the election of the holder, will be repurchased for cash (or, in the case of (iii) below, a promissory note) or our shares, following: (i) the listing of our common stock on a national securities exchange, or (ii) a merger, consolidation or sale of substantially all of our assets or any similar transaction, a purchase, tender or exchange offer that results in the tender or exchange of more than 50% of our outstanding common shares, or any transaction pursuant to which a majority of our board of directors then in office are replaced or removed or (iii) the occurrence of certain events that result in the termination or non-renewal of our Advisory Agreement.

The Dealer Manager

The Dealer Manager manages our public offering and serves as our dealer manager. The Dealer Manager is an affiliate of Hines and is wholly-owned, indirectly, by Jeffrey C. Hines and his father, Gerald D. Hines. We entered into an amended and restated dealer manager agreement with the Dealer Manager on December 12, 2014 related to our initial public offering.

Pursuant to the Dealer Manager Agreement for our public offering, we paid the Dealer Manager a selling commission of up to 7.0% of gross offering proceeds and a dealer manager fee of up to 3.0% of gross offering proceeds, both of which are recorded as an offset to additional paid-in-capital in our financial statements. Pursuant to separately negotiated agreements, the Dealer Manager may reallow all of its commission and a portion of its dealer manager fee to broker-dealers participating in our public offering. The Dealer Manager, in its sole discretion, may also pay to participating broker dealers out of its dealer manager fee a marketing fee and may reimburse participating broker dealers for distribution and marketing-related costs and expenses, such as costs associated with attending or sponsoring conferences and technology costs. No selling commissions or dealer manager fees will be paid for sales under our distribution reinvestment plan.

During the year ended December 31, 2014:

- the Dealer Manager earned approximately \$27,511 in dealer manager fees in connection with the Offering, a portion of which the Dealer Manager re-allowed to participating broker dealers; and
- the Dealer Manager earned approximately \$60,697 in selling commissions in connection with the Offering, a portion of which the Dealer Manager re-allowed to participating broker dealers.

Property Management Agreement

Hines or its affiliates manage the properties in which we invest. When we acquire properties directly, we expect that we will pay Hines property management fees, leasing fees, tenant construction fees and other fees customarily paid to a property manager. Hines is wholly-owned by Jeffrey C. Hines and his father, Gerald D. Hines.

During the year ended December 31, 2014, Hines earned the following approximate amounts pursuant to property management agreements under which Hines manages some of our properties:

- approximately \$1,769 in property management fees; and
- approximately \$1,260, for all costs Hines incurred in providing property management and leasing services pursuant to
 the property management and leasing agreements. Included in this reimbursement of operating costs are the cost of
 personnel and overhead expenses related to such personnel located at the property as well as off-site personnel located
 in Hines' headquarters and regional offices, to the extent the same relate to or support the performance of Hines' duties
 under the agreements.

Hines Credit Facility

In addition, in December 2014, we entered into a credit facility with Hines. See <u>Note 5</u>—Debt Financing in the Notes to the Consolidated Financial Statements for a description of the loan agreement.

Ownership Interests

The Operating Partnership

On July 31, 2013, we, along with HALP II, formed the Operating Partnership. On September 6, 2013, Hines Global REIT II Investor Limited Partnership, an affiliate of the Advisor, purchased 1,111.111 shares of common stock for \$10,000 and was admitted as our initial stockholder. Additionally, HALP II made initial capital contributions to the Operating Partnership of \$190,000 and, accordingly, owned a 95.0% limited partner interest in the Operating Partnership. As of December 31, 2014, HALP II owned an 4.8% limited partner interest in the Operating Partnership.

Policies and Procedures for Review of Related Party Transactions

Potential conflicts of interest exist among us, Hines, our Advisor and other affiliates of Hines in relation to our existing agreements and how we will operate. Currently, three of our five directors are independent directors, and each of our independent directors, serve on the Conflicts Committee of our board of directors. The Conflicts Committee reviews and approves all matters that our board of directors believes may involve conflicts of interest.

In order to reduce the effect of certain potential conflicts of interest, our charter contains a number of restrictions relating to transactions we enter into with Hines, our Advisor, any of our directors and their respective affiliates. These restrictions include, among others, the following:

- Except as otherwise permitted in our charter, we will not engage in transactions with Hines, our Advisor, any of our directors or their respective affiliates unless a majority of our directors, including a majority of our independent directors not otherwise interested in the transaction, approve such transactions as fair and reasonable to us and on terms and conditions not less favorable to us than those available from unaffiliated third parties.
- We will not purchase a property from Hines, our Advisor, any of our directors or their respective affiliates without a determination by a majority of our directors, including a majority of our independent directors not otherwise interested in the transaction, that the transaction is fair and reasonable to us and at a price no greater than the cost of the property to Hines, our Advisor, any of our directors or their respective affiliates, unless there is substantial justification for any amount that exceeds such cost and such excess amount is determined to be reasonable. In all cases where assets are acquired from Hines, our Advisor, any of our directors or their respective affiliates, the fair market value of such assets will be determined by an independent expert selected by our independent directors. In no event will we acquire any property from Hines, our Advisor, any of our directors or their respective affiliates at a price that exceeds the appraised value of the property; provided that in the case of a development, redevelopment or refurbishment project that we agree to acquire prior to completion of the project, the appraised value will be based upon the completed value of the project as determined at the time the agreement to purchase the property is entered into. We will not sell or lease a property to Hines, our Advisor, any of our directors or their respective affiliates unless a majority of our directors, including a majority of the independent directors not otherwise interested in the transaction, determine the transaction is fair and reasonable to us. Even following these procedures, Hines and its affiliates (including our officers and directors) may make substantial profits in connection with the acquisition or sale of properties from other investment vehicles sponsored by Hines, our Advisor, any of our directors or their respective affiliates.
- We will not enter into joint ventures with Hines, our Advisor, any of our directors or their respective affiliates, unless a majority of our independent directors approves such transaction as being fair and reasonable to us and determines that our investment is on terms substantially similar to the terms of third parties making comparable investments.
- We will not make any loan to Hines, our Advisor, any of our directors or their respective affiliates except in the case of loans to our wholly owned subsidiaries and loans in which an independent expert has appraised the underlying asset. Any loans to us by Hines, our Advisor, any of our directors or their respective affiliates must be approved by a majority of our directors, including a majority of the independent directors not otherwise interested in the transaction, as fair, competitive and commercially reasonable, and on terms no less favorable to us than loans between unaffiliated parties under the same circumstances.

Director Independence

We have a total of five directors, three of whom are independent of us, the Advisor and our respective affiliates. Our full board of directors has determined that each of our independent directors is independent within the meaning of the applicable provisions set forth in our charter; the applicable requirements set forth in the Exchange Act and the applicable SEC rules; and although our shares are not listed on the New York Stock Exchange ("NYSE"), the independence rules set forth in the NYSE Listed Company Manual. Our board applies the NYSE rules governing independence as part of its policy of maintaining strong corporate governance practices.

Our charter defines an "independent director" as a person who has not been, directly or indirectly, associated with the Sponsor or the Advisor within the previous two years by virtue of:

- Ownership of an interest in the Advisor or Sponsor or their Affiliates, other than the Company or any other Affiliate with securities registered under the Exchange Act;
- Employment by the Advisor or Sponsor or their Affiliates;
- Service as an officer, trust manager or director of the Advisor or Sponsor or their Affiliates, other than as a director of the Company or any other Affiliate with securities registered under the Exchange Act;
- Performance of services, other than as a director, for the Company or any other Affiliate with securities registered under the Exchange Act;
- Service as a director, trust manager or trustee of more than three real estate investment trusts advised by the Advisor or organized by the Sponsor; or
- Maintenance of a material business or professional relationship with the Advisor or Sponsor or any of their Affiliates.

Our board of directors may establish committees it deems appropriate to address specific areas in more depth than may be possible at a full board meeting, provided that the majority of the members of each committee are independent directors. Members of each committee will be appointed by our board of directors to serve a one year term or until their successors are duly elected and qualify or until their earlier death, resignation, retirement or removal. Our board of directors has established an audit committee, conflicts committee, nominating and corporate governance committee and compensation committee. Our independent directors are the sole members of all of these committees so that these important areas can be addressed in more depth than may be possible at a full board meeting and to also ensure that these areas are addressed by non-interested members of the board. A copy of each such charter is available on our website, www.hinessecurities.com/reits/hines-global-reit-2.

Item 14. Principal Accounting Fees and Services

Deloitte & Touche LLP, the member firms of Deloitte Touche Tohmatsu, and their respective affiliates (collectively "Deloitte & Touche") serve as our principal accounting firm. Deloitte & Touche audited our financial statements for the year ended December 31, 2014 and for the Period from July 31, 2013 (date of inception) through December 31, 2013. Deloitte & Touche reports directly to our Audit Committee.

Fees

Deloitte & Touche's aggregate fees billed to us for the fiscal year ended December 31, 2014 and for the period from July 31, 2013 (date of inception) through December 31, 2013 are as follows:

	2014	Pe (da	riod from July 31, 2013 te of inception) through December 31, 2013
Audit Fees:	\$ 140,000	\$	47,121
Audit-Related Fees ⁽¹⁾ :	\$ 155,000	\$	31,414
Tax Fees:	\$ 	\$	—
All Other Fees:	\$ —	\$	_
Total Fees:	\$ 295,000	\$	78,535

(1) These fees primarily relate to the issuance of comfort letters, accounting consultations and other attestation services.

Pre-approval Policies and Procedures

Our Audit Committee has adopted a pre-approval policy requiring the Audit Committee to pre-approve all audit and permissible non-audit services to be performed by Deloitte & Touche. In determining whether or not to pre-approve services, the Audit Committee will consider whether the service is a permissible service under the rules and regulations promulgated by the SEC, and, if permissible, the potential effect of such services on the independence of Deloitte & Touche. All services performed for us in 2014 were pre-approved or ratified by our Audit Committee.

PART IV

Item 15. Exhibits, Financial Statement Schedules

(a)(1) Financial Statements

Hines Global REIT II, Inc.

Consolidated Financial Statements — as of December 31, 2014 and 2013 and for the Year Ended December 31, 2014 and the period from July 31, 2013 (date of inception) to December 31, 2013.

Report of Independent Registered Public Accounting Firm	<u>54</u>
Audited Consolidated Financial Statements	
Consolidated Balance Sheets	<u>55</u>
Consolidated Statements of Operations	<u>56</u>
Consolidated Statements of Equity	<u>57</u>
Consolidated Statements of Cash Flows	<u>58</u>
Notes to Consolidated Financial Statements	<u>59</u>

(2) Financial Statement Schedules

Schedule III — Real Estate Assets and Accumulated Depreciation is set forth beginning on page 84 hereof.

All other schedules for which provision is made in the applicable accounting regulations of the SEC are not required under the related instructions or are not applicable and therefore have been omitted.

(b) Exhibits

Reference is made to the Index beginning on page 86 for a list of all exhibits filed as a part of this report.

* * * * * *

Schedule III — Real Estate Assets and Accumulated Depreciation December 31, 2014

						Initial Cost		Gross Amount at Which Carried at 12/31/2014								
Description ^(a)	Location	Er	ncumbrances	Land	Buildings and Land Improvements Total		Costs Capitalized Subsequent to Acquisition		Land		Buildings and nprovements	Total ^(b)	umulated eciation ^(c)	Date of Construction	Date Acquired	
2819 Loker Avenue East	t Carlsbad, California	\$	24,200,000	\$7,200,000	\$	14,170,000	\$21,370,000	\$	—	\$7,200,000	\$	14,170,000	\$21,370,000	\$ (14,579)	1998	December-14
		\$	24,200,000	\$7,200,000	\$	14,170,000	\$21,370,000	\$		\$7,200,000	\$	14,170,000	\$21,370,000	\$ (14,579)		

(a) Asset consists of an industrial building.

(b) The aggregate cost for federal income tax purposes is \$21.5 million as of December 31, 2014.

(c) Real estate assets are depreciated or amortized using the straight-line method over the useful lives of the assets by class. The estimated useful lives for computing depreciation are generally 10 years for furniture and fixtures, 15-20 years for electrical and mechanical installations and 40 years for buildings.

The changes in total real estate assets for the year ended December 31, 2014:

Gross real estate assets

Balance, beginning of period	\$
Additions during the period:	
Acquisitions	 21,370,000
Balance, end of period	\$ 21,370,000
Accumulated Depreciation	
Balance, beginning of period	\$ _
Depreciation	 (14,579)
Balance, end of period	\$ (14,579)

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the Registrant has duly caused this Annual Report to be signed on its behalf by the undersigned, thereunto duly authorized representative.

HINES GLOBAL REIT II, INC. (registrant)

March 25, 2015

By: /s/ Sherri W. Schugart Sherri W. Schugart President and Chief Executive Officer

Pursuant to the requirements of the Securities Exchange Act of 1934, this Report has been signed below by the following persons on behalf of the Registrant and in the capacities indicated on March 25, 2015.

Signature	Title	Date
/s/ Jeffrey C. Hines Jeffrey C. Hines	Chairman of the Board of Directors	March 25, 2015
/s/ Sherri W. Schugart Sherri W. Schugart	President and Chief Executive Officer (Principal Executive Officer)	March 25, 2015
/s/ Ryan T. Sims Ryan T. Sims	Chief Financial Officer and Secretary (Principal Financial Officer)	March 25, 2015
/s/ J. Shea Morgenroth J. Shea Morgenroth	Chief Accounting Officer and Treasurer (Principal Accounting Officer)	March 25, 2015
/s/ Charles M. Baughn Charles M. Baughn	Director	March 25, 2015
/s/ Humberto "Burt" Cabañas Humberto "Burt" Cabañas	Director	March 25, 2015
/s/ Dougal A. Cameron Dougal A. Cameron	Director	March 25, 2015
/s/ John O. Niemann, Jr.	Director	March 25, 2015

John O. Niemann, Jr.

SUPPLEMENTAL INFORMATION TO BE FURNISHED WITH REPORTS FILED PURSUANT TO SECTION 15(d) OF THE ACT BY REGISTRANTS WHICH HAVE NOT REGISTERED SECURITIES PURSUANT TO SECTION 12 OF THE ACT

No annual report or proxy materials have been sent to our stockholders for the period covered by this Annual Report on Form 10-K. If a proxy statement is delivered to more than ten of our stockholders with respect to an annual or other meeting of stockholders, copies of such materials will be furnished to the SEC at that time. We will deliver to our stockholders a copy of this Annual Report on Form 10-K.

INDEX TO EXHIBITS

Exhibit No.	Description
3.1	Articles of Amendment and Restatement of Hines Global REIT II, Inc. (filed as Exhibit 3.1 to Pre-Effective Amendment No. 5 to the Registrant's Registration Statement on Form S-11 (File No. 333-191106), as amended and supplemented (the "Registration Statement") on August 15, 2014 and incorporated by reference herein).
3.2	Articles Supplementary of Hines Global REIT II, Inc. (filed as Exhibit 1.1 to Post-Effective Amendment No. 1 to the Registration Statement on December 12, 2014 and incorporated by reference herein).
3.3	Amended and Restated Bylaws of Hines Global REIT II, Inc. (filed as Exhibit 3.2 to Pre-Effective Amendment No. 5 to the Registration Statement on August 15, 2014 and incorporated by reference herein).
10.1*	Amended and Restated Agreement of Limited Partnership of Hines Global REIT II Properties LP, dated as of December 12, 2014
10.2	Advisory Agreement, dated as of August 15, 2014, among Hines Global REIT II Advisors LP, Hines Global REIT II Properties LP and Hines Global REIT II, Inc. (filed as Exhibit 10.2 to Pre-Effective Amendment No. 5 to the Registration Statement on August 15, 2014 and incorporated herein by reference).
10.3	Amended and Restated Escrow Agreement, dated as of December 12, 2014, by and among Hines Securities, Inc., Hines Global REIT II, Inc. and UMB Bank, N.A. (filed as Exhibit 10.3 to Post-Effective Amendment No. 1 to the Registration Statement on December 12, 2014 and incorporated herein by reference).
10.4	Form of Indemnification Agreement entered into between Hines Global REIT II, Inc. and each of the following persons as of August 15, 2014: Jeffrey C. Hines, Charles M. Baughn, Humberto Cabañas, Dougal A. Cameron, John O. Niemann, Jr., Sherri W. Schugart, Ryan T. Sims, David L. Steinbach, Kevin L. McMeans and J. Shea Morgenroth (filed as Exhibit 10.4 to the Registration Statement on September 11, 2013 and incorporated herein by reference).
10.5	Waiver to Hines Global REIT II Advisory Agreement (filed as Exhibit 10.1 to the Registrant's Current Report on Form 8-K on September 24, 2014 and incorporated herein by reference).
10.6	Purchase and Sale Agreement, dated as of November 12, 2014, by and between Hines Interests Limited Partnership and Canoga-Rincon Loker Industrial, LLC (filed as Exhibit 10.1 to the Registrant's Current Report on Form 8-K filed November 20, 2014 and incorporated herein by reference).
10.7	Assignment of Contract of Purchase and Sale, dated as of November 14, 2014, by and between Hines Interests Limited Partnership and HGREIT II 2819 Loker LP (filed as Exhibit 10.2 to the Registrant's Current Report on Form 8-K filed November 20, 2014 and incorporated herein by reference).
10.8	Uncommitted Loan Agreement, dated as of December 15, 2014, by and between Hines Global REIT II, Inc., as borrower, and Hines Interests Limited Partnership, as lender (filed as Exhibit 10.3 to the Registrant's Current Report on Form 8-K filed December 19, 2014 and incorporated herein by reference).
10.9	Selected Dealer Agreement, dated as of December 31, 2014, by and between Hines Global REIT II, Inc., Hines Securities, Inc., Hines Global REIT II Advisors LP and Ameriprise Financial Services, Inc. (filed as Exhibit 10.9 to Post-Effective Amendment No. 3 to the Registration Statement on March 5, 2015, and incorporated by reference herein).
21.1*	List of Subsidiaries of Hines Global REIT II, Inc.
31.1*	Certification
31.2*	Certification
32.1*	Certification of Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C., Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002. Pursuant to SEC Release 34-47551 this Exhibit is furnished to the SEC herewith and shall not be deemed to be "filed."
101*	The following materials from Hines Global REIT II, Inc.'s Annual Report on Form 10-K for the year ended December 31, 2014, filed on March 25, 2015, formatted in XBRL (eXtensible Business Reporting Language): (i) Consolidated Balance Sheets, (ii) Consolidated Statements of Operations, (iii) Consolidated Statements of Equity (Deficit), (iv) Consolidated Statements of Cash Flows, and (v) Notes to the Consolidated Financial Statements.
*	Filed herewith

Exhibit 10.1

AMENDED AND RESTATED

LIMITED PARTNERSHIP AGREEMENT

OF

HINES GLOBAL REIT II PROPERTIES LP

A DELAWARE LIMITED PARTNERSHIP

December 12, 2014

		Page
Artic	le 1 DEFINED TERMS	<u>2</u>
Artic	e 2 PARTNERSHIP FORMATION AND IDENTIFICATION	<u>12</u>
2.1	Formation	<u>12</u>
2.2	Name, Office and Registered Agent	<u>12</u>
2.3	Partners	<u>12</u>
2.4	Term and Dissolution	<u>13</u>
2.5	Filing of Certificate and Perfection of Limited Partnership	<u>13</u>
2.6	Certificates Describing Partnership Units and Special OP Units	<u>13</u>
Artic	ele 3 BUSINESS OF THE PARTNERSHIP	<u>14</u>
Artic	le 4 CAPITAL CONTRIBUTIONS AND ACCOUNTS	14
4.1	Capital Contributions	<u>14</u>
4.2	Additional Capital Contributions and Issuances of Additional Partnership Interests	<u>14</u>
4.3	Additional Funding	<u>17</u>
4.4	Capital Accounts	<u>17</u>
4.5	Percentage Interests	<u>18</u>
4.6	No Interest On Contributions	<u>18</u>
4.7	Return Of Capital Contributions	<u>18</u>
4.8	No Third Party Beneficiary	<u>18</u>
Artic	le 5 PROFITS AND LOSSES; DISTRIBUTIONS	<u>19</u>
5.1	Allocation of Profit and Loss.	<u>19</u>
5.2	Distribution of Cash	21
5.3	REIT Distribution Requirements	23
5.4	No Right to Distributions in Kind	23
5.5	Limitations on Return of Capital Contributions	23
5.6	Asset Acquisition Distributions	23
5.7	Distributions Upon Liquidation	24
5.8	Substantial Economic Effect	<u>24</u>
Artic	ele 6 RIGHTS, OBLIGATIONS AND POWERS OF THE GENERAL PARTNER	24
6.1	Management of the Partnership.	<u>24</u> <u>24</u>
6.2	Delegation of Authority	<u>24</u> <u>27</u>
6.3	Indemnification and Exculpation of Indemnitees	<u>27</u> <u>27</u>
6.4	Liability of the General Partner	<u>27</u> <u>28</u>
6.5	Reimbursement of General Partner	<u>28</u> <u>30</u>
6.6	Outside Activities	<u>30</u>
6.7	Employment or Retention of Affiliates	<u>30</u>
0.7	Employment of Actention of Aminutes	<u> </u>

TABLE OF CONTENTS

6.8	General Partner Participation	<u>31</u>
6.9	Title to Partnership Assets	<u>31</u>
6.10	Redemption of REIT Shares	<u>31</u>
6.11	No Duplication of Fees or Expenses	<u>31</u>
Article	7 CHANGES IN GENERAL PARTNER	<u>31</u>
7.1	Transfer of the General Partner's Partnership Interest	<u>32</u>
7.2	Admission of a Substitute or Additional General Partner	<u>33</u>
7.3	Effect of Bankruptcy, Withdrawal, Death or Dissolution of a General Partner.	<u>34</u>
7.4	Removal of a General Partner	<u>35</u>
Article	8 RIGHTS AND OBLIGATIONS OF THE LIMITED PARTNERS	<u>36</u>
8.1	Management of the Partnership	<u>36</u>
8.2	Power of Attorney	<u>36</u>
8.3	Limitation on Liability of Limited Partners	<u>36</u>
8.4	Redemption Right	<u>36</u>
	Redemption or Conversion of Special OP Units and Partnership Units owned by the Advisor	•
8.5	or its Affiliates.	<u>38</u>
	9 TRANSFERS OF LIMITED PARTNERSHIP INTERESTS	<u>41</u>
9.1	Purchase for Investment	<u>41</u>
9.2	Restrictions on Transfer of Limited Partnership Interests	<u>41</u>
9.3	Admission of Substitute Limited Partner	<u>43</u>
9.4	Rights of Assignees of Partnership Interests	<u>44</u>
9.5	Effect of Bankruptcy, Death, Incompetence or Termination of a Limited Partner	<u>44</u>
9.6	Joint Ownership of Interests	<u>44</u>
	10 BOOKS AND RECORDS; ACCOUNTING; TAX MATTERS	<u>44</u>
10.1	Books and Records	<u>45</u>
10.2	Custody of Partnership Funds; Bank Accounts	<u>45</u>
10.3	Fiscal and Taxable Year	<u>45</u>
10.4	Annual Tax Information and Report	<u>45</u>
10.5	Tax Matters Partner; Tax Elections; Special Basis Adjustments	<u>45</u>
10.6	Reports to Limited Partners	<u>46</u>
10.7	Safe Harbor Election	<u>46</u>
Article	11 AMENDMENT OF AGREEMENT; MERGER	<u>47</u>
	12 GENERAL PROVISIONS	<u>47</u>
12.1	Notices	<u>47</u>
12.2	Survival of Rights	<u>47</u>
12.3	Additional Documents	<u>48</u>
12.4	Severability	<u>48</u>

12.5	Entire Agreement	<u>48</u>
12.6	Pronouns and Plurals	<u>48</u>
12.7	Headings	<u>48</u>
12.8	Counterparts	<u>48</u>
12.9	Governing Law	<u>48</u>

EXHIBITS

EXHIBIT A — Partners, Capital Contributions and Percentage Interests EXHIBIT B — Notice of Exercise of Redemption Right

AMENDED AND RESTATED LIMITED PARTNERSHIP AGREEMENT

OF

HINES GLOBAL REIT II PROPERTIES LP

This Amended and Restated Limited Partnership Agreement (this "Agreement") is entered into this 12th day of December, 2014, between Hines Global REIT II, Inc., as the General Partner, Hines Global REIT II Associates Limited Partnership, as a Limited Partner, and the Limited Partners set forth on <u>Exhibit A</u> attached hereto. Capitalized terms used herein but not otherwise defined shall have the meanings given them in <u>Article 1</u>.

AGREEMENT

WHEREAS, the parties hereto entered into a Limited Partnership Agreement on August 15, 2014 (the "Original Agreement") and now desire to amend and restate the Original Agreement;

WHEREAS, the General Partner intends to qualify as a real estate investment trust under the Internal Revenue Code of 1986, as amended;

WHEREAS, Hines Global REIT II Properties LP (the "Partnership"), was formed on July 31, 2013 as a limited partnership under the laws of the State of Delaware, pursuant to a Certificate of Limited Partnership filed with the Office of the Secretary of State of the State of Delaware on July 31, 2013;

WHEREAS, the General Partner desires to conduct its current and future business primarily through the Partnership;

WHEREAS, in furtherance of the foregoing, the General Partner desires to contribute certain assets to the Partnership from time to time;

WHEREAS, in exchange for the General Partner's contribution of assets, the parties desire that the Partnership issue Partnership Units to the General Partner in accordance with the terms of this Agreement;

WHEREAS, the Limited Partner has contributed and it and future Limited Partners may contribute certain of their property to the Partnership in exchange for Partnership Units or the Special OP Units in accordance with the terms of this Agreement;

WHEREAS, in furtherance of the Partnership's business, the Partnership will acquire Properties and other assets from time to time by means of the contribution of such Properties or other assets to the Partnership by the owners thereof in exchange for Partnership Units; and

WHEREAS, the parties hereto wish to establish herein their respective rights and obligations in connection with all of the foregoing and certain other matters;

NOW, THEREFORE, in consideration of the foregoing, of mutual covenants between the parties hereto, and of other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

ARTICLE 1

DEFINED TERMS

The following defined terms used in this Agreement shall have the meanings specified below:

"ACT" means the Delaware Revised Uniform Limited Partnership Act, as it may be amended from time to time.

"ADDITIONAL FUNDS" has the meaning set forth in Section 4.3 hereof.

"ADDITIONAL SECURITIES" means any additional REIT Shares (other than REIT Shares issued in connection with a redemption pursuant to <u>Section 8.4</u> hereof) or rights, options, warrants or convertible or exchangeable securities containing the right to subscribe for or purchase REIT Shares, as set forth in <u>Section 4.2(a)(ii)</u>.

"ADMINISTRATIVE EXPENSES" means (i) all administrative and operating costs and expenses incurred by the Partnership, (ii) those administrative costs and expenses of the General Partner, including any salaries or other payments to directors, officers or employees of the General Partner, and any accounting and legal expenses of the General Partner, which expenses, the Partners have agreed, are expenses of the Partnership and not the General Partner, (iii) costs and expenses relating to the formation and continuity of existence and operation of the General Partner and any Subsidiaries thereof (which Subsidiaries shall, for purposes hereof, be included within the definition of General Partner), including taxes, fees and assessments associated therewith, (iv) costs and expenses relating to any Offering and registration of securities by the General Partner and all statements, reports, fees and expenses incidental thereto, including, without limitation, underwriting discounts and selling commissions applicable to any such Offering, and any costs and expenses associated with any claims made by any holders of such securities or any underwriters or placement agents thereof, (v) costs and expenses associated with any repurchase of any securities by the General Partner, (vi) costs and expenses associated with the preparation and filing of any periodic or other reports and communications by the General Partner under federal, state or local laws or regulations, including filings with the Commission, (vii) costs and expenses associated with compliance by the General Partner with laws, rules and regulations promulgated by any regulatory body, including the Commission and any securities exchange, (viii) costs and expenses associated with any 401(k) plan, incentive plan, bonus plan or other plan providing for compensation for the employees of the General Partner, (ix) costs and expenses incurred by the General Partner relating to any issuing or redemption of Partnership Interests and (x) all other operating or administrative costs of the General Partner incurred in the ordinary course of its business on behalf of or in connection with the Partnership; provided, however, that Administrative Expenses shall not include any administrative costs and expenses incurred by the General Partner that are attributable to

Properties or partnership interests in a Subsidiary Partnership that are owned by the General Partner directly.

"ADVISOR" or "ADVISORS" has the meaning set forth in the Advisory Agreement.

"ADVISORY AGREEMENT" means the agreement between the General Partner, the Partnership and the Advisor pursuant to which the Advisor will direct or perform the day-to-day business affairs of the General Partner.

"AFFILIATE" means, with respect to any Person, (i) any Person directly or indirectly owning, controlling or holding, with the power to vote, ten percent or more of the outstanding voting securities of such other Person; (ii) any Person ten percent or more of whose outstanding voting securities are directly or indirectly owned, controlled or held, with the power to vote, by such other Person; (iii) any Person directly or indirectly controlling, controlled by or under common control with such other Person; (iv) any executive officer, director, trustee or general partner of such other Person and (v) any legal entity for which such Person acts as an executive officer, director, trustee or general partner.

"AGREED VALUE" means the fair market value of a Partner's non-cash Capital Contribution as of the date of contribution as agreed to by such Partner and the General Partner. The names and addresses of the Partners, number of Partnership Units issued to each Partner, and the Agreed Value of any non-cash Capital Contributions as of the date of contribution are set forth on <u>Exhibit A</u>.

"AGREEMENT" has the meaning set forth in the preamble.

"APPLICABLE PERCENTAGE" has the meaning set forth in Section 8.4(b) hereof.

"ASSET" means any Property, Mortgage, other debt or other investment (other than investments in bank accounts, money market funds or other current assets) owned by the General Partner, directly or indirectly through one or more of its Affiliates.

"ASSET ACQUISITION CONTRIBUTION" has the meaning set forth in <u>Section 4.2(a)</u> (ii) hereof.

"ASSET ACQUISITION DISTRIBUTION" has the meaning set forth in <u>Section 5.6</u> hereof.

"ASSET ACQUISITION REDEMPTION" has the meaning set forth in <u>Section 8.4</u> hereof.

"BUSINESS DAY" means any day on which the New York Stock Exchange is open for trading.

"CAPITAL ACCOUNT" has the meaning set forth in Section 4.4 hereof.

"CAPITAL CONTRIBUTION" means the total amount of cash, cash equivalents, and the Agreed Value of any Property or other asset (other than cash) contributed or agreed to be contributed, as the context requires, to the Partnership by each Partner pursuant to the terms of this Agreement. Any reference to the Capital Contribution of a Partner shall include the Capital Contribution made by a predecessor holder of the Partnership Interest of such Partner. Any reference to a Capital Contribution shall not include any amounts contributed to the Partnership which are generated from the operation or sale of a General Partner Property acquired in whole or in part with the proceeds from an Asset Acquisition Distribution, an Asset Acquisition Redemption or an Asset Acquisition Contribution.

"CARRYING VALUE" means, with respect to any asset of the Partnership, the asset's adjusted net basis for federal income tax purposes or, in the case of any asset contributed to the Partnership, the fair market value of such asset at the time of contribution, reduced by any amounts attributable to the inclusion of liabilities in basis pursuant to Section 752 of the Code, except that the Carrying Values of all assets may, at the discretion of the General Partner, be adjusted to equal their respective fair market values (as determined by the General Partner), in accordance with the rules set forth in Regulations Section 1.704-1(b)(2)(iv)(f), as provided for in Section 4.4. In the case of any asset of the Partnership that has a Carrying Value that differs from its adjusted tax basis, the Carrying Value shall be adjusted by the amount of depreciation, depletion and amortization calculated for purposes of the allocations of net profit and net loss pursuant to <u>Article 5</u> hereof rather than the amount of depreciation, depletion and amortization determined for federal income tax purposes.

"CASH AMOUNT" means an amount of cash per Partnership Unit equal to the lesser of (i) the Value of the REIT Shares Amount on the date of receipt by the General Partner of a Notice of Redemption or (ii) the applicable Redemption Price determined by the General Partner.

"CERTIFICATE" means any instrument or document that is required under the laws of the State of Delaware, or any other jurisdiction in which the Partnership conducts business, to be signed and sworn to by the Partners of the Partnership (either by themselves or pursuant to the power-of-attorney granted to the General Partner in <u>Section 8.2</u> hereof) and filed for recording in the appropriate public offices within the State of Delaware or such other jurisdiction to perfect or maintain the Partnership as a limited partnership, to effect the admission, withdrawal, or substitution of any Partner of the Partnership, or to protect the limited liability of the Limited Partners as limited partners under the laws of the State of Delaware or such other jurisdiction.

"CLASS" means a class of REIT Shares or Partnership Units, as the context may require.

"CLASS A REIT SHARES" means the REIT Shares classified as "Class A" shares in the Charter.

"CLASS A UNIT" means a Partnership Unit entitling the holder thereof to the rights of a holder of a Class A Unit as provided in this Agreement.

"CHARTER" means the Amended and Restated Articles of Incorporation of the General Partner filed with the Maryland State Department of Assessments and Taxation, as amended or restated from time to time.

"CODE" means the Internal Revenue Code of 1986, as amended from time to time, or any successor statute thereto. Reference to any particular provision of the Code shall mean that provision as in effect from time to time, as the same may be amended, and any successor provision thereto, as interpreted by any applicable regulations as in effect from time to time.

"COMMISSION" means the U.S. Securities and Exchange Commission.

"CONVERSION FACTOR" means 1.0, provided that in the event that the General Partner (i) declares or pays a dividend on its outstanding REIT Shares in REIT Shares or makes a distribution to all holders of its outstanding REIT Shares in REIT Shares, (ii) subdivides its outstanding REIT Shares, or (iii) combines its outstanding REIT Shares into a smaller number of REIT Shares, the Conversion Factor shall be adjusted by multiplying the Conversion Factor by a fraction, the numerator of which shall be the number of REIT Shares issued and outstanding on the record date for such dividend, distribution, subdivision or combination (assuming for such purposes that such dividend, distribution, subdivision or combination has occurred as of such time), and the denominator of which shall be the actual number of REIT Shares (determined without the above assumption) issued and outstanding on such date and, provided further, that in the event that an entity other than an Affiliate of the General Partner shall become General Partner pursuant to any merger, consolidation or combination of the General Partner with or into another entity (the "Successor Entity"), the Conversion Factor shall be adjusted by multiplying the Conversion Factor by the number of shares of the Successor Entity into which one REIT Share is converted pursuant to such merger, consolidation or combination, determined as of the date of such merger, consolidation or combination. Any adjustment to the Conversion Factor shall become effective immediately after the effective date of such event retroactive to the record date, if any, for such event; provided, however, that if the General Partner receives a Notice of Redemption after the record date, but prior to the effective date of such dividend, distribution, subdivision or combination, the Conversion Factor shall be determined as if the General Partner had received the Notice of Redemption immediately prior to the record date for such dividend, distribution, subdivision or combination. A separate Conversion Factor shall be determined for each Class of Partnership Units by taking into account only the outstanding REIT Shares having the same Class designation as the applicable Class of Partnership Units.

"DEFAULTING LIMITED PARTNER" has the meaning set forth in Section 5.2(c).

"DIRECTOR" has the meaning set forth in the Charter.

"EVENT OF BANKRUPTCY" as to any Person means the filing of a petition for relief as to such Person as debtor or bankrupt under the Bankruptcy Code of 1978 or similar provision of law of any jurisdiction (except if such petition is contested by such Person and has been dismissed within 90 days); insolvency or bankruptcy of such Person as finally determined by a court proceeding; filing by such Person of a petition or application to accomplish the same or for the appointment of a receiver or a trustee for such Person or a substantial part of his assets; commencement of any proceedings relating to such Person as a debtor under any other reorganization, arrangement, insolvency, adjustment of debt or liquidation law of any jurisdiction, whether now in existence or hereinafter in effect, either by such Person or by another, provided that if such proceeding is commenced by another, such Person indicates his approval of such proceeding, consents thereto or acquiesces therein, or such proceeding is contested by such Person and has not been finally dismissed within 90 days.

"EXCEPTED HOLDER LIMIT" has the meaning set forth in the Charter.

"GENERAL PARTNER" means Hines Global REIT II, Inc., a Maryland corporation, and any Person who becomes a substitute or additional General Partner as provided herein, and any of their successors as General Partner.

"GENERAL PARTNER LOAN" has the meaning set forth in Section 5.2(c) hereof.

"GENERAL PARTNER PROPERTY" has the meaning set forth in Section 4.2(a)(i) hereof.

"GENERAL PARTNER PROPERTY AMOUNTS" has the meaning set forth in <u>Section</u> <u>4.2(a)(ii)</u> hereof.

"GENERAL PARTNERSHIP INTEREST" means a Partnership Interest held by the General Partner that is a general partnership interest.

"INDEMNITEE" means the General Partner, the Advisor or any of its Affiliates or any employee, director or Affiliate of the General Partner or the Partnership.

"INDEPENDENT DIRECTORS" has the meaning set forth in the Charter.

"JOINT VENTURE" means those joint venture, co-investment, co-ownership or partnership arrangements in which the General Partner or any of its subsidiaries is a co-venturer or general partner established to acquire or hold Assets.

"LIMITED PARTNER" means any Person named as a Limited Partner on <u>Exhibit A</u> attached hereto (including without limitation the Special OP Unitholder), and any Person who becomes a Substitute Limited Partner, in such Person's capacity as a Limited Partner in the Partnership.

"LIMITED PARTNERSHIP INTEREST" means the ownership interest of a Limited Partner in the Partnership at any particular time, including the right of such Limited Partner to any and all benefits to which such Limited Partner may be entitled as provided in this Agreement and in the Act, together with the obligations of such Limited Partner to comply with all the provisions of this Agreement and of such Act.

"LISTING" means the listing of the REIT Shares on a national securities exchange. Upon such Listing, the REIT Shares shall be deemed "Listed." "MORTGAGES" means, in connection with mortgage financing provided, invested in, participated in or purchased, all of the notes, deeds of trust, security interests or other evidences of indebtedness or obligations, which are secured or collateralized by Real Property owned by the borrowers under such notes, deeds of trust, security interests or other evidences of indebtedness or obligations.

"NET SALES PROCEEDS" means, in the case of a transaction described in clause (i) (A) of the definition of Sale, the proceeds of any such transaction less the amount of selling expenses incurred by or on behalf of the General Partner or the Partnership, including all real estate commissions, closing costs and legal fees and expenses. In the case of a transaction described in clause (i)(B) of such definition. Net Sales Proceeds means the proceeds of any such transaction less the amount of selling expenses incurred by or on behalf of the General Partner or the Partnership, including any legal fees and expenses and other selling expenses incurred in connection with such transaction. In the case of a transaction described in clause (i)(C) of such definition, Net Sales Proceeds means the proceeds of any such transaction actually distributed to the General Partner or the Partnership from the Joint Venture less the amount of any selling expenses, including legal fees and expenses incurred by or on behalf of the General Partner (other than those paid by the Joint Venture). In the case of a transaction or series of transactions described in clause (i)(D) of the definition of Sale, Net Sales Proceeds means the proceeds of any such transaction (including the aggregate of all payments under a Mortgage or in satisfaction thereof other than regularly scheduled interest payments) less the amount of selling expenses incurred by or on behalf of the General Partner or the Partnership, including all commissions, closing costs and legal fees and expenses. In the case of a transaction described in clause (i) (E) of such definition, Net Sales Proceeds means the proceeds of any such transaction less the amount of selling expenses incurred by or on behalf of the General Partner or the Partnership, including any legal fees and expenses and other selling expenses incurred in connection with such transaction. In the case of a transaction described in clause (ii) of the definition of Sale, Net Sales Proceeds means the proceeds of such transaction or series of transactions less all amounts generated thereby which are reinvested in one or more Assets within 180 days thereafter and less the amount of any real estate commissions, closing costs, and legal fees and expenses and other selling expenses incurred by or allocated to the General Partner or the Partnership in connection with such transaction or series of transactions. Net Sales Proceeds shall also include any amounts that the General Partner determines, in its discretion, to be economically equivalent to proceeds of a Sale. Net Sales Proceeds shall not include any reserves established by the General Partner in its sole discretion.

"NOTICE OF REDEMPTION" means the Notice of Exercise of Redemption Right substantially in the form attached as <u>Exhibit B</u> hereto.

"OFFER" has the meaning set forth in Section 7.1(c)(ii) hereof.

"OFFERING" means the offer and sale of REIT Shares to the public.

"OP UNITHOLDERS" means all holders of Partnership Interests other than the Special OP Unitholder in its capacity as holder of the Special OP Unit.

"ORIGINAL LIMITED PARTNER" means the Limited Partners designated as "Original Limited Partners" on Exhibit A hereto.

"OWNERSHIP LIMIT" has the meaning set forth in the Charter.

"PARTNER" means any General Partner or Limited Partner.

"PARTNER NONRECOURSE DEBT MINIMUM GAIN" has the meaning set forth in Regulations Section 1.704-2(i). A Partner's share of Partner Nonrecourse Debt Minimum Gain shall be determined in accordance with Regulations Section 1.704-2(i)(5).

"PARTNERSHIP" has the meaning set forth in the recitals.

"PARTNERSHIP INTEREST" means an ownership interest in the Partnership held by either a Limited Partner or the General Partner and includes any and all benefits to which the holder of such a Partnership Interest may be entitled as provided in this Agreement, together with all obligations of such Person to comply with the terms and provisions of this Agreement.

"PARTNERSHIP LOAN" has the meaning set forth in Section 5.2(c) hereof.

"PARTNERSHIP MINIMUM GAIN" has the meaning set forth in Regulations Section 1.704-2(d). In accordance with Regulations Section 1.704-2(d), the amount of Partnership Minimum Gain is determined by first computing, for each Partnership nonrecourse liability, any gain the Partnership would realize if it disposed of the property subject to that liability for no consideration other than full satisfaction of the liability, and then aggregating the separately computed gains. A Partner's share of Partnership Minimum Gain shall be determined in accordance with Regulations Section 1.704-2(g)(1).

"PARTNERSHIP RECORD DATE" means the record date established by the General Partner for the distribution of cash pursuant to <u>Section 5.2</u> hereof, which record date shall be the same as the record date established by the General Partner for a distribution to its shareholders of some or all of its portion of such distribution.

"PARTNERSHIP UNIT" means a fractional, undivided share of the Partnership Interests of all Partners issued hereunder excluding the Partnership Interests represented by Special OP Units. The allocation of Partnership Units among the Partners shall be as set forth on <u>Exhibit A</u>, as such Exhibit may be amended from time to time.

"PERCENTAGE INTEREST" means the percentage ownership interest in the Partnership of each Partner, as determined by dividing the Partnership Units owned by a Partner by the total number of Partnership Units then outstanding. The Percentage Interest of each Partner shall be as set forth on <u>Exhibit A</u>, as such Exhibit may be amended from time to time.

"PERSON" means an individual, corporation, partnership, limited liability company, estate, trust (including a trust qualified under Sections 401(a) or 501(c)(17) of the Code), a portion of a trust permanently set aside for or to be used exclusively for the purposes described in Section 642(c) of the Code, association, private foundation within the meaning of Section 509

(a) of the Code, joint stock company or other entity and also includes a group as that term is used for purposes of Section 13(d)(3) of the Securities Exchange Act of 1934, as amended from time to time, and a group to which an Excepted Holder Limit applies.

"PROPERTY" means, as the context requires, all or a portion of each Real Property acquired by the General Partner, directly or indirectly through joint venture or co-ownership arrangements or other partnership or investment entities.

"PROSPECTUS" means the same as that term is defined in Section 2(10) of the Securities Act, including a preliminary prospectus, an offering circular as described in Rule 256 of the general rules and regulations under the Securities Act, or, in the case of an intrastate offering, any document by whatever name known, utilized for the purpose of offering and selling REIT Shares to the public.

"REAL PROPERTY" means land, rights in land (including leasehold interests), and any buildings, structures, improvements, furnishings, fixtures and equipment located on or used in connection with land and rights or interests in land.

"REDEMPTION" has the meaning set forth in Section 8.4(a).

"REDEMPTION PRICE" means the Value of the REIT Shares Amount on the date of receipt by the General Partner of a Notice of Redemption discounted by any applicable discount that would apply had the Partnership Units first been converted to REIT Shares and then redeemed by the General Partner pursuant to the General Partner's existing redemption plan, if any (provided further, that in determining any such discount, to the extent it is based on a holding period, such REIT Shares will be deemed to have been held for the same period of time as the related underlying Partnership Units had been held by the applicable holder).

"REDEMPTION RIGHT" has the meaning set forth in Section 8.4(a) hereof.

"REGULATIONS" means the Federal income tax regulations promulgated under the Code, as amended and as hereafter amended from time to time. Reference to any particular provision of the Regulations shall mean that provision of the Regulations on the date hereof and any successor provision of the Regulations.

"REGULATORY ALLOCATIONS" has the meaning set forth in Section 5.1(h) hereof.

"REIT" means a corporation, trust, association or other legal entity (other than a real estate syndication) that qualifies as a real estate investment trust under Sections 856 through 860 of the Code, and any successor or other provisions of the Code relating to real estate investment trusts (including provisions as to the attribution of ownership of beneficial interests therein) and the regulations promulgated thereunder.

"REIT SHARE" means a common share of beneficial interest in the General Partner (or successor entity, as the case may be).

"REIT SHARES AMOUNT" means, with respect to Tendered Units of a Class, a number of REIT Shares of the corresponding REIT Share Class equal to the product of the number of Partnership Units of such Class offered for exchange by a Tendering Party, multiplied by the Conversion Factor for such Class of Tendered Units, as adjusted to and including the Specified Redemption Date; provided that in the event the General Partner issues to all holders of REIT Shares of such Class rights, options, warrants or convertible or exchangeable securities entitling the shareholders to subscribe for or purchase REIT Shares of such Class, or any other securities or property (collectively, the "rights"), and the rights have not expired at the Specified Redemption Date, then the REIT Shares Amount shall also include the rights issuable to a holder of the REIT Shares Amount of REIT Shares of such Class on the record date fixed for purposes of determining the holders of REIT Shares entitled to rights.

"RELATED PARTY" means, with respect to any Person, any other Person whose ownership of shares of the General Partner's capital stock would be attributed to the first such Person under Code Section 544 (as modified by Code Section 856(h)(1)(B)).

"RESTRICTION NOTICE" has the meaning set forth in Section 8.4(e) hereof.

"SAFE HARBOR" means, the election described in the Safe Harbor Regulation, pursuant to which a partnership and all of its partners may elect to treat the fair market value of a partnership interest that is transferred in connection with the performance of services as being equal to the liquidation value of that interest.

"SAFE HARBOR ELECTION" means the election by a partnership and its partners to apply the Safe Harbor, as described in the Safe Harbor Regulation and Internal Revenue Service Notice 2005-43, issued on May 19, 2005.

"SAFE HARBOR REGULATION" means Proposed Treasury Regulations Section 1.83-3 (1) issued on May 19, 2005.

"SALE" means (i) any transaction or series of transactions whereby: (A) the General Partner or the Partnership directly or indirectly (except as described in other subsections of this definition) sells, grants, transfers, conveys, or relinquishes its ownership of any Property or portion thereof, including the lease of any Property consisting of a building only, and including any event with respect to any Property which gives rise to a significant amount of insurance proceeds or condemnation awards; (B) the General Partner or the Partnership directly or indirectly (except as described in other subsections of this definition) sells, grants, transfers, conveys, or relinquishes its ownership of all or substantially all of the interest of the General Partner or the Partnership in any Joint Venture in which it is a co-venturer or partner; (C) any Joint Venture directly or indirectly (except as described in other subsections of this definition) in which the General Partner or the Partnership as a co-venturer or partner sells, grants, transfers, conveys, or relinquishes its ownership of any Property or portion thereof, including any event with respect to any Property which gives rise to insurance claims or condemnation awards; (D) the General Partner or the Partnership directly or indirectly (except as described in other subsections of this definition) sells, grants, conveys or relinquishes its interest in any Mortgage or portion thereof (including with respect to any Mortgage, all payments thereunder or in

satisfaction thereof other than regularly scheduled interest payments) of amounts owed pursuant to such Mortgage and any event which gives rise to a significant amount of insurance proceeds or similar awards; or (E) the General Partner or the Partnership directly or indirectly (except as described in other subsections of this definition) sells, grants, transfers, conveys, or relinquishes its ownership of any other Asset not previously described in this definition or any portion thereof, but (ii) not including any transaction or series of transactions specified in clause (i)(A) through (E) above in which the proceeds of such transaction or series of transactions are reinvested by the General Partner in one or more Assets within 180 days thereafter.

"SECURITIES ACT" means the Securities Act of 1933, as amended from time to time, or any successor statute thereto. Reference to any provision of the Securities Act shall mean such provision as in effect from time to time, as the same may be amended, and any successor provision thereto, as interpreted by any applicable regulations as in effect from time to time.

"SERVICE" means the United States Internal Revenue Service.

"SPECIAL OP UNITS" means units of a series of Partnership Interests, designated as Special OP Units, issued pursuant to <u>Section 4.1</u>. The holder of the Special OP Units shall have the same rights and preferences as a holder of a Partnership Unit under this Agreement that is a Limited Partner except as otherwise set forth in this Agreement.

"SPECIAL OP UNIT DISTRIBUTION" has the meaning set forth in <u>Section 5.2(b)</u> hereof.

"SPECIAL OP UNITHOLDER" means Hines Global REIT II Associates Limited Partnership.

"SPECIAL OP UNIT VALUE" has the meaning set forth in Section 8.5(b)(i) hereof.

"SPECIFIED REDEMPTION DATE" means the first business day of the month that is at least sixty (60) Business Days after the receipt by the General Partner of the Notice of Redemption.

"SUBSIDIARY" means, with respect to any Person, any corporation or other entity of which a majority of (i) the voting power of the voting equity securities or (ii) the outstanding equity interests is owned, directly or indirectly, by such Person.

"SUBSIDIARY PARTNERSHIP" means any partnership of which the partnership interests therein are owned by the General Partner or a direct or indirect subsidiary of the General Partner.

"SUBSTITUTE LIMITED PARTNER" means any Person admitted to the Partnership as a Limited Partner pursuant to <u>Section 9.3</u> hereof.

"SUCCESSOR ENTITY" has the meaning set forth in the definition of "Conversion Factor" contained herein.

"SURVIVOR" has the meaning set forth in <u>Section 7.1(d)</u> hereof.

"TAX MATTERS PARTNER" has the meaning set forth in Section 10.5(a) hereof.

"TENDERED UNITS" has the meaning set forth in <u>Section 8.4(a)</u> hereof.

"TENDERING PARTY" has the meaning set forth in <u>Section 8.4(a)</u> hereof.

"TERMINATION EVENT" has the meaning set forth in <u>Section 8.5(a)</u>.

"TRANSACTION" has the meaning set forth in <u>Section 7.1(c)</u> hereof.

"TRANSFER" has the meaning set forth in <u>Section 9.2(a)</u> hereof.

"VALUE" means for each Class of REIT Shares, the fair market value of that Class of REIT Shares which will equal: (i) if REIT Shares of that Class are Listed, the average closing price per share for the previous thirty Business Days, (ii) if REIT Shares of that Class are not Listed, (a) the most recent offering price per share or share equivalent of REIT Shares of that Class, until December 31st of the year following the year in which the most recently completed offering of REIT Shares of that Class has expired, and (b) thereafter, such price per REIT Share of that Class as a majority of the Directors of the General Partner determines in good faith.

"VALUATION MECHANISMS" has the meaning set forth in Section 8.5(b)(i) hereof.

ARTICLE 2PARTNERSHIP FORMATION AND IDENTIFICATION

2.1 <u>Formation</u>. The Partnership was formed as a limited partnership pursuant to the Act and all other pertinent laws of the State of Delaware, for the purposes and upon the terms and conditions set forth in this Agreement.

2.2 <u>Name, Office and Registered Agent</u>. The name of the Partnership is Hines Global REIT II Properties LP, a Delaware limited partnership. The specified office and place of business of the Partnership shall be 2800 Post Oak Blvd., Suite 5000 Houston, TX 77056-6118. The General Partner may at any time change the location of such office, provided the General Partner gives notice to the Partners of any such change. The name and address of the Partnership's registered agent is The Corporation Trust Company, 1209 Orange Street, Wilmington, Delaware 19801. The sole duty of the registered agent as such is to forward to the Partnership any notice that is served on him as registered agent.

2.3 <u>Partners</u>.

(a) The General Partner of the Partnership is Hines Global REIT II, Inc., a Maryland corporation. Its principal place of business is the same as that of the Partnership.

(b) The Limited Partners are those Persons identified as Limited Partners on Exhibit A hereto, as amended from time to time.

2.4 <u>Term and Dissolution</u>.

(a) The term of the Partnership shall continue in full force and effect until the first to occur of any of the following events:

(i) The occurrence of an Event of Bankruptcy as to a General Partner or the dissolution, removal or withdrawal of a General Partner unless the business of the Partnership is continued pursuant to <u>Section 7.3(b)</u> hereof;

(ii) The passage of ninety (90) days after the sale or other disposition of all or substantially all of the assets of the Partnership (provided that if the Partnership receives an installment obligation as consideration for such sale or other disposition, the Partnership shall continue, unless sooner dissolved under the provisions of this Agreement, until such time as such note or notes are paid in full); or

dissolved.

(iii) The election by the General Partner that the Partnership should be

(b) Upon dissolution of the Partnership (unless the business of the Partnership is continued pursuant to <u>Section 7.3(b)</u> hereof), the General Partner (or its trustee, receiver, successor or legal representative) shall amend or cancel any Certificate(s) and liquidate the Partnership's assets and apply and distribute the proceeds thereof in accordance with <u>Section 5.7</u> hereof. Notwithstanding the foregoing, the liquidating General Partner may either (i) defer liquidation of, or withhold from distribution for a reasonable time, any assets of the Partnership (including those necessary to satisfy the Partnership's debts and obligations), or (ii) distribute the assets to the Partners in kind.

2.5 <u>Filing of Certificate and Perfection of Limited Partnership</u>. The General Partner shall execute, acknowledge, record and file at the expense of the Partnership, any and all amendments to the Certificate(s) and all requisite fictitious name statements and notices in such places and jurisdictions as may be necessary to cause the Partnership to be treated as a limited partnership under, and otherwise to comply with, the laws of each state or other jurisdiction in which the Partnership conducts business.

2.6 <u>Certificates Describing Partnership Units and Special OP Units</u>. At the request of a Limited Partner, the General Partner, at its option, may issue (but in no way is obligated to issue) a certificate summarizing the terms of such Limited Partner's interest in the Partnership, including the number of Partnership Units (and, if applicable the Special Op Units), as of the date of such certificate. Any such certificate (i) shall be in form and substance as approved by the General Partner, (ii) shall not be negotiable and (iii) shall bear a legend to the following effect:

This certificate is not negotiable. The Partnership Units and Special OP Units represented by this certificate are governed by and transferable only in accordance with the provisions of the Amended and Restated Limited Partnership Agreement of Hines Global REIT II Properties LP, as amended from time to time.

ARTICLE 3

BUSINESS OF THE PARTNERSHIP

The purpose and nature of the business to be conducted by the Partnership is (i) to conduct any business that may be lawfully conducted by a limited partnership organized pursuant to the Act, provided, however, that such business shall be limited to and conducted in such a manner as to permit the General Partner at all times to qualify as a REIT, unless the General Partner otherwise ceases to qualify as a REIT, and in a manner such that the General Partner will not be subject to any taxes under Section 857 or 4981 of the Code, (ii) to enter into any partnership, joint venture, co-ownership or other similar arrangement to engage in any of the foregoing or the ownership of interests in any entity engaged in any of the foregoing and (iii) to do anything necessary or incidental to the foregoing. In connection with the foregoing, and without limiting the General Partner's right in its sole and absolute discretion to qualify or cease qualifying as a REIT, the Partners acknowledge that the General Partner intends to qualify as a REIT for federal income tax purposes and upon such qualification the avoidance of income and excise taxes on the General Partner inures to the benefit of all the Partners and not solely to the General Partner. Notwithstanding the foregoing, the Limited Partners agree that the General Partner may terminate its status as a REIT under the Code at any time to the full extent permitted under the Charter. The General Partner on behalf of the Partnership shall also be empowered to do any and all acts and things necessary or prudent to ensure that the Partnership will not be classified as a "publicly traded partnership" for purposes of Section 7704 of the Code.

ARTICLE 4

CAPITAL CONTRIBUTIONS AND ACCOUNTS

4.1 <u>Capital Contributions</u>. The General Partner and the initial Limited Partners have made capital contributions to the Partnership in exchange for the Partnership Interests set forth opposite their names on <u>Exhibit A</u>, as such Exhibit may be amended from time to time. The Partners shall own Partnership Units in the amounts set forth in <u>Exhibit A</u> and shall have a Percentage Interest in the Partnership as set forth in <u>Exhibit A</u>, which Percentage Interest shall be adjusted in <u>Exhibit A</u> from time to time by the General Partner to the extent necessary to reflect accurately exchanges, Redemptions, Capital Contributions, the issuance of additional Partnership Units or similar events having an effect on a Partner's Percentage Interest.

4.2 <u>Additional Capital Contributions and Issuances of Additional Partnership</u> <u>Interests</u>. Except as provided in this <u>Section 4.2</u> or in <u>Section 4.3</u>, the Partners shall have no right or obligation to make any additional Capital Contributions or loans to the Partnership. The General Partner may contribute additional capital to the Partnership, from time to time, and receive additional Partnership Interests in respect thereof, in the manner contemplated in this <u>Section 4.2</u>.

(a) <u>Issuances of Additional Partnership Interests</u>.

General. The General Partner is hereby authorized to cause the (i) Partnership to issue such additional Partnership Interests in the form of Partnership Units for any Partnership purpose at any time or from time to time, including but not limited to Partnership Units issued in connection with acquisitions of properties, to the Partners (including the General Partner) or to other Persons for such consideration and on such terms and conditions as shall be established by the General Partner in its sole and absolute discretion, all without the approval of any Limited Partners. No additional Partnership Interests shall be issued in connection with any amounts paid to the Partnership which are generated from the operation or sale of a property or interest therein acquired either directly or indirectly by the General Partner in whole or in part with the proceeds from an Asset Acquisition Distribution, an Asset Acquisition Redemption or an Asset Acquisition Contribution ("General Partner Property"). The Partners agree that solely for Federal income tax purposes, the General Partner Property shall be treated as being owned by the Partnership. Any additional Partnership Interests issued may be issued in one or more Classes (including the Classes specified in this Agreement or any other Classes), or one or more series of any of such Classes, with such designations, preferences and relative, participating, optional or other special rights, powers and duties, including rights, powers and duties senior to Limited Partnership Interests, all as shall be determined by the General Partner in its sole and absolute discretion and without the approval of any Limited Partner, subject to Delaware law, including, without limitation, (i) the allocations of items of Partnership income, gain, loss, deduction and credit to each such Class or series of Partnership Interests; (ii) the right of each such Class or series of Partnership Interests to share in Partnership distributions; and (iii) the rights of each such Class or series of Partnership Interests upon dissolution and liquidation of the Partnership; provided, however, that no additional Partnership Interests shall be issued to the General Partner unless:

(1) (A) the additional Partnership Interests are issued in connection with an issuance of REIT Shares of or other interests in the General Partner, which shares or interests have designations, preferences and other rights, all such that the economic interests are substantially similar to the designations, preferences and other rights of the additional Partnership Interests issued to the General Partner by the Partnership in accordance with this <u>Section 4.2</u> (without limiting the foregoing, for example, the Partnership shall issue Partnership Interests consisting of Class A Units to the General Partner in connection with the issuance of Class A REIT Shares) and (B) the General Partner shall make a Capital Contribution to the Partnership in an amount equal to the proceeds raised in connection with the issuance of such shares of stock of or other interests in the General Partner;

(2) the additional Partnership Interests are issued in exchange for property owned by the General Partner with a fair market value, as determined by the General Partner, in good faith, equal to the value of the Partnership Interests; or

(3) the additional Partnership Interests are issued to all Partners holding Partnership Units in proportion to their respective Percentage Interests. Without limiting the foregoing, the General Partner is expressly authorized to cause the Partnership to issue Partnership Units for less than fair market value, so long as the General Partner concludes in good faith that such issuance is in the best interests of the General Partner and the Partnership.

Upon Issuance of Additional Securities. The General Partner shall (ii) not issue any Additional Securities other than to all holders of REIT Shares, unless (A) the General Partner shall cause the Partnership to issue to the General Partner, as the General Partner may designate, Partnership Interests or rights, options, warrants or convertible or exchangeable securities of the Partnership having designations, preferences and other rights, all such that the economic interests are substantially similar to those of the Additional Securities, and (B) the General Partner contributes the proceeds from the issuance of such Additional Securities and from any exercise of rights contained in such Additional Securities, directly and through the General Partner, to the Partnership (without limiting the foregoing, for example, the Partnership shall issue Limited Partnership Interests consisting of Class A Units to the General Partner in connection with the issuance of Class A REIT Shares); provided, however, that the General Partner is allowed to issue Additional Securities and use the proceeds from such issuance ("Asset Acquisition Contributions") in connection with an acquisition of a General Partner Property and any Asset Acquisition Contributions are not required to be contributed to the Partnership. As indicated above, the Partners agree that for Federal income tax purposes, General Partner Property (and all associated items of income, gain, loss and deduction) will be treated as being owned by the Partnership and, as such, the General Partner agrees to transfer to the Partnership any amounts it receives from the operation and/or disposition of General Partner Property ("General Partner Property Amounts") and all General Partner Property Amounts shall then be paid by the Partnership in accordance with Section 5.2(b) of this Agreement. Without limiting the foregoing, the General Partner is expressly authorized to issue Additional Securities for less than fair market value, and to cause the Partnership to issue to the General Partner corresponding Partnership Interests, so long as (x) the General Partner concludes in good faith that such issuance is in the best interests of the General Partner and the Partnership, including without limitation, the issuance of REIT Shares and corresponding Partnership Units pursuant to an employee share purchase plan providing for employee purchases of REIT Shares at a discount from fair market value or employee stock options that have an exercise price that is less than the fair market value of the REIT Shares, either at the time of issuance or at the time of exercise, and (y) the General Partner contributes all proceeds from such issuance to the Partnership. For example, in the event the General Partner issues REIT Shares of any Class for a cash purchase price and contributes all of the proceeds of such issuance to the Partnership, the General Partner shall be issued a number of additional Partnership Units having the same Class designation as the issued REIT Shares equal to the product of (A) the number of such REIT Shares of that Class issued by the General Partner, the proceeds of which were so contributed, multiplied by (B) a fraction, the numerator of which is 100%, and the denominator of which is the Conversion Factor for that Class of Partnership Units in effect on the date of such contribution.

(b) <u>Certain Deemed Contributions of Proceeds of Issuance of REIT Shares</u>. Except as otherwise permitted hereunder, in connection with any and all issuances of REIT Shares, the General Partner shall make Capital Contributions to the Partnership of the proceeds therefrom, provided that if the proceeds actually received and contributed by the General Partner are less than the gross proceeds of such issuance as a result of any underwriter's discount or other expenses paid or incurred in connection with such issuance, then the General Partner shall be deemed to have made Capital Contributions to the Partnership in the aggregate amount of the gross proceeds of such issuance and the Partnership shall be deemed simultaneously to have paid such offering expenses in accordance with Section 6.5 hereof and in connection with the required issuance of additional Partnership Units to the General Partner for such Capital Contributions pursuant to Section 4.2(a) hereof.

4.3 <u>Additional Funding</u>. If the General Partner determines that it is in the best interests of the Partnership to provide for additional Partnership funds ("Additional Funds") for any Partnership purpose, the General Partner may (i) cause the Partnership to obtain such funds from outside borrowings, or (ii) elect to have the General Partner or any of its Affiliates provide such Additional Funds to the Partnership through loans or otherwise, provided, however, that the Partnership may not borrow money from its Affiliates, unless a majority of the Directors of the General Partner (including a majority of Independent Directors) not otherwise interested in such transaction approve the transaction as being fair, competitive, and commercially reasonable and no less favorable to the Partnership than comparable loans between unaffiliated parties.

4.4 <u>Capital Accounts</u>.

A separate capital account (each a "Capital Account") shall be maintained (a) for each Partner in accordance with the rules of Treasury Regulations Section 1.704-1(b)(2)(iv), and this Section 4.4 and Article 5 shall be interpreted and applied in a manner consistent therewith. Whenever the Partnership would be permitted to adjust the Capital Accounts of the Partners pursuant to Treasury Regulations Section 1.704-1(b)(2)(iv)(f) to reflect revaluations of Partnership property, the Partnership may so adjust the Capital Accounts of the Partners. In the event that the Capital Accounts of the Partners are adjusted pursuant to Treasury Regulations Section 1.704-1(b)(2)(iv) (f) to reflect revaluations of Partnership property, (i) the Capital Accounts of the Partners shall be adjusted in accordance with Treasury Regulations Section 1.704-1(b)(2)(iv)(g) for allocations of depreciation, depletion, amortization and gain or loss, as computed for book purposes, with respect to such property, (ii) the Partners' distributive shares of depreciation, depletion, amortization and gain or loss, as computed for tax purposes, with respect to such property shall be determined so as to take account of the variation between the adjusted tax basis and book value of such property in the same manner as under Code Section 704(c), and (iii) the amount of upward and/or downward adjustments to the book value of the Partnership property shall be treated as income, gain, deduction and/or loss for purposes of applying the allocation provisions of Article 5. In the event that Code Section 704(c) applies to Partnership property, the Capital Accounts of the Partners shall be adjusted in accordance with Treasury Regulations Section 1.704-1(b)(2)(iv)(g) for allocations of depreciation, depletion, amortization and gain and loss, as computed for book purposes, with respect to such property.

(b) Notwithstanding any provision herein to the contrary, any fees, expenses or other costs of the Partnership that are required to be paid by the General Partner without reimbursement and that are required to be treated as capital contributions to the Partnership for purposes of the Treasury Regulations promulgated under Section 704(b) of the Code, shall be added to the balance of the General Partner's Capital Account.

4.5 Percentage Interests. If the number of outstanding Partnership Units increases or decreases during a taxable year, each Partner's Percentage Interest shall be adjusted by the General Partner effective as of the effective date of each such increase or decrease to a percentage equal to the number of Partnership Units held by such Partner divided by the aggregate number of Partnership Units outstanding after giving effect to such increase or decrease. If the Partners' Percentage Interests are adjusted pursuant to this Section 4.5, the net profits and net losses (and items thereof) for the taxable year in which the adjustment occurs shall be allocated between the part of the year ending on the day when the Partnership's property is revalued by the General Partner and the part of the year beginning on the following day either (i) as if the taxable year had ended on the date of the adjustment or (ii) based on the number of days in each part. The General Partner, in its sole and absolute discretion, shall determine which method shall be used to allocate net profits and net losses (or items thereof) for the taxable year in which the adjustment occurs. The allocation of net profits and net losses (or items thereof) for the earlier part of the year shall be based on the Percentage Interests before adjustment, and the allocation of net profits and net losses (or items thereof) for the later part shall be based on the adjusted Percentage Interests.

4.6 No Interest On Contributions. No Partner shall be entitled to interest on its Capital Contribution.

4.7 <u>**Return Of Capital Contributions**</u>. No Partner shall be entitled to withdraw any part of its Capital Contribution or its Capital Account or to receive any distribution from the Partnership, except as specifically provided in this Agreement. Except as otherwise provided herein, there shall be no obligation to return to any Partner or withdrawn Partner any part of such Partner's Capital Contribution for so long as the Partnership continues in existence.

4.8 No Third Party Beneficiary. No creditor or other third party having dealings with the Partnership shall have the right to enforce the right or obligation of any Partner to make Capital Contributions or loans or to pursue any other right or remedy hereunder or at law or in equity, it being understood and agreed that the provisions of this Agreement shall be solely for the benefit of, and may be enforced solely by, the parties hereto and their respective successors and assigns. None of the rights or obligations of the Partners herein set forth to make Capital Contributions or loans to the Partnership shall be deemed an asset of the Partnership for any purpose by any creditor or other third party, nor may such rights or obligations be sold, transferred or assigned by the Partnership or pledged or encumbered by the Partnership to secure any debt or other obligation of the Partnership or of any of the Partners. In addition, it is the intent of the parties hereto that no distribution to any Limited Partner shall be deemed a return of money or other property in violation of the Act. However, if any court of competent jurisdiction holds that, notwithstanding the provisions of this Agreement, any Limited Partner is obligated to return such money or property, such obligation shall be the obligation of such Limited Partner and not of the General Partner. Without limiting the generality of the foregoing, a deficit Capital Account of a Partner shall not be deemed to be a liability of such Partner nor an asset or property of the Partnership.

ARTICLE 5

PROFITS AND LOSSES; DISTRIBUTIONS

5.1 Allocation of Profit and Loss.

(a) <u>General</u>.

(i) Net income and net loss (or items thereof) of the Partnership for each fiscal year or other applicable period of the Partnership shall be allocated among the OP Unitholders in accordance with their respective Percentage Interests;

(ii) Notwithstanding the foregoing, and subject only to the provisions of paragraphs (b) and (h) and, to the extent set forth in this clause (ii) below, paragraph (c), net income shall first be allocated to the holder of the Special OP Units until such holder has received aggregate allocations of income for all fiscal years equal to the aggregate amount of distributions such holder is entitled to receive or has received with respect to such Special OP Units for such fiscal year and all prior fiscal years, provided that in the event the holder of the Special OP Unit's entitlement to income allocations in such fiscal year would be satisfied pursuant to the allocations set forth in paragraph (c) below, then such allocations shall be made pursuant to paragraph (c) below in lieu of the provisions of this clause (ii).

(b) <u>General Partner Gross Income Allocation</u>. There shall be specially allocated to the General Partner an amount of (i) first, items of Partnership income and (ii) second, items of Partnership gain during each fiscal year or other applicable period, before any other allocations are made hereunder, in an amount equal to the excess, if any, of (A) the cumulative distributions made to the General Partner under <u>Section 6.5(b)</u> hereof, other than distributions which would properly be treated as "guaranteed payments" or which are attributable to the reimbursement of expenses which would properly be deductible by the Partnership, over (B) the cumulative allocations of Partnership income and gain to the General Partner under this <u>Section 5.1(b)</u>.

Special Allocation with Respect to Sales. Items of income, gain, credit, loss (c) and deduction of the Partnership for each fiscal year or other applicable period from Sales, other than any such items allocated under Section 5.1(b), shall be allocated among the Partners in a manner that will, as nearly as possible (after giving effect to the allocations under Sections 5.1(a) and 5.1 (d)) cause the Capital Account balance of each Partner at the end of such fiscal year or other applicable period to equal (i) the amount of the hypothetical distribution that such Partner would receive if the Partnership were liquidated on the last day of such period and all assets of the Partnership, including cash, were sold for cash equal to their Carrying Value, taking into account any adjustments thereto for such period, all liabilities of the Partnership were satisfied in full in cash according to their terms (limited with respect to each nonrecourse liability to the Carrying Value of the assets securing such liability) and Net Sales Proceeds (after satisfaction of such liabilities) were distributed in full pursuant to Section 5.2(b), minus (ii) the sum of such Partner's share of Partnership Minimum Gain and Partner Nonrecourse Debt Minimum Gain and the amount, if any and without duplication, that the Partner would be obligated to contribute to the capital of the Partnership, all computed as of the date of the hypothetical sale of assets.

(d) Nonrecourse Deductions; Minimum Gain Chargeback. Notwithstanding any provision to the contrary, (i) any expense of the Partnership that is a "nonrecourse deduction" within the meaning of Regulations Section 1.704-2(b)(1) shall be allocated in accordance with the Partners' respective Percentage Interests, (ii) any expense of the Partnership that is a "partner nonrecourse deduction" within the meaning of Regulations Section 1.704-2(i)(2) shall be allocated to the Partner that bears the "economic risk of loss" with respect to the liability to which such deductions are attributable in accordance with Regulations Section 1.704-2(i)(1), (iii) if there is a net decrease in Partnership Minimum Gain within the meaning of Regulations Section 1.704-2(f)(1) for any Partnership taxable year, then, subject to the exceptions set forth in Regulations Section 1.704-2(f) (2),(3), (4) and (5), items of gain and income shall be allocated among the Partners in accordance with Regulations Section 1.704-2(f) and the ordering rules contained in Regulations Section 1.704-2 (i), and (iv) if there is a net decrease in Partner Nonrecourse Debt Minimum Gain within the meaning of Regulations Section 1.704-2(i)(4) for any Partnership taxable year, then, subject to the exceptions set forth in Regulations Section 1.704-(2)(g), items of gain and income shall be allocated among the Partners in accordance with Regulations Section 1.704-2(i)(4) and the ordering rules contained in Regulations Section 1.704-2(j). A Partner's "interest in partnership profits" for purposes of determining its share of the excess nonrecourse liabilities of the Partnership within the meaning of Regulations Section 1.752-3(a)(3) shall be such Partner's Percentage Interest.

(e) <u>Qualified Income Offset</u>. If a Partner unexpectedly receives in any taxable year an adjustment, allocation, or distribution described in subparagraphs (4), (5), or (6) of Regulations Section 1.704-1(b)(2)(ii)(d) that causes or increases a deficit balance in such Partner's Capital Account that exceeds the sum of such Partner's shares of Partnership Minimum Gain and Partner Nonrecourse Debt Minimum Gain, as determined in accordance with Regulations Sections 1.704-2(g)(1) and 1.704-2(i)(5), such Partner shall be allocated specially for such taxable year (and, if necessary, later taxable years) items of income and gain in an amount and manner sufficient to eliminate such deficit Capital Account balance as quickly as possible as provided in Regulations Section 1.704-1(b)(2)(ii)(d). This Section 5.1(e) is intended to constitute a "qualified income offset" under Section 1.704-1(b)(2)(ii)(d) of the Regulations and shall be interpreted consistently therewith. After the occurrence of an allocation of income or gain to a Partner in accordance with this Section 5.1(e), to the extent permitted by Regulations Section 1.704-1(b), items of expense or loss shall be allocated to such Partner in an amount necessary to offset the income or gain previously allocated to such Partner under this Section 5.1(e).

(f) <u>Capital Account Deficits</u>. Loss (or items of loss) shall not be allocated to a Limited Partner to the extent that such allocation would cause or increase a deficit in such Partner's Capital Account at the end of any fiscal year (after reduction to reflect the items described in Regulations Section 1.704-1(b)(2)(ii)(d)(4), (5) and (6)) to exceed the sum of such Partner's shares of Partnership Minimum Gain and Partner Nonrecourse Debt Minimum Gain, as determined in accordance with Regulations Sections 1.704-2(g)(1) and 1.704-2(i)(5). Any net loss in excess of that limitation shall be allocated to the General Partner. After the occurrence of an allocation of net loss to the General Partner in accordance with this <u>Section 5.1(f)</u>, to the extent permitted by Regulations Section 1.704-1(b), Profit shall be allocated to such Partner in an amount necessary to offset the net loss previously allocated to such Partner under this <u>Section 5.1(f)</u>.

(g) <u>Allocations Between Transferor and Transferee</u>. If a Partner transfers any part or all of its Partnership Interest, the distributive shares of the various items of profit and loss allocable among the Partners during such fiscal year of the Partnership shall be allocated between the transferor and the transferee Partner either (i) as if the Partnership's fiscal year had ended on the date of the transfer, or (ii) based on the number of days of such fiscal year that each was a Partner without regard to the results of Partnership activities in the respective portions of such fiscal year in which the transferor and the transferee were Partners. The General Partner, in its sole and absolute discretion, shall determine which method shall be used to allocate the distributive shares of the various items of Profit and profit and loss between the transferor and the transferee Partner.

(h) <u>Special Allocations of Class-Specific Items</u>. To the extent that any items of income, gain, loss or deduction of the General Partner are allocable to a specific Class or Classes of REIT Shares as provided in the Prospectus, such items, or an amount equal thereto, shall be specially allocated to the Class or Classes of Partnership Units corresponding to such Class or Classes of REIT Shares.

(i) <u>Curative Allocations</u>. The allocations set forth in <u>Sections 5.1(d)</u>, <u>5.1(e)</u> and <u>5.1(f)</u> of this Agreement (the "Regulatory Allocations") are intended to comply with certain requirements of the Regulations. The General Partner is authorized to offset all Regulatory Allocations either with other Regulatory Allocations or with special allocations of other items of Partnership income, gain, loss or deduction pursuant to this <u>Section 5.1(i)</u>. Therefore, notwithstanding any other provision of this <u>Section 5.1</u> (other than the Regulatory Allocations), the General Partner shall make such offsetting special allocations of Partnership income, gain, loss or deduction in whatever manner it deems appropriate so that, after such offsetting allocations are made, each Partner's Capital Account is, to the extent possible, equal to the Capital Account balance such Partner would have had if the Regulatory Allocations were not part of this Agreement and all Partnership items were allocated pursuant to <u>Sections 5.1(a)</u>, <u>5.1(c)</u>, <u>5.1(g)</u> and <u>5.1(h)</u>.

5.2 <u>Distribution of Cash</u>.

(a) The Partnership may distribute cash on a quarterly (or, at the election of the General Partner, more or less frequent) basis, in an amount determined by the General Partner in its sole and absolute discretion, to the Partners who are Partners on the Partnership Record Date with respect to such quarter (or other distribution period) in accordance with <u>Section 5.2(b)</u>; provided, however, that if a new or existing Partner acquires an additional Partnership Interest in exchange for a Capital Contribution on any date other than a Partnership Record Date, the cash distribution attributable to such additional Partnership Interest relating to the Partnership Record Date next following the issuance of such additional Partnership Interest shall be reduced in the proportion equal to one minus (i) the number of days that such additional Partnership Record Date and the immediately preceding Partnership Record Date.

(b) Except for distributions pursuant to Section 5.6 in connection with an Asset Acquisition Distribution and Section 5.7 in connection with the dissolution and liquidation of the Partnership and subject to the provisions of Sections 5.2(c), 5.2(d), 5.3, 5.5 and 8.5, all distributions of cash shall be made: (i) first, 100% to the OP Unitholders in accordance with their respective

Percentage Interests on the Partnership Record Date until (A) the OP Unitholders (other than the General Partner) have received cumulative distributions under this Section 5.2(b) equal to aggregate Capital Contributions made by such OP Unitholders to the Partnership plus a cumulative, noncompounded pre-tax rate of return thereon of 6.0% per annum, determined by taking into account the dates on which all such Capital Contributions and distributions were made and (B) the General Partner has received cumulative distributions under this Section 5.2, equal to (1) the aggregate Capital Contributions made by the General Partner to the Partnership; plus (2) the Asset Acquisition Contributions; plus (3) a cumulative, noncompounded pre-tax rate of return on (1) and (2) of 6.0% per annum, determined by taking into account the dates on which all such Capital Contributions, Asset Acquisition Contributions and distributions were made and (ii) second, (A) 85% to the OP Unitholders, in accordance with their respective Percentage Interests on the Partnership Record Date and (B) 15% to the holder of the Special OP Units, provided however, notwithstanding the foregoing, in the event that the Special OP Unitholder has received a distribution under the circumstances described in Section 8.5(b)(iv) hereof (a "Special OP Unit Distribution") and there is a subsequent Termination Event, no further amount shall be distributed to the Special OP Unitholder until the OP Unitholders have collectively received aggregate distributions equal to the sum of (x) the amount such OP Unitholders are entitled to receive pursuant to this Section 5.2(b) (i) plus (y) an amount equal to 85% of (i) the Special OP Unit Distribution divided by (ii) .15. In applying this Section 5.2(b), the amount distributed per Partnership Unit of any Class may differ from the amount per Partnership Unit of another Class on account of differences in Class-specific expense allocations with respect to REIT Shares as described in the Prospectus (and of corresponding special allocations among Classes of Partnership Units in accordance with Section 5.1(h) hereof) or for other reasons as determined by the board of directors of the General Partner. Any such differences shall correspond to differences in the amount of distributions per REIT Share for REIT Shares of different Classes, with the same adjustments being made to the amount of distributions per Partnership Unit for Partnership Units of a particular Class as are made to the distributions per REIT Share by the General Partner with respect to REIT Shares having the same Class designation.

Notwithstanding any other provision of this Agreement, the General Partner (c) is authorized to take any action that it determines to be necessary or appropriate to cause the Partnership to comply with any withholding requirements established under the Code or any other federal, state or local law including, without limitation, pursuant to Sections 1441, 1442, 1445, 1446, 1471, 1472 and 3406 of the Code. To the extent that the Partnership is required to withhold and pay over to any taxing authority any amount resulting from the allocation or distribution of income to any Partner or assignee (including by reason of Section 1446 of the Code), either (i) if the actual amount to be distributed to the Partner equals or exceeds the amount required to be withheld by the Partnership, the amount withheld shall be treated as a distribution of cash in the amount of such withholding to such Partner, or (ii) if the actual amount to be distributed to the Partner is less than the amount required to be withheld by the Partnership, the actual amount shall be treated as a distribution of cash in the amount of such withholding and the additional amount required to be withheld shall be treated as a loan (a "Partnership Loan") from the Partnership to the Partner on the day the Partnership pays over such amount to a taxing authority. A Partnership Loan shall be repaid through withholding by the Partnership with respect to subsequent distributions to the applicable Partner or assignee. In the event that a Limited Partner (a "Defaulting Limited Partner") fails to pay any amount owed to the Partnership with respect to the Partnership Loan

within fifteen (15) days after demand for payment thereof is made by the Partnership on the Limited Partner, the General Partner, in its sole and absolute discretion, may elect to make the payment to the Partnership on behalf of such Defaulting Limited Partner. In such event, on the date of payment, the General Partner shall be deemed to have extended a loan (a "General Partner Loan") to the Defaulting Limited Partner in the amount of the payment made by the General Partner and shall succeed to all rights and remedies of the Partnership against the Defaulting Limited Partner as to that amount. Without limitation, the General Partner shall have the right to receive any distributions that otherwise would be made by the Partnership to the Defaulting Limited Partner until such time as the General Partner Loan has been paid in full, and any such distributions so received by the General Partner shall be treated as having been received by the Defaulting Limited Partner and immediately paid to the General Partner. Any amounts treated as a Partnership Loan or a General Partner Loan pursuant to this Section 5.2(c) shall bear interest at the lesser of (i) the base rate on corporate loans at large United States money center commercial banks, as published from time to time in The Wall Street Journal, or (ii) the maximum lawful rate of interest on such obligation, such interest to accrue from the date the Partnership or the General Partner, as applicable, is deemed to extend the loan until such loan is repaid in full.

(d) In no event may a Partner receive a distribution of cash with respect to a Partnership Unit if such Partner is entitled to receive a cash distribution as the holder of record of a REIT Share for which all or part of such Partnership Unit has been or will be exchanged.

5.3 <u>**REIT Distribution Requirements**</u>. The General Partner shall use its commercially reasonable efforts to cause the Partnership to distribute amounts sufficient to enable the General Partner to make shareholder distributions that will allow the General Partner to (i) meet its distribution requirement for qualification as a REIT as set forth in Section 857 of the Code and (ii) avoid any federal income or excise tax liability imposed by the Code.

5.4 <u>No Right to Distributions in Kind</u>. No Partner shall be entitled to demand property other than cash in connection with any distributions by the Partnership.

5.5 <u>Limitations on Return of Capital Contributions</u>. Notwithstanding any of the provisions of this <u>Article 5</u>, no Partner shall have the right to receive and the General Partner shall not have the right to make, a distribution that includes a return of all or part of a Partner's Capital Contributions, unless after giving effect to the return of a Capital Contribution, the sum of all Partnership liabilities, other than the liabilities to a Partner for the return of his Capital Contribution, does not exceed the fair market value of the Partnership's assets.

5.6 <u>Asset Acquisition Distributions</u>. Notwithstanding any of the provisions of this Article 5, to the extent the General Partner has made Capital Contributions to the Partnership of the proceeds from the issuance of Additional Securities pursuant to Section 4.2(a)(ii) hereof or the Partnership has borrowed funds or otherwise has funds available for real estate related acquisitions, and it is determined by the General Partner that (i) the General Partner should acquire a General Partner Property and (ii) funds are needed in order for the General Partner to acquire a General Partner Property, then the General Partner shall elect to receive such funds from the Partnership and the Partnership shall pay such funds to the General Partner either as an Asset Acquisition Distribution under this Section 5.6 or as an Asset Acquisition Redemption under Section 8.4.

5.7 Distributions Upon Liquidation. Upon liquidation of the Partnership, after payment of, or adequate provision for, debts and obligations of the Partnership, including any Partner loans, any remaining assets of the Partnership shall be distributed to all Partners in proportion to their respective positive Capital Account balances, determined after taking into account all allocations required to be made pursuant to Section 5.1 hereof and all prior distributions made pursuant to this Article 5, in compliance with Treasury Regulation Section 1.704-1(b)(2)(ii)(b)(2). Notwithstanding any other provision of this Agreement, the amount by which the value, as determined in good faith by the General Partner, of any property other than cash to be distributed in kind to the Partners exceeds or is less than the Carrying Value of such property shall, to the extent not otherwise recognized by the Partnership, be taken into account in computing net profit and net loss of the Partnership (or items thereof) for purposes of crediting or charging the Capital Accounts of, and distributing proceeds to, the Partners, pursuant to this Agreement. To the extent deemed advisable by the General Partner, appropriate arrangements (including the use of a liquidating trust) may be made to assure that adequate funds are available to pay any contingent debts or obligations.

5.8 <u>Substantial Economic Effect</u>. It is the intent of the Partners that the allocations of net profit and net loss (and items thereof) under this Agreement have substantial economic effect (or be consistent with the Partners' interests in the Partnership in the case of the allocation of losses attributable to nonrecourse debt) within the meaning of Section 704(b) of the Code as interpreted by the Regulations promulgated pursuant thereto. Article 5 and other relevant provisions of this Agreement shall be interpreted in a manner consistent with such intent.

ARTICLE 6

RIGHTS, OBLIGATIONS AND POWERS OF THE GENERAL PARTNER

6.1 <u>Management of the Partnership</u>.

(a) Except as otherwise expressly provided in this Agreement, the General Partner shall have full, complete and exclusive discretion to manage and control the business of the Partnership for the purposes herein stated, and shall make all decisions affecting the business and assets of the Partnership. Subject to the restrictions specifically contained in this Agreement, the powers of the General Partner shall include, without limitation, the authority to take the following actions on behalf of the Partnership:

(i) to acquire, purchase, own, operate, lease, dispose and exchange of any Assets, that the General Partner determines are necessary or appropriate or in the best interests of the business of the Partnership;

(ii) to construct buildings and make other improvements on the properties owned or leased by the Partnership;

(iii) to authorize, issue, sell, redeem or otherwise purchase any Partnership Interests or any securities (including secured and unsecured debt obligations of the Partnership, debt obligations of the Partnership convertible into any Class or series of Partnership Interests, or options, rights, warrants or appreciation rights relating to any Partnership Interests) of the Partnership;

(iv) to borrow or lend money for the Partnership or the General Partner or in connection with a General Partner Property, issue or receive evidences of indebtedness in connection therewith, refinance, increase the amount of, modify, amend or change the terms of, or extend the time for the payment of, any such indebtedness, and secure such indebtedness by mortgage, deed of trust, pledge or other lien on the Partnership's assets;

(v) to pay, either directly or by reimbursement, for all operating costs and general administrative expenses of the Partnership to third parties or to the General Partner or its Affiliates as set forth in this Agreement;

(vi) to guarantee or become a co-maker of indebtedness of the General Partner or any Subsidiary thereof, refinance, increase the amount of, modify, amend or change the terms of, or extend the time for the payment of, any such guarantee or indebtedness, and secure such guarantee or indebtedness by mortgage, deed of trust, pledge or other lien on the Partnership's assets;

(vii) to use assets of the Partnership (including, without limitation, cash on hand) for any purpose consistent with this Agreement, including, without limitation, payment, either directly or by reimbursement, of all operating costs and general administrative expenses of the General Partner, the Partnership or any Subsidiary of either, to third parties or to the General Partner as set forth in this Agreement;

(viii) to lease all or any portion of any of the Partnership's assets, whether or not the terms of such leases extend beyond the termination date of the Partnership and whether or not any portion of the Partnership's assets so leased are to be occupied by the lessee, or, in turn, subleased in whole or in part to others, for such consideration and on such terms as the General Partner may determine;

(ix) to prosecute, defend, arbitrate, or compromise any and all claims or liabilities in favor of or against the Partnership, on such terms and in such manner as the General Partner may reasonably determine, and similarly to prosecute, settle or defend litigation with respect to the Partners, the Partnership, or the Partnership's assets;

(x) to file applications, communicate, and otherwise deal with any and all governmental agencies having jurisdiction over, or in any way affecting, the Partnership's assets or any other aspect of the Partnership business;

(xi) to make or revoke any election permitted or required of the Partnership by any taxing authority;

(xii) to maintain such insurance coverage for public liability, fire and casualty, and any and all other insurance for the protection of the Partnership, for the conservation

of Partnership assets, or for any other purpose convenient or beneficial to the Partnership, in such amounts and such types, as it shall determine from time to time;

(xiii) to determine whether or not to apply any insurance proceeds for any property to the restoration of such property or to distribute the same;

(xiv) to establish one or more divisions of the Partnership, to hire and dismiss employees of the Partnership or any division of the Partnership, and to retain legal counsel, accountants, consultants, real estate brokers, and such other persons, as the General Partner may deem necessary or appropriate in connection with the Partnership business and to pay therefor such remuneration as the General Partner may deem reasonable and proper;

(xv) to retain other services of any kind or nature in connection with the Partnership business, and to pay therefor such remuneration as the General Partner may deem reasonable and proper;

(xvi) to negotiate and conclude agreements on behalf of the Partnership with respect to any of the rights, powers and authority conferred upon the General Partner;

(xvii) to maintain accurate accounting records and to file promptly all federal, state and local income tax returns on behalf of the Partnership;

(xviii) to distribute Partnership cash or other Partnership assets in accordance with this Agreement;

(xix) to form or acquire an interest in, and contribute property to, any further limited or general partnerships, joint ventures or other relationships that it deems desirable (including, without limitation, the acquisition of interests in, and the contributions of property to, its Subsidiaries and any other Person in which it has an equity interest from time to time);

(xx) to establish Partnership reserves for working capital, capital expenditures, contingent liabilities, or any other valid Partnership purpose;

Person;

 $(xxi) \quad \text{to merge, consolidate or combine the Partnership with or into another} \\$

(xxii) to do any and all acts and things necessary or prudent to ensure that the Partnership will not be classified as a "publicly traded partnership" for purposes of Section 7704 of the Code; and

(xxiii) to take such other action, execute, acknowledge, swear to or deliver such other documents and instruments, and perform any and all other acts that the General Partner deems necessary or appropriate for the formation, continuation and conduct of the business and affairs of the Partnership (including, without limitation, all actions consistent with allowing the General Partner at all times to qualify as a REIT unless the General Partner voluntarily terminates its REIT status) and to possess and enjoy all of the rights and powers of a general partner as provided by the Act. (b) Except as otherwise provided herein, to the extent the duties of the General Partner require expenditures of funds to be paid to third parties, the General Partner shall not have any obligations hereunder except to the extent that partnership funds are reasonably available to it for the performance of such duties, and nothing herein contained shall be deemed to authorize or require the General Partner, in its capacity as such, to expend its individual funds for payment to third parties or to undertake any individual liability or obligation on behalf of the Partnership.

6.2 <u>Delegation of Authority</u>. The General Partner may delegate any or all of its powers, rights and obligations hereunder, and may appoint, employ, contract or otherwise deal with any Person for the transaction of the business of the Partnership, which Person may, under supervision of the General Partner, perform any acts or services for the Partnership as the General Partner may approve.

6.3 <u>Indemnification and Exculpation of Indemnitees</u>.

(a) The Partnership shall indemnify an Indemnitee from and against any and all losses, claims, damages, liabilities, joint or several, expenses (including reasonable legal fees and expenses), judgments, fines, settlements, and other amounts arising from any and all claims, demands, actions, suits or proceedings, civil, criminal, administrative or investigative, that relate to the operations of the Partnership as set forth in this Agreement in which any Indemnitee may be involved, or is threatened to be involved, as a party or otherwise, unless it is established that: (i) the act or omission of the Indemnitee was material to the matter giving rise to the proceeding and either was committed in bad faith or was the result of active and deliberate dishonesty; (ii) the Indemnitee actually received an improper personal benefit in money, property or services; or (iii) in the case of any criminal proceeding, the Indemnitee had reasonable cause to believe that the act or omission was unlawful. Any indemnification pursuant to this <u>Section 6.3</u> shall be made only out of the assets of the Partnership.

(b) The Partnership shall reimburse an Indemnitee for reasonable expenses incurred by an Indemnitee who is a party to a proceeding in advance of the final disposition of the proceeding upon receipt by the Partnership of (i) a written affirmation by the Indemnitee of the Indemnitee's good faith belief that the standard of conduct necessary for indemnification by the Partnership as authorized in this <u>Section 6.3</u> has been met, and (ii) a written undertaking by or on behalf of the Indemnitee to repay the amount if it shall ultimately be determined that the standard of conduct has not been met.

(c) The indemnification provided by this <u>Section 6.3</u> shall be in addition to any other rights to which an Indemnitee or any other Person may be entitled under any agreement, pursuant to any vote of the Partners, as a matter of law or otherwise, and shall continue as to an Indemnitee who has ceased to serve in such capacity.

(d) The Partnership may purchase and maintain insurance, on behalf of the Indemnitees and such other Persons as the General Partner shall determine, against any liability that may be asserted against or expenses that may be incurred by such Person in connection with the Partnership's activities, regardless of whether the Partnership would have the power to indemnify such Person against such liability under the provisions of this Agreement.

(e) For purposes of this <u>Section 6.3</u>, the Partnership shall be deemed to have requested an Indemnitee to serve as fiduciary of an employee benefit plan whenever the performance by it of its duties to the Partnership also imposes duties on, or otherwise involves services by, it to the plan or participants or beneficiaries of the plan; excise taxes assessed on an Indemnitee with respect to an employee benefit plan pursuant to applicable law shall constitute fines within the meaning of this <u>Section 6.3</u>; and actions taken or omitted by the Indemnitee with respect to an employee benefit plan in the performance of its duties for a purpose reasonably believed by it to be in the interest of the participants and beneficiaries of the plan shall be deemed to be for a purpose which is not opposed to the best interests of the Partnership.

(f) In no event may an Indemnitee subject the Limited Partners to personal liability by reason of the indemnification provisions set forth in this Agreement.

(g) An Indemnitee shall not be denied indemnification in whole or in part under this <u>Section 6.3</u> because the Indemnitee had an interest in the transaction with respect to which the indemnification applies if the transaction was otherwise permitted by the terms of this Agreement.

(h) The provisions of this <u>Section 6.3</u> are for the benefit of the Indemnitees, their heirs, successors, assigns and administrators and shall not be deemed to create any rights for the benefit of any other Persons.

Notwithstanding the foregoing, the Partnership may not indemnify or hold (i) harmless an Indemnitee for any liability or loss unless all of the following conditions are met: (i) the Indemnitee has determined, in good faith, that the course of conduct that caused the loss or liability was in the best interests of the Partnership; (ii) the Indemnitee was acting on behalf of or performing services for the Partnership; (iii) the liability or loss was not the result of (A) negligence or misconduct, in the case that the Indemnitee is a director of the General Partner (other than an Independent Director), the Advisor or an Affiliate of the Advisor or (B) gross negligence or willful misconduct, in the case that the Indemnitee is an Independent Director; and (iv) the indemnification or agreement to hold harmless is recoverable only out of net assets of the Partnership. In addition, the Partnership shall not provide indemnification for any loss, liability or expense arising from or out of an alleged violation of federal or state securities laws by such party unless one or more of the following conditions are met: (i) there has been a successful adjudication on the merits of each count involving alleged material securities law violations as to the Indemnitee; (ii) such claims have been dismissed with prejudice on the merits by a court of competent jurisdiction as to the Indemnitee; or (iii) a court of competent jurisdiction approves a settlement of the claims against the Indemnitee and finds that indemnification of the settlement and the related costs should be made, and the court considering the request for indemnification has been advised of the position of the Securities and Exchange Commission and of the published position of any state securities regulatory authority in which Securities were offered or sold as to indemnification for violations of securities laws.

6.4 <u>Liability of the General Partner</u>.

(a) Notwithstanding anything to the contrary set forth in this Agreement, the General Partner shall not be liable for monetary damages to the Partnership or any Partners for losses sustained or liabilities incurred as a result of errors in judgment or of any act or omission if

the General Partner acted in good faith. The General Partner shall not be in breach of any duty that the General Partner may owe to the Limited Partners or the Partnership or any other Persons under this Agreement or of any duty stated or implied by law or equity provided the General Partner, acting in good faith, abides by the terms of this Agreement.

(b) Each Limited Partner expressly acknowledges and agrees that whenever in this Agreement the General Partner is permitted to take any action, make any decision or determination or otherwise vote on or give its consent to any action, the General Partner shall be entitled to exercise its sole and absolute discretion in connection therewith after considering only such interests and factors as it desires and, without limiting the generality of the foregoing, it is specifically agreed and acknowledged that the General Partner in taking any action or declining to take any action hereunder may consider exclusively its own interests or the interests of its shareholders and shall have no duty or obligation to consider the separate interests of or factors affecting the Partnership or any other Partner (including, without limitation, the tax consequences to Limited Partners or the tax consequences of some, but not all, of the Limited Partners). The General Partner shall not be liable for monetary damages for losses sustained, liabilities incurred, or benefits not derived by Limited Partners in connection with such decisions, provided that the General Partner has acted in good faith.

(c) Subject to its obligations and duties as General Partner set forth in <u>Section 6.1</u> hereof, the General Partner may exercise any of the powers granted to it under this Agreement and perform any of the duties imposed upon it hereunder either directly or by or through its agents. The General Partner shall not be responsible for any misconduct or negligence on the part of any such agent appointed by it in good faith.

(d) Notwithstanding any other provisions of this Agreement or the Act, any action of the General Partner on behalf of the Partnership or any decision of the General Partner to refrain from acting on behalf of the Partnership, undertaken in the good faith belief that such action or omission is necessary or advisable in order (i) to protect the ability of the General Partner to continue to qualify as a REIT or (ii) to prevent the General Partner from incurring any taxes under Section 857, Section 4981, or any other provision of the Code, is expressly authorized under this Agreement and is deemed approved by all of the Limited Partners.

(e) Any amendment, modification or repeal of this <u>Section 6.4</u> or any provision hereof shall be prospective only and shall not in any way affect the limitations on the General Partner's liability to the Partnership and the Limited Partners under this <u>Section 6.4</u> as in effect immediately prior to such amendment, modification or repeal with respect to matters occurring, in whole or in part, prior to such amendment, modification or repeal, regardless of when claims relating to such matters may arise or be asserted.

(f) In accordance with Section 17-1101(d) of the Act, the Partners hereby acknowledge and agree that the provisions of this Agreement, including the provisions of this <u>Article</u> <u>6</u>, to the extent they restrict or eliminate the duties (including fiduciary duties) and liabilities relating thereto otherwise existing at law or in equity, replace completely and absolutely such other duties (including fiduciary duties) and liabilities relating thereto and further acknowledge and agree that

the provisions of this subsection (f) and the other provisions of this <u>Article 6</u> are fundamental elements to the agreement of the Partners to enter into this Agreement.

6.5 <u>Reimbursement of General Partner</u>.

(a) Except as provided in this <u>Section 6.5</u> and elsewhere in this Agreement (including the provisions of <u>Article 5</u> and <u>Article 6</u> regarding distributions, payments, and allocations to which it may be entitled), the General Partner shall not be compensated for its services as general partner of the Partnership.

(b) The General Partner shall be reimbursed on a monthly basis, or such other basis as the General Partner may determine in its sole and absolute discretion, for all Administrative Expenses incurred by the General Partner.

6.6 Outside Activities. Subject to (a) Section 6.8 hereof, (b) the Charter and (c) any agreements entered into by the General Partner or its Affiliates with the Partnership, a Subsidiary or any officer, director, employee, agent, trustee, Affiliate or shareholder of the General Partner, the General Partner shall be entitled to and may have business interests and engage in business activities in addition to those relating to the Partnership, including business interests and activities substantially similar or identical to those of the Partnership. Neither the Partnership nor any of the Limited Partners shall have any rights by virtue of this Agreement in any such business ventures, interests or activities, and the General Partner shall have no obligation pursuant to this Agreement to offer any interest in any such business ventures, interests and activities to the Partnership or any Limited Partner, even if such opportunity is of a character which, if presented to the Partnership or any Limited Partner, could be taken by such Person.

6.7 <u>Employment or Retention of Affiliates</u>.

(a) Any Affiliate of the General Partner may be employed or retained by the Partnership and may otherwise deal with the Partnership (whether as a buyer, lessor, lessee, manager, furnisher of goods or services, broker, agent, lender or otherwise) and may receive from the Partnership any compensation, price, or other payment therefor which the General Partner determines to be fair and reasonable.

(b) The Partnership may lend or contribute to its Subsidiaries or other Persons in which it has an equity investment, and such Persons may borrow funds from the Partnership, on terms and conditions established in the sole and absolute discretion of the General Partner. The foregoing authority shall not create any right or benefit in favor of any Subsidiary or any other Person.

(c) The Partnership may transfer assets to joint ventures, other partnerships, corporations or other business entities in which it is or thereby becomes a participant upon such terms and subject to such conditions as the General Partner deems are consistent with this Agreement, applicable law and the REIT status of the General Partner.

(d) Except as expressly permitted by this Agreement, neither the General Partner nor any of its Affiliates shall sell, transfer or convey any property to, or purchase any property from, the Partnership, directly or indirectly, except pursuant to transactions that are, in the General Partner's sole discretion, on terms that are fair and reasonable to the Partnership.

6.8 <u>General Partner Participation</u>. The General Partner agrees that all business activities of the General Partner, including activities pertaining to the acquisition, development or ownership of any Asset, shall be conducted through the Partnership or one or more Subsidiary Partnerships; provided, however, that the General Partner is allowed to acquire General Partner Property under <u>Section 4.2(a)(ii)</u> hereof.

6.9 <u>Title to Partnership Assets</u>. Title to Partnership assets, whether real, personal or mixed and whether tangible or intangible, shall be deemed to be owned by the Partnership as an entity, and no Partner, individually or collectively, shall have any ownership interest in such Partnership assets or any portion thereof. Title to any or all of the Partnership assets may be held in the name of the Partnership, the General Partner or one or more nominees, as the General Partner may determine, including Affiliates of the General Partner. The General Partner hereby declares and warrants that any Partnership assets for which legal title is held in the name of the General Partner for the use and benefit of the Partnership or one or more Subsidiary Partnerships in accordance with the provisions of this Agreement; provided, however, that the General Partner shall use its commercially reasonable efforts to cause beneficial and record title to such assets to be vested in the Partnership as soon as reasonably practicable. All Partnership assets shall be recorded as the property of the Partnership in its books and records, irrespective of the name in which legal title to such Partnership assets is held.

6.10 <u>Redemption of REIT Shares</u>. In the event the General Partner redeems any REIT Shares, then the General Partner shall cause the Partnership to purchase from the General Partner a number of Partnership Units as determined based on the application of the Conversion Factor for that Class of Partnership Units on the same terms that the General Partner redeemed such REIT Shares. Moreover, if the General Partner makes a cash tender offer or other offer to acquire REIT Shares, then the General Partner shall cause the Partnership to make a corresponding offer to the General Partner to acquire an equal number of Partnership Units held by the General Partner that have the same Class designation as the REIT Shares that are subject to the offer. In the event any REIT Shares are redeemed by the General Partner 's Partnership Units having the same Class designation as the redeemed REIT Shares for an equivalent purchase price based on the application of the Conversion Factor for that Class of Partnership Units.

6.11 No Duplication of Fees or Expenses. The Partnership may not incur or be responsible for any fee or expense (in connection with the Offering or otherwise) that would be duplicative of fees and expenses paid by the General Partner.

ARTICLE 7CHANGES IN GENERAL PARTNER

7.1 <u>Transfer of the General Partner's Partnership Interest</u>.

(a) The General Partner shall not transfer all or any portion of its General Partnership Interest or withdraw as General Partner except as provided in, or in connection with a transaction contemplated by, Sections 7.1(c), 7.1(d) or 7.1(e).

(b) The General Partner agrees that its Percentage Interest will at all times be in the aggregate, at least 0.1%.

(c) Except as otherwise provided in <u>Section 6.4(b)</u> or <u>Section 7.1(d)</u> or <u>7.1(e)</u> hereof, the General Partner shall not engage in any merger, consolidation or other combination with or into another Person or sale of all or substantially all of its assets (other than in connection with a change in the General Partner's state of incorporation or organizational form) in each case which results in a change of control of the General Partner (a "Transaction"), unless:

(i) the consent of Limited Partners holding more than 50% of the Percentage Interests and the consent of the Special OP Unitholder is obtained;

(ii) as a result of such Transaction: (A) all Limited Partners will receive for each Partnership Unit of each Class (other than Special Units) an amount of cash, securities, or other property equal to the product of the Conversion Factor for that Class of Partnership Units and the greatest amount of cash, securities or other property paid in the Transaction to a holder of one REIT Share having the same Class designation as the Partnership Unit in consideration of such REIT Share, provided that if, in connection with the Transaction, a purchase, tender or exchange offer ("Offer") shall have been made to and accepted by the holders of more than 50% of the outstanding REIT Shares, each holder of Partnership Units shall be given the option to exchange its Partnership Units for the greatest amount of cash, securities, or other property which a Limited Partner holding Partnership Units would have received had it (1) exercised its Redemption Right and (2) sold, tendered or exchanged pursuant to the Offer the REIT Shares received upon exercise of the Redemption Right immediately prior to the expiration of the Offer and (B) the Special OP Unitholder will receive for the Special OP Units an amount of cash, securities or other property (as applicable based upon the type of consideration and the proportions thereof paid to holders of REIT Shares in the Transaction) determined as set forth pursuant to Section 5.2(b) or Section 8.5 hereof, as applicable; or

(iii) the General Partner is the surviving entity in the Transaction and either (A) the holders of REIT Shares do not receive cash, securities, or other property in the Transaction or (B) all Limited Partners (other than the General Partner or any Subsidiary) receive (1) in exchange for their Partnership Units of each Class (other than the Special Units), an amount of cash, securities, or other property (expressed as an amount per REIT Share) that is no less than the product of the Conversion Factor for that Class of Partnership Units and the greatest amount of cash, securities, or other property (expressed as an amount per REIT Share) received in the Transaction by any holder of REIT Shares having the same Class designation as the Partnership Units being exchanged, and (2) the Special OP Unitholder receives in exchange for the Special OP Units, an amount of cash, securities or other property (as applicable based upon the type of consideration and the proportions thereof paid to holders of REIT Shares in the Transaction) determined as set forth pursuant to <u>Section 8.5(a)</u> hereof.

Notwithstanding Section 7.1(c), the General Partner may merge with or into (d) or consolidate with another entity if immediately after such merger or consolidation (i) substantially all of the assets of the successor or surviving entity (the "Survivor"), other than Partnership Units held by the General Partner, are contributed, directly or indirectly, to the Partnership as a Capital Contribution in exchange for Partnership Units with a fair market value equal to the value of the assets so contributed as determined by the Survivor in good faith and (ii) the Survivor expressly agrees to assume all obligations of the General Partner, as appropriate, hereunder. Upon such contribution and assumption, the Survivor shall have the right and duty to amend this Agreement as set forth in this Section 7.1(d). The Survivor shall in good faith arrive at a new method for the calculation of the Cash Amount, the REIT Shares Amount and Conversion Factor for a Partnership Unit of each Class after any such merger or consolidation so as to approximate the existing method for such calculation as closely as reasonably possible. Such calculation shall take into account, among other things, the kind and amount of securities, cash and other property that was receivable upon such merger or consolidation by a holder of REIT Shares of each Class or options, warrants or other rights relating thereto, and which a holder of Partnership Units of any Class could have acquired had such Partnership Units been exchanged immediately prior to such merger or consolidation. Such amendment to this Agreement shall provide for adjustment to such method of calculation, which shall be as nearly equivalent as may be practicable to the adjustments provided for with respect to the Conversion Factor for each Class of Partnership Units. The Survivor also shall in good faith modify the definition of REIT Shares and make such amendments to Sections 8.4 and 8.5 hereof so as to approximate the existing rights and obligations set forth in Sections 8.4 and 8.5 as closely as reasonably possible. The above provisions of this Section 7.1 (d) shall similarly apply to successive mergers or consolidations permitted hereunder.

(e) Notwithstanding <u>Section 7.1(c)</u>,

(i) a General Partner may transfer all or any portion of its General Partnership Interest to (A) a wholly-owned Subsidiary of such General Partner or (B) the owner of all of the ownership interests of such General Partner, and following a transfer of all of its General Partnership Interest, may withdraw as General Partner; and

(ii) the General Partner may engage in any transaction that is not required to be submitted to the vote of the holders of the REIT Shares by (A) law or (B) the rules of any national securities exchange on which one or more Classes of REIT Shares are Listed.

7.2 <u>Admission of a Substitute or Additional General Partner</u>. A Person shall be admitted as a substitute or additional General Partner of the Partnership only if the following terms and conditions are satisfied:

(a) the Person to be admitted as a substitute or additional General Partner shall have accepted and agreed to be bound by all the terms and provisions of this Agreement by executing a counterpart thereof and such other documents or instruments as may be required or appropriate in order to effect the admission of such Person as a General Partner, and a certificate evidencing

the admission of such Person as a General Partner shall have been filed for recordation and all other actions required by <u>Section 2.5</u> hereof in connection with such admission shall have been performed;

(b) if the Person to be admitted as a substitute or additional General Partner is a corporation or a partnership it shall have provided the Partnership with evidence satisfactory to counsel for the Partnership of such Person's authority to become a General Partner and to be bound by the terms and provisions of this Agreement; and

(c) counsel for the Partnership shall have rendered an opinion (relying on such opinions from other counsel and the state or any other jurisdiction as may be necessary) that (x) the admission of the person to be admitted as a substitute or additional General Partner is in conformity with the Act and (y) none of the actions taken in connection with the admission of such Person as a substitute or additional General Partner will cause (i) the Partnership to be classified other than as a partnership for federal tax purposes, or (ii) the loss of any Limited Partner's limited liability.

7.3 Effect of Bankruptcy, Withdrawal, Death or Dissolution of a General Partner.

(a) Upon the occurrence of an Event of Bankruptcy as to the sole remaining General Partner (and its removal pursuant to Section 7.4(a) hereof) or the death, withdrawal, deemed removal or dissolution of the sole remaining General Partner (except that, if the sole remaining General Partner is on the date of such occurrence a partnership, the withdrawal, death, dissolution, Event of Bankruptcy as to, or removal of a partner in, such partnership shall be deemed not to be a dissolution of such General Partner if the business of such General Partner is continued by the remaining partner or partners), the Partnership shall be dissolved and terminated unless the Partnership is continued pursuant to Section 7.3(b) hereof. The merger of the General Partner with or into any entity that is admitted as a substitute or successor General Partner pursuant to Section 7.2 hereof shall not be deemed to be the withdrawal, dissolution or removal of the General Partner.

(b) Following the occurrence of an Event of Bankruptcy as to the sole remaining General Partner (and its removal pursuant to Section 7.4(a) hereof) or the death, withdrawal, removal or dissolution of the sole remaining General Partner (except that, if the sole remaining General Partner is, on the date of such occurrence, a partnership, the withdrawal of, death, dissolution, Event of Bankruptcy as to, or removal of a partner in, such partnership shall be deemed not to be a dissolution of such General Partner if the business of such General Partner is continued by the remaining partner or partners), the Limited Partners, within ninety (90) days after such occurrence, may elect to continue the business of the Partnership for the balance of the term specified in Section 2.4 hereof by selecting, subject to Section 7.2 hereof and any other provisions of this Agreement, a substitute General Partner by consent of a majority in interest of the Limited Partners. If the Limited Partners elect to continue the business of the Partnership and admit a substitute General Partner, the relationship with the Partners and of any Person who has acquired an interest of a Partner in the Partnership shall be governed by this Agreement.

7.4 <u>Removal of a General Partner</u>.

(a) Upon the occurrence of an Event of Bankruptcy as to, or the dissolution of, a General Partner, such General Partner shall be deemed to be removed automatically; provided, however, that if a General Partner is on the date of such occurrence a partnership, the withdrawal, death or dissolution of, Event of Bankruptcy as to, or removal of, a partner in, such partnership shall be deemed not to be a dissolution of the General Partner if the business of such General Partner is continued by the remaining partner or partners. The Limited Partners may not remove the General Partner, with or without cause.

If a General Partner has been removed pursuant to this Section 7.4 and the (b) Partnership is continued pursuant to Section 7.3 hereof, such General Partner shall promptly transfer and assign its General Partnership Interest in the Partnership to the substitute General Partner approved by a majority in interest of the Limited Partners in accordance with Section 7.3(b) hereof and otherwise admitted to the Partnership in accordance with Section 7.2 hereof. At the time of assignment, the removed General Partner shall be entitled to receive from the substitute General Partner the fair market value of the General Partnership Interest of such removed General Partner as reduced by any damages caused to the Partnership by such General Partner. Such fair market value shall be determined by an appraiser mutually agreed upon by the General Partner and a majority in interest of the Limited Partners within ten (10) days following the removal of the General Partner. In the event that the parties are unable to agree upon an appraiser, the removed General Partner and a majority in interest of the Limited Partners each shall select an appraiser. Each such appraiser shall complete an appraisal of the fair market value of the removed General Partner's General Partnership Interest within thirty (30) days of the General Partner's removal, and the fair market value of the removed General Partner's General Partnership Interest shall be the average of the two appraisals; provided, however, that if the higher appraisal exceeds the lower appraisal by more than 20% of the amount of the lower appraisal, the two appraisers, no later than forty (40) days after the removal of the General Partner, shall select a third appraiser who shall complete an appraisal of the fair market value of the removed General Partner's General Partnership Interest no later than sixty (60) days after the removal of the General Partner. In such case, the fair market value of the removed General Partner's General Partnership Interest shall be the average of the two appraisals closest in value.

(c) The General Partnership Interest of a removed General Partner, until transfer under Section 7.4(b), shall be converted to that of a special Limited Partner; provided, however, such removed General Partner shall not have any rights to participate in the management and affairs of the Partnership, and shall not be entitled to any portion of the income, expense, profit, gain or loss allocations or cash distributions allocable or payable, as the case may be, to the Limited Partners. Instead, such removed General Partner shall receive and be entitled only to retain distributions or allocations of such items that it would have been entitled to receive in its capacity as General Partner, until the transfer is effective pursuant to Section 7.4(b).

(d) All Partners shall have given and hereby do give such consents, shall take such actions and shall execute such documents as shall be legally necessary, desirable and sufficient to effect all the foregoing provisions of this Section.

ARTICLE 8

RIGHTS AND OBLIGATIONS OF THE LIMITED PARTNERS

8.1 <u>Management of the Partnership</u>. The Limited Partners shall not participate in the management or control of Partnership business nor shall they transact any business for the Partnership, nor shall they have the power to sign for or bind the Partnership, such powers being vested solely and exclusively in the General Partner.

8.2 <u>Power of Attorney</u>. Each Limited Partner hereby irrevocably appoints the General Partner its true and lawful attorney-in-fact, who may act for each Limited Partner and in its name, place and stead, and for its use and benefit, to sign, acknowledge, swear to, deliver, file or record, at the appropriate public offices, any and all documents, certificates, and instruments as may be deemed necessary or desirable by the General Partner to carry out fully the provisions of this Agreement and the Act in accordance with their terms, which power of attorney is coupled with an interest and shall survive the death, dissolution or legal incapacity of the Limited Partner, or the transfer by the Limited Partner of any part or all of its Partnership Interest.

8.3 <u>Limitation on Liability of Limited Partners</u>. No Limited Partner shall be liable for any debts, liabilities, contracts or obligations of the Partnership. A Limited Partner shall be liable to the Partnership only to make payments of its Capital Contribution, if any, as and when due hereunder. After its Capital Contribution is fully paid, no Limited Partner shall, except as otherwise required by the Act, be required to make any further Capital Contributions or other payments or lend any funds to the Partnership.

8.4 <u>Redemption Right</u>.

(a) Subject to <u>Sections 8.4(b)</u>, <u>8.4(c)</u>, <u>8.4(d)</u>, <u>8.4(e)</u>, <u>8.4(f)</u> and <u>8.5</u> hereof, the provisions of any agreements between the Partnership and one or more Limited Partners with respect to Partnership Units held by them, each Limited Partner, other than the General Partner (except as permitted below), shall, after holding its Partnership Units for at least one year (other than the Advisor and its Affiliates), have the right (subject to the terms and conditions set forth herein) to require the Partnership to redeem (a "Redemption") all or a portion of the Partnership Units (other than Special OP Units), held by such Limited Partner (such Units, the "Tendered Units"), in exchange (a "Redemption Right"), alternatively, for either REIT Shares having the same Class designation as the Partnership Units subject to the Redemption Right or the Cash Amount, as determined by the General Partner in its sole discretion. The consideration payable in respect of Tendered Units shall be issued or paid, as the case may be, on the Specified Redemption Date. Any Redemption Right shall be exercised pursuant to a Notice of Redemption delivered to the Partnership (with a copy to the General Partner) by the Limited Partner exercising the Redemption Right (the "Tendering Party"). A Limited Partner may not exercise the Redemption Right for less than 1,000 Partnership Units or, if such Limited Partner holds less than 1,000 Partnership Units, all of the Partnership Units held by such Partner. The Tendering Party shall have no right, with respect to any Partnership Units so redeemed, to receive any distribution paid with respect to Partnership Units if the record date for such distribution is on or after the Specified Redemption Date. Notwithstanding the foregoing, the General Partner will be entitled to have its Partnership Units redeemed for the Cash Amount (an "Asset Acquisition Redemption"), at any time and under the circumstances described in <u>Section</u> <u>5.6</u> hereof.

(b) If the General Partner elects to cause the Tendered Units to be exchanged for REIT Shares having the same Class designation as the Tendered Units rather than the Cash Amount, then the Partnership shall direct the General Partner to issue and deliver such REIT Shares to the Tendering Party pursuant to the terms set forth in this Section 8.4(b), in which case, (i) the General Partner, acting as a distinct legal entity, shall assume directly the Partnership's redemption obligation with respect thereto and shall satisfy the Tendering Party's exercise of its Redemption Right, and (ii) such transaction shall be treated, for federal income tax purposes, as a transfer by the Tendering Party of such Tendered Units to the General Partner in exchange for REIT Shares. The percentage of the Tendered Units which are to be so exchanged for REIT Shares (rather than the Cash Amount) is referred to as the "Applicable Percentage." In making such election to exchange Tendered Units for cash or REIT Shares, the General Partner shall act in a fair, equitable and reasonable manner that neither prefers one group or class of Limited Partners over another nor discriminates against a group or class of Limited Partners. If the General Partner determines to redeem any Tendered Units for REIT Shares, rather than the Cash Amount, on the Specified Redemption Date, the Tendering Party shall sell such number of the Tendered Units to the General Partner in exchange for a number of REIT Shares having the same Class designation as the Tendered Units equal to the product of the REIT Shares Amount and the Applicable Percentage. Such amount of REIT Shares having the same Class designation as the Tendered Units shall be delivered by the General Partner as duly authorized, validly issued, fully paid and nonassessable REIT Shares, free of any pledge, lien, encumbrance or restriction, other than the Ownership Limit (as calculated in accordance with the Charter) and other restrictions provided in the Articles of Incorporation, the bylaws of the General Partner, the Securities Act and relevant state securities or "blue sky" laws. Notwithstanding the provisions of Section 8.4(a) and this Section 8.4(b), the Tendering Parties shall have no rights under this Agreement that would otherwise be prohibited under the Charter.

(c) In connection with an exercise of Redemption Rights pursuant to this <u>Section 8.4</u>, the Tendering Party shall submit the following to the General Partner, in addition to the Notice of Redemption:

(i) A written affidavit, dated the same date as the Notice of Redemption, (a) disclosing the actual and constructive ownership, as determined for purposes of Code Sections 856(a)(6) and 856(h), of REIT Shares by (i) such Tendering Party and (ii) any Related Party and (b) representing that, after giving effect to the Redemption, neither the Tendering Party nor any Related Party will own REIT Shares in excess of the Ownership Limit (or, if applicable the Excepted Holder Limit);

(ii) A written representation that neither the Tendering Party nor any Related Party has any intention to acquire any additional REIT Shares prior to the closing of the Redemption on the Specified Redemption Date; and

(iii) An undertaking to certify, at and as a condition to the closing of the Redemption on the Specified Redemption Date, that either (a) the actual and constructive ownership of REIT Shares by the Tendering Party and any Related Party remain unchanged from that disclosed

in the affidavit required by <u>Section 8.4(c)(i)</u> or (b) after giving effect to the Redemption, neither the Tendering Party nor any Related Party shall own REIT Shares in violation of the Ownership Limit (or, if applicable, the Excepted Holder Limit).

(iv) Any other documents as the General Partner may reasonably require in connection with the issuance of REIT Shares upon the exercise of the Redemption Right.

(d) Any Cash Amount to be paid to a Tendering Party pursuant to this <u>Section 8.4</u> shall be paid on the Specified Redemption Date; provided, however, that the General Partner may elect to cause the Specified Redemption Date to be delayed for up to an additional 180 days to the extent required for the General Partner to provide financing to be used to make such payment of the Cash Amount, by causing the issuance of additional REIT Shares or otherwise. Notwithstanding the foregoing, the General Partner agrees to use its commercially reasonable efforts to cause the closing of the acquisition of Tendered Units hereunder to occur as quickly as reasonably possible.

(e) Notwithstanding any other provision of this Agreement, the General Partner shall place appropriate restrictions on the ability of the Limited Partners to exercise their Redemption Rights to prevent, among other things, (i) any person from owning shares in excess of the Ownership Limit and the Excepted Holder Limit, (ii) the General Partner's common stock from being owned by less than 100 persons, (iii) the General Partner from being "closely held" within the meaning of Section 856(h) of the Code, (iv) violations or what would be likely to constitute a violation of any applicable federal or state securities law, (v) violations of any provision of the General Partner's Charter or Bylaws and (vi) as and if deemed necessary to ensure that the Partnership does not constitute a "publicly traded partnership" under Section 7704 of the Code. If and when the General Partner determines that imposing such restrictions is necessary, the General Partner shall give prompt written notice thereof (a "Restriction Notice") to each of the Limited Partners holding Partnership Units, which notice shall be accompanied by a copy of an opinion of counsel to the Partnership which states that, in the opinion of such counsel, restrictions are necessary in order to avoid having the Partnership be treated as a "publicly traded partnership" under Section 7704 of the Code.

(f) A redemption fee may be charged in connection with an exercise of Redemption Rights pursuant to this <u>Section 8.4</u>.

8.5 <u>Redemption or Conversion of Special OP Units and Partnership Units owned</u> by the Advisor or its Affiliates.

(a) <u>Termination Events</u>. In connection with: (i) a Listing, (ii) a merger, consolidation or sale of substantially all of the Partnership's assets, a purchase, tender or exchange offer accepted by the holders of more than 50% of the outstanding REIT Shares, or any similar transaction, (iii) any transaction pursuant to which a majority of the Directors then in office are replaced or removed which is not otherwise described in (ii) above, or (iv) the termination or nonrenewal of the Advisory Agreement for any reason other than by the Advisor and other than in connection with (i), (ii) or (iii) above (the events described in (i) through (iv) are hereinafter referred to individually as a "Termination Event" and collectively as the "Termination Events"), then at the election of the Special OP Unitholder, and as further provided in <u>Section 8.5(b)</u> below, such holder

may (1) exchange its Special OP Units for Class A Units, (2) exchange its Special OP Units for Class A Units and immediately thereafter redeem its Class A Units received in such exchange pursuant to <u>Section 8.5(b)</u> hereof, or (3) retain its Special OP Units.

(b) <u>Special OP Unit Exchange; Redemption of Partnership Units of the Special</u> <u>OP Unitholder, the Advisor or its Affiliates; Entitlement to Distributions</u>.

(i) If the Special OP Unitholder elects to exchange its Special OP Units for Class A Units in connection with a Termination Event, then the Special OP Units shall be exchanged for a number of Class A Units (the "Special OP Unit Value") equal in value to the aggregate amount of distributions that would have been made with respect to the Special OP Units under Section 5.2(b)(ii) if all assets (subject to their liabilities) of the Partnership were sold for their fair market value, as determined in good faith by the Directors of the General Partner in office prior to the Termination Event and in the manner set forth below, any remaining liabilities of the Partnership were satisfied in full in cash according to their terms and assuming for these purposes that all such remaining liabilities had matured, and Net Sales Proceeds (after satisfaction of such liabilities) were distributed in full pursuant to Section 5.2(b). The Special OP Unit Value shall be determined (1) in connection with a Termination Event described in Section 8.5(a)(i) above, by reference to the market value of the Class A REIT Shares based upon the average closing price, or average of bid and asked prices (if closing prices are not available) during a period of thirty (30) days during which such shares are traded beginning 90 days after the Listing; (2) in connection with a Termination Event described in Section 8.5(a)(ii) above, by reference to the value of the consideration received or to be received by the holders of Class A REIT Shares and the implied value of the assets and liabilities of the General Partner and the Partnership as a result thereof, or (3) in connection with a Termination Event described in Sections 8.5(a)(iii) or 8.5(a)(iv) above, by reference to the fair market value of the assets and liabilities of the General Partner and the Partnership as determined by an independent third party mutually agreed to by the Partnership on the one hand and the holder of the applicable Special OP Units on the other. The valuation mechanisms referred to in the immediately preceding clauses (1) through (3) are hereinafter referred to as the "Valuation Mechanisms". If multiple Termination Events are triggered in connection with a series of related events, the Special OP Unitholder, in its sole discretion, may determine which Valuation Mechanism should be used to value the Class A Units and determine the Special OP Unit Value. In the case of any Termination Event other than a Listing, the exchange of the Special OP Units for Class A Units shall occur simultaneously with the occurrence or consummation of such Termination Event or as soon as is reasonably practicable thereafter, and in the case of a Listing, the exchange of the Special OP Units for Class A Units shall occur within 125 days after the Listing shall have occurred.

(ii) If the Special OP Unitholder elects to receive Class A Units in exchange for its Special OP Units in connection with the Termination Event but not to have such Class A Units redeemed, then such Class A Units shall thereafter be subject to all of the applicable provisions of this Agreement, including Section 8.5(d) hereof. If the Special OP Unitholder elects to immediately redeem its Class A Units, then the Special OP Unitholder shall receive, at its option, upon redemption of such Partnership Units, cash (or in the case of a Termination Event described in Section 8.5(a)(iv) hereof, a non-interest bearing promissory note) or Class A REIT Shares with

an aggregate value equal to the Special OP Unit Value as determined under subsection (i) of this <u>Section 8.5(b)</u>. Notwithstanding anything to the contrary above, in the case of any Termination Event described in <u>Section 8.5(a)(iv)</u> hereof, the holder may elect to receive, at its option, upon redemption of such Class A Units, a non-interest bearing promissory note or Class A REIT Shares. If the holder elects payment in the form of a non-interest bearing promissory note, such non-interest bearing promissory note shall be payable in 12 equal quarterly installments, provided, however, that no payment will be made in any quarter in which such payment would impair the General Partner's capital or jeopardize the General Partner's REIT status, in which case any such payment or payments shall be delayed until the next quarter in which payment would not impair the General Partner's REIT status.

(iii) Notwithstanding anything to the contrary contained in Section 8.4 hereof, if in connection with a Termination Event the Advisor and any of its Affiliates hold Class A Units that were not received in connection with such Termination Event, such holder or holders may elect, in the holder's sole discretion, to have such Class A Units valued in the manner set forth in Section 8.5(b)(i) and, using the applicable Valuation Mechanism, redeemed for the resulting amount of consideration. Such consideration shall be payable in the same form and at the same time that any consideration that would be payable upon a redemption described in subparagraph (i) or (ii) of this Section 8.5(b) would be due.

(iv) In connection with a Termination Event, the Special OP Unitholder may elect not to have the Special OP Units exchanged for Class A Units. In such event, the Special OP Unitholder shall receive a cash distribution or a non-interest bearing promissory note equal to the aggregate amount that the Special OP Unitholder would have been entitled to receive under subparagraph (ii) of this <u>Section 8.5(b)</u> above if the Special OP Unitholder had elected to exchange the Special OP Units into Class A Units which were thereafter immediately redeemed; provided, however that the Special OP Units shall not, under such circumstances, be exchanged for Class A Units and shall instead remain outstanding and subject to all of the applicable provisions of this Agreement.

(v) If Class A REIT Shares are to be issued in connection with the redemption of Class A Units pursuant to this <u>Section 8.5(b)</u>, then the General Partner shall issue such Class A REIT Shares in accordance herewith and the exchange of Class A Units shall be treated in accordance with <u>Section 8.4(b)</u> as if the Class A Units were Tendered Units. All cash payments required to be made pursuant to this <u>Section 8.5</u> shall be made by wire transfer of immediately available funds to an account designated by the recipient of such payment.

(c) <u>Limitation on Exchange and Redemption</u>. Notwithstanding anything herein to the contrary, no exchange or redemption pursuant to <u>Section 8.5(b)</u> shall be permitted unless and until OP Unitholders have received (or are deemed to have received pursuant to the deemed valuations set forth in such sections) aggregate, cumulative distributions from the Partnership to OP Unitholders for all years from operating income, sales proceeds and other sources in an amount equal to (i) the sum of the aggregate capital contributions to the Partnership by the OP Unitholders for all years plus (ii) an 6.0% cumulative non-compounded annual pre-tax return on the amount described in the immediately preceding subclause (i).

(d) <u>Redemption of Partnership Units Following a Termination Event</u>. If the Advisor or any of its Affiliates retains any of their Class A Units following a Termination Event, the Advisor and its Affiliates shall have the right to redeem such Partnership Units pursuant to all of the terms and conditions of <u>Section 8.4</u> hereof; provided, however, that the holder of such Class A Units and not the General Partner shall be entitled to elect cash or Class A REIT Shares.

ARTICLE 9

TRANSFERS OF LIMITED PARTNERSHIP INTERESTS

9.1 <u>Purchase for Investment</u>.

(a) Each Limited Partner hereby represents and warrants to the General Partner and to the Partnership that the acquisition of his Partnership Interest is made as a principal for his account for investment purposes only and not with a view to the resale or distribution of such Partnership Interest.

(b) Each Limited Partner agrees that he will not sell, assign or otherwise transfer his Partnership Interest or any fraction thereof, whether voluntarily or by operation of law or at judicial sale or otherwise, to any Person who does not make the representations and warranties to the General Partner set forth in Section 9.1(a) above and similarly agree not to sell, assign or transfer such Partnership Interest or fraction thereof to any Person who does not similarly represent, warrant and agree.

9.2 <u>Restrictions on Transfer of Limited Partnership Interests</u>.

(a) Subject to the provisions of <u>Section 9.2(b)</u> and <u>9.2(c)</u>, no Limited Partner may offer, sell, assign, hypothecate, pledge or otherwise transfer all or any portion of his Limited Partnership Interest, or any of such Limited Partner's economic rights as a Limited Partner, whether voluntarily or by operation of law or at judicial sale or otherwise (collectively, a "Transfer") without the prior consent of the General Partner, which consent may be granted or withheld in its sole and absolute discretion. Any such purported transfer undertaken without such consent shall be considered to be null and void ab initio and shall not be given effect. The General Partner may require, as a condition of any Transfer to which it consents, that the transferor assume all costs incurred by the Partnership in connection therewith.

(b) No Limited Partner may withdraw from the Partnership other than: (i) as a result of a permitted Transfer (i.e., a Transfer consented to as contemplated by clause (a) above or clause (c) below or a Transfer pursuant to <u>Section 9.5</u> below) of all of its Partnership Interest pursuant to this <u>Article 9</u>, (ii) pursuant to a redemption of all of its Partnership Units pursuant to <u>Section 8.4</u> or (iii) pursuant to the redemption of the Limited Partner's Special OP Units and Partnership Units pursuant to <u>Section 8.5</u>. Upon the permitted Transfer or redemption of all of a Limited Partner's Partnership Units and Special OP Units, if any, such Limited Partner shall cease to be a Limited Partner.

(c) Notwithstanding Section 9.2(a) and subject to Sections 9.2(d), 9.2(e) and 9.2 (f) below, a Limited Partner may not Transfer, without the prior consent of the General Partner, which consent will not be unreasonably withheld, all or a portion of its Partnership Interest to (i) a parent or parent's spouse, natural or adopted descendant or descendants, spouse of such descendant, or brother or sister, or a trust created by such Limited Partner for the benefit of such Limited Partner and/or any such person(s), of which trust such Limited Partner or any such person(s) is a trustee, (ii) a corporation controlled by a Person or Persons named in (i) above, or (iii) if the Limited Partner is an entity, its beneficial owners.

(d) No Limited Partner may effect a Transfer of its Limited Partnership Interest, in whole or in part, if, in the opinion of legal counsel for the Partnership, such proposed Transfer would require the registration of the Limited Partnership Interest under the Securities Act or would otherwise violate any applicable federal or state securities or blue sky law (including investment suitability standards).

(e) No Transfer by a Limited Partner of its Partnership Interest, in whole or in part, may be made to any Person if (i) in the opinion of legal counsel for the Partnership, the transfer would result in the Partnership's being treated as an association taxable as a corporation (other than a qualified REIT subsidiary within the meaning of Section 856(i) of the Code), (ii) in the opinion of legal counsel for the Partnership, it would adversely affect the ability of the General Partner to continue to qualify as a REIT or subject the General Partner to any additional taxes under Section 857 or Section 4981 of the Code, or (iii) such transfer is effectuated through an "established securities market" or a "secondary market (or the substantial equivalent thereof)" within the meaning of Section 7704 of the Code.

(f) No Transfer by a Limited Partner of any Partnership Interest may be made to a lender to the Partnership or any Person who is related (within the meaning of Regulations Section 1.752-4(b)) to any lender to the Partnership whose loan constitutes a nonrecourse liability (within the meaning of Regulations Section 1.752-1(a)(2)), without the consent of the General Partner, which may be withheld in its sole and absolute discretion, provided that as a condition to such consent the lender will be required to enter into an arrangement with the Partnership and the General Partner to exchange or redeem for the Cash Amount any Partnership Units in which a security interest is held simultaneously with the time at which such lender would be deemed to be a Partner in the Partnership for purposes of allocating liabilities to such lender under Section 752 of the Code.

(g) Any Transfer in contravention of any of the provisions of this <u>Article 9</u> shall be void and ineffectual and shall not be binding upon, or recognized by, the Partnership.

(h) Prior to the consummation of any Transfer under this <u>Article 9</u>, the transferor and/or the transferee shall deliver to the General Partner such opinions, certificates and other documents as the General Partner shall request in connection with such Transfer.

9.3 Admission of Substitute Limited Partner.

(a) Subject to the other provisions of this <u>Article 9</u>, an assignee of the Limited Partnership Interest of a Limited Partner (which shall be understood to include any purchaser, transferee, donee, or other recipient of any disposition of such Limited Partnership Interest) shall be deemed admitted as a Limited Partner of the Partnership only with the consent of the General Partner, which consent may be granted or withheld in its sole and absolute discretion, and upon the satisfactory completion of the following:

(i) The assignee shall have accepted and agreed to be bound by the terms and provisions of this Agreement by executing a counterpart or an amendment thereof, including a revised <u>Exhibit A</u>, and such other documents or instruments as the General Partner may require in order to effect the admission of such Person as a Limited Partner.

(ii) To the extent required, an amended Certificate evidencing the admission of such Person as a Limited Partner shall have been signed, acknowledged and filed for record in accordance with the Act.

(iii) The assignee shall have delivered a letter containing the representation set forth in Section 9.1(a) hereof and the agreement set forth in Section 9.1(b) hereof.

(iv) If the assignee is a corporation, partnership or trust, the assignee shall have provided the General Partner with evidence satisfactory to counsel for the Partnership of the assignee's authority to become a Limited Partner under the terms and provisions of this Agreement.

(v) The assignee shall have executed a power of attorney containing the terms and provisions set forth in <u>Section 8.2</u> hereof.

(vi) The assignee shall have paid all legal fees and other expenses of the Partnership and the General Partner and filing and publication costs in connection with its substitution as a Limited Partner.

(vii) The assignee has obtained the prior written consent of the General Partner to its admission as a Substitute Limited Partner, which consent may be given or denied in the exercise of the General Partner's sole and absolute discretion.

(b) For the purpose of allocating profits and losses and distributing cash received by the Partnership, a Substitute Limited Partner shall be treated as having become, and appearing in the records of the Partnership as, a Partner upon the filing of the Certificate described in <u>Section 9.3</u> (a)(ii) hereof or, if no such filing is required, the later of the date specified in the transfer documents or the date on which the General Partner has received all necessary instruments of transfer and substitution.

(c) The General Partner shall cooperate with the Person seeking to become a Substitute Limited Partner by preparing the documentation required by this Section and making all official filings and publications. The Partnership shall take all such action as promptly as practicable

after the satisfaction of the conditions in this <u>Article 9</u> to the admission of such Person as a Limited Partner of the Partnership.

9.4 <u>Rights of Assignees of Partnership Interests</u>.

(a) Subject to the provisions of <u>Sections 9.1</u> and <u>9.2</u> hereof, except as required by operation of law, the Partnership shall not be obligated for any purposes whatsoever to recognize the assignment by any Limited Partner of its Partnership Interest until the Partnership has received notice thereof.

(b) Any Person who is the assignee of all or any portion of a Limited Partner's Limited Partnership Interest, but does not become a Substitute Limited Partner and desires to make a further assignment of such Limited Partnership Interest, shall be subject to all the provisions of this <u>Article 9</u> to the same extent and in the same manner as any Limited Partner desiring to make an assignment of its Limited Partnership Interest.

9.5 Effect of Bankruptcy, Death, Incompetence or Termination of a Limited Partner. The occurrence of an Event of Bankruptcy as to a Limited Partner, the death of a Limited Partner or a final adjudication that a Limited Partner is incompetent (which term shall include, but not be limited to, insanity) shall not cause the termination or dissolution of the Partnership, and the business of the Partnership shall continue if an order for relief in a bankruptcy proceeding is entered against a Limited Partner, the trustee or receiver of his estate or, if he dies, his executor, administrator or trustee, or, if he is finally adjudicated incompetent, his committee, guardian or conservator, shall have the rights of such Limited Partner for the purpose of settling or managing his estate property and such power as the bankrupt, deceased or incompetent Limited Partner possessed to assign all or any part of his Partnership Interest and to join with the assignee in satisfying conditions precedent to the admission of the assignee as a Substitute Limited Partner.

9.6 Joint Ownership of Interests. A Partnership Interest may be acquired by two individuals as joint tenants with right of survivorship, provided that such individuals either are married or are related and share the same home as tenants in common. The written consent or vote of both owners of any such jointly held Partnership Interest shall be required to constitute the action of the owners of such Partnership Interest; provided, however, that the written consent of only one joint owner will be required if the Partnership has been provided with evidence satisfactory to the counsel for the Partnership that the actions of a single joint owner can bind both owners under the applicable laws of the state of residence of such joint owners. Upon the death of one owner of a Partnership Interest held in a joint tenancy with a right of survivorship, the Partnership Interest shall become owned solely by the survivor as a Limited Partner and not as an assignee. The Partnership need not recognize the death of one of the owners of a jointly-held Partnership Interest until it shall have received notice of such death. Upon notice to the General Partner from either owner, the General Partner shall cause the Partnership Interest to be divided into two equal Partnership Interests, which shall thereafter be owned separately by each of the former owners.

ARTICLE 10

BOOKS AND RECORDS; ACCOUNTING; TAX MATTERS

10.1 <u>Books and Records</u>. At all times during the continuance of the Partnership, the Partners shall keep or cause to be kept at the Partnership's specified office true and complete books of account in accordance with generally accepted accounting principles, including: (a) a current list of the full name and last known business address of each Partner, (b) a copy of the Certificate of Limited Partnership and all Certificates of amendment thereto, (c) copies of the Partnership's federal, state and local income tax returns and reports, (d) copies of this Agreement and amendments thereto and any financial statements of the Partnership for the three most recent years and (e) all documents and information required under the Act. Any Partner or its duly authorized representative, upon paying the costs of collection, duplication and mailing, shall be entitled to inspect or copy such records during ordinary business hours.

10.2 Custody of Partnership Funds; Bank Accounts.

(a) All funds of the Partnership not otherwise invested shall be deposited in one or more accounts maintained in such banking or brokerage institutions as the General Partner shall determine, and withdrawals shall be made only on such signature or signatures as the General Partner may, from time to time, determine

(b) All deposits and other funds not needed in the operation of the business of the Partnership may be invested by the General Partner in investment grade instruments (or investment companies whose portfolio consists primarily thereof), government obligations, certificates of deposit, bankers' acceptances and municipal notes and bonds. The funds of the Partnership shall not be commingled with the funds of any other Person except for such commingling as may necessarily result from an investment in those investment companies permitted by this <u>Section 10.2(b)</u>.

10.3 <u>**Fiscal and Taxable Year**</u>. The fiscal and taxable year of the Partnership shall be the calendar year.

10.4 <u>Annual Tax Information and Report</u>. Within seventy-five (75) days after the end of each fiscal year of the Partnership, the General Partner shall furnish to each person who was a Limited Partner at any time during such year the tax information necessary to file such Limited Partner's individual tax returns as shall be reasonably required by law.

10.5 <u>Tax Matters Partner; Tax Elections; Special Basis Adjustments</u>.

(a) The General Partner shall be the Tax Matters Partner of the Partnership within the meaning of Section 6231(a)(7) of the Code. As Tax Matters Partner, the General Partner shall have the right and obligation to take all actions authorized and required, respectively, by the Code for the Tax Matters Partner. The General Partner shall have the right to retain professional assistance in respect of any audit of the Partnership by the Service and all out-of-pocket expenses and fees incurred by the General Partner on behalf of the Partnership as Tax Matters Partner shall constitute Partnership expenses. In the event the General Partner receives notice of a final Partnership adjustment under Section 6223(a)(2) of the Code, the General Partner shall either (i) file a court petition for judicial review of such final adjustment within the period provided under Section 6226(a) of the Code, a copy of which petition shall be mailed to all Limited Partners on the date such petition is filed, or (ii) mail a written notice to all Limited Partners, within such period, that describes the General Partner's reasons for determining not to file such a petition.

(b) All elections required or permitted to be made by the Partnership under the Code or any applicable state or local tax law shall be made by the General Partner in its sole and absolute discretion.

(c) In the event of a transfer of all or any part of the Partnership Interest of any Partner, the Partnership, at the option of the General Partner, may elect pursuant to Section 754 of the Code to adjust the basis of the Partnership's assets. Notwithstanding anything contained in <u>Article 5</u> of this Agreement, any adjustments made pursuant to Section 754 of the Code shall affect only the successor in interest to the transferring Partner and in no event shall be taken into account in establishing, maintaining or computing Capital Accounts for the other Partners for any purpose under this Agreement. Each Partner will furnish the Partnership with all information necessary to give effect to such election.

10.6 <u>Reports to Limited Partners</u>.

(a) As soon as practicable after the close of each fiscal quarter (other than the last quarter of the fiscal year), the General Partner shall cause to be mailed to each Limited Partner a quarterly report containing financial statements of the Partnership, or of the General Partner if such statements are prepared solely on a consolidated basis with the General Partner, for such fiscal quarter, presented in accordance with generally accepted accounting principles. As soon as practicable after the close of each fiscal year, the General Partner shall cause to be mailed to each Limited Partner an annual report containing financial statements of the Partnership, or of the General Partner, for such fiscal year, presented in accordance with generally accepted accounting principles. The annual financial statements shall be audited by accountants selected by the General Partner.

(b) Any Partner shall further have the right to a private audit of the books and records of the Partnership at the expense of such Partner, provided such audit is made for Partnership purposes and is made during normal business hours.

10.7 <u>Safe Harbor Election</u>. The Partners agree that, in the event the Safe Harbor Regulation is finalized, the Partnership shall be authorized and directed to make the Safe Harbor Election and the Partnership and each Partner (including any person to whom an interest in the Partnership is transferred in connection with the performance of services) agrees to comply with all requirements of the Safe Harbor with respect to all interests in the Partnership transferred in connection with the performance of services while the Safe Harbor Election remains effective. The Tax Matters Partner shall be authorized to (and shall) prepare, execute, and file the Safe Harbor Election.

ARTICLE 11

AMENDMENT OF AGREEMENT; MERGER

The General Partner's consent shall be required for any amendment to this Agreement. The General Partner, without the consent of the Limited Partners, may amend this Agreement in any respect or merge or consolidate the Partnership with or into any other partnership or business entity (as defined in Section 17-211 of the Act) in a transaction pursuant to Section 7.1(c)(ii) or 7.1(c) (iii), 7.1(d) or 7.1(e) hereof; provided, however, that the following amendments and any other merger or consolidation of the Partnership shall require the consent of Limited Partners holding more than 67% of the Percentage Interests of the Limited Partners and the Special OP Unitholder:

(a) any amendment affecting the operation of the Conversion Factor or the Redemption Right (except as provided in Section 7.4(d) or 7.1(d) hereof) in a manner adverse to the Limited Partners;

(b) any amendment that would adversely affect the rights of the Limited Partners to receive the distributions payable to them hereunder, other than with respect to the issuance of additional Partnership Units pursuant to <u>Section 4.2</u> hereof; or

(c) any amendment that would alter the Partnership's allocations of profit and loss to the Limited Partners, other than with respect to the issuance of additional Partnership Units pursuant to <u>Section 4.2</u> hereof; and any amendment that would impose on any Limited Partner any obligation to make additional Capital Contributions to the Partnership or otherwise alter such Limited Partner's right to receive distributions of cash or other property or allocations of items of income, gain, deduction loss or credit shall require the written consent of both the General Partner and any such Limited Partner. In addition, any amendment to <u>Section 8.5</u> shall require the consent of the Special OP Unitholder, and any amendment to this <u>Article 11</u> shall require the written consent of all Partners.

ARTICLE 12

GENERAL PROVISIONS

12.1 <u>Notices</u>. All communications required or permitted under this Agreement shall be in writing and shall be deemed to have been given when delivered personally or upon deposit in the United States mail, registered, postage prepaid return receipt requested, to the Partners at the addresses set forth in <u>Exhibit A</u> attached hereto; provided, however, that any Partner may specify a different address by notifying the General Partner in writing of such different address. Notices to the Partnership shall be delivered at or mailed to its specified office.

12.2 <u>Survival of Rights</u>. Subject to the provisions hereof limiting transfers, this Agreement shall be binding upon and inure to the benefit of the Partners and the Partnership and their respective legal representatives, successors, transferees and assigns.

12.3 <u>Additional Documents</u>. Each Partner agrees to perform all further acts and execute, swear to, acknowledge and deliver all further documents which may be reasonable, necessary, appropriate or desirable to carry out the provisions of this Agreement or the Act.

12.4 <u>Severability</u>. If any provision of this Agreement shall be declared illegal, invalid, or unenforceable in any jurisdiction, then such provision shall be deemed to be severable from this Agreement (to the extent permitted by law) and in any event such illegality, invalidity or unenforceability shall not affect the remainder hereof.

12.5 <u>Entire Agreement</u>. This Agreement and exhibits attached hereto constitute the entire Agreement of the Partners and supersede all prior written agreements and prior and contemporaneous oral agreements, understandings and negotiations with respect to the subject matter hereof.

12.6 <u>**Pronouns and Plurals**</u>. When the context in which words are used in the Agreement indicates that such is the intent, words in the singular number shall include the plural and the masculine gender shall include the neuter or female gender as the context may require.

12.7 <u>Headings</u>. The Article headings or sections in this Agreement are for convenience only and shall not be used in construing the scope of this Agreement or any particular Article.

12.8 <u>Counterparts</u>. This Agreement may be executed in several counterparts, each of which shall be deemed to be an original copy and all of which together shall constitute one and the same instrument binding on all parties hereto, notwithstanding that all parties shall not have signed the same counterpart.

12.9 <u>**Governing Law**</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware; provided, however, that any cause of action for violation of federal or state securities laws shall not be governed by this <u>Section 12.9</u>.

IN WITNESS WHEREOF, the parties hereto have hereunder affixed their signatures to this Agreement of Limited Partnership, all as of the 12th day of December, 2014.

GENERAL PARTNER:

HINES GLOBAL REIT II, INC., a Maryland corporation

By: <u>/s/ Sherri W. Schugart</u> Name: Sherri W. Schugart Title: President and Chief Executive Officer

LIMITED PARTNER:

HINES GLOBAL REIT II ASSOCIATES LIMITED PARTNERSHIP

By: <u>/s/ Charles M. Baughn</u> Name: Charles M. Baughn Title: Senior Managing Director/Chief Financial Officer

EXHIBIT A

PARTNERS, CAPITAL CONTRIBUTIONS AND PERCENTAGE INTERESTS

General Partner and Limited Partners (other than the Special OP Unitholder)

Partner	Cash Contribution as of August 15, 2014	Agreed Value of Non-Cash Capital Contribution	Partnership Units as of August 15, 2014	Percentage Interest as of August 15, 2014
GENERAL PARTNER:				
Hines Global REIT II, Inc. 2800 Post Oak Boulevard, Suite 5000 Houston, Texas 77056-6118	\$10,000	_	1,111.111 Class A Units	5.0%
ORIGINAL LIMITED PARTNER:				
Hines Global REIT II Associates Limited Partnership 2800 Post Oak Boulevard, Suite 5000 Houston, Texas 77056-6118	\$190,000		21,111.111 Class A Units	95.0%
TOTALS	\$200,000	—	22,222.222 Class A Units	100.00%

Special OP Unitholder

Hines Global REIT II Associates Limited Partnership 2800 Post Oak Boulevard, Suite 5000 Houston, Texas 77056-6118

EXHIBIT B

NOTICE OF EXERCISE OF REDEMPTION RIGHT

In accordance with <u>Section 8.4</u> of the Amended and Restated Limited Partnership Agreement (the "Agreement") of Hines Global REIT II Properties LP, the undersigned hereby irrevocably (i) presents for redemption ______ Partnership Units in Hines Global REIT II Properties LP in accordance with the terms of the Agreement and the Redemption Right referred to in <u>Section 8.4</u> thereof, (ii) surrenders such Partnership Units and all right, title and interest therein, and (iii) directs that the Cash Amount or REIT Shares Amount (as defined in the Agreement) as determined by the General Partner deliverable upon exercise of the Redemption Right be delivered to the address specified below, and if REIT Shares (as defined in the Agreement) are to be delivered, such REIT Shares be registered or placed in the name(s) and at the address(es) specified below.

Dated:

(Name of Limited Partner)

(Signature of Limited Partner)

(Mailing Address)

(City) (State) (Zip Code)

Signature Guaranteed by:

If REIT Shares are to be issued, issue to: Name:

Social Security or Tax I.D. Number: _____

LIST OF SUBSIDIARIES

NameJurisdiction ofHGREIT II 2819 Loker LPDelaware

CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Sherri W. Schugart, certify that:

- 1. I have reviewed this Annual Report on Form 10-K of Hines Global REIT II, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - a. designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. intentionally omitted;
 - c. evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

March 25, 2015

By: /s/ Sherri W. Schugart Sherri W. Schugart President and Chief Executive Officer

CERTIFICATION PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Ryan T. Sims, certify that:

- 1. I have reviewed this Annual Report on Form 10-K of Hines Global REIT II, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:
 - a. designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b. intentionally omitted;
 - c. evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d. disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a. all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b. any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

March 25, 2015

By: /s/ Ryan T. Sims

Ryan T. Sims Chief Financial Officer and Secretary

WRITTEN STATEMENT OF CHIEF EXECUTIVE OFFICER AND CHIEF FINANCIAL OFFICER PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

The undersigned, the Chief Executive Officer and the Chief Financial Officer of Hines Global REIT II, Inc. (the "Company"), pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, each hereby certifies that to his/her knowledge, on the date hereof:

- (a) the Annual Report Form 10-K of the Company for the year ended December 31, 2014 filed on the date hereof with the Securities and Exchange Commission (the "Report") fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (b) the information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: March 25, 2015

/s/ Sherri W. Schugart Sherri W. Schugart President and Chief Executive Officer

Date: March 25, 2015

/s/ Ryan T. Sims

Ryan T. Sims Chief Financial Officer and Secretary